ENVIRONMENTAL ASSESSMENT

PROPOSED CONSTRUCTION, DEMOLITION, AND OTHER ACTIONS AT THE 126^{TH} AIR REFUELING WING ILLINOIS AIR NATIONAL GUARD

SCOTT AIR FORCE BASE, ILLINOIS

AIR NATIONAL GUARD ENVIRONMENTAL DIVISION

Report Documentation Page

Form Approved OMB No. 0704-0188

Public reporting burden for the collection of information is estimated to average 1 hour per response, including the time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding this burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden, to Washington Headquarters Services, Directorate for Information Operations and Reports, 1215 Jefferson Davis Highway, Suite 1204, Arlington VA 22202-4302. Respondents should be aware that notwithstanding any other provision of law, no person shall be subject to a penalty for failing to comply with a collection of information if it does not display a currently valid OMB control number.

1. REPORT DATE 05 JUN 2006	2. REPORT TYPE	3. DATES COVERED 00-00-2006 to 00-00-2006
4. TITLE AND SUBTITLE		5a. CONTRACT NUMBER
Environmental Assessment: Proposed Other Actions at the 126th Air Refueli		5b. GRANT NUMBER
Guard		5c. PROGRAM ELEMENT NUMBER
6. AUTHOR(S)		5d. PROJECT NUMBER
		5e. TASK NUMBER
		5f. WORK UNIT NUMBER
7. PERFORMING ORGANIZATION NAME(S) AND AI Science Applications International Co Street,Tucson,AZ,85716		8. PERFORMING ORGANIZATION REPORT NUMBER
9. SPONSORING/MONITORING AGENCY NAME(S) A	AND ADDRESS(ES)	10. SPONSOR/MONITOR'S ACRONYM(S)
		11. SPONSOR/MONITOR'S REPORT NUMBER(S)
12. DISTRIBUTION/AVAILABILITY STATEMENT Approved for public release; distribut	ion unlimited	
13. SUPPLEMENTARY NOTES		
^{14. ABSTRACT} Under the Proposed Action, the 126 A their five-year construction program t facilities, modifications to some existin Action, the eight construction and den	hat would include eight specific pro g facilities, and demolition of one fa	jects: construction of some new acility. Under the Alternative Proposed Action would still

occur, in addition to three additional construction projects. The 126 ARW would also release their eight assigned KC-135E model aircraft and would acquire 12 KC-135R model aircraft. The number of aircraft operations would increase by approximately 50 percent as a result of the four additional assigned aircraft. There would also be an increase of 27 personnel (firefighters) as a result of this alternative. During the preparation of this EA, no significant impacts as a result of implementation of either the Proposed Action or the Alternative Action have been identified, and therefore, an Environmental Impact Statement is not required, and will not be prepared. Resources considered in the EA include earth resources, water resources, biological resources, air quality, noise, land use, socioeconomics and environmental justice, cultural resources, safety, solid and hazardous materials and wastes, and infrastructure. The EA was made available to the public for comment for a 30-day period (February 5 though March 10). No public comments were received. The EA was also coordinated with local, state, and federal agencies at the same time. There were only two agency responses (United States Fish and Wildlife Service [USFWS] and State Historic Preservation Office [SHPO]). Both were in concurrence with the findings in the EA.

a. REPORT unclassified	ь. ABSTRACT unclassified	c. THIS PAGE unclassified	ABSTRACT Same as Report (SAR)	OF PAGES 159	RESPONSIBLE PERSON				
16. SECURITY CLASSIFIC	17. LIMITATION OF	18. NUMBER	19a. NAME OF						
15. SUBJECT TERMS									

Standard Form 298 (Rev. 8-98) Prescribed by ANSI Std Z39-18

FINDING OF NO SIGNIFICANT IMPACT FOR PROPOSED CONSTRUCTION, DEMOLITION, AND OTHER ACTIONS AT THE 126TH AIR REFUELING WING ILLINOIS AIR NATIONAL GUARD SCOTT AIR FORCE BASE, ILLINOIS

PURPOSE: The purpose of the Proposed Action is to provide the 126th Air Refueling Wing (126 ARW) with properly sized and configured facilities that are required to effectively accomplish their mission. The Proposed Action is necessary to reconfigure facilities to accommodate the continuously evolving mission of the 126 ARW. The new facilities would enhance the 126 ARW's ability to maintain a level of wartime readiness necessary to support the mission.

PROPOSED ACTION: Under the Proposed Action, the 126 ARW will implement construction projects in support of their air refueling mission that are planned for the next five-year period. Associated with this construction activity will be demolition of one facility that is obsolete, deteriorated, and in the footprint of one of the proposed facilities.

ALTERNATIVE ACTION: Under the Alternative Action, all construction and demolition associated with the Proposed Action would occur in addition to three other construction projects. Additionally, the 126 ARW would release their 8 Primary Assigned Aircraft (PAA) KC-135E model aircraft and would acquire 12 PAA KC-135R model aircraft. The number of aircraft operations would increase by approximately 50 percent as a result of the four additional PAA. In addition, there would be an increase of 27 personnel at the installation.

NO ACTION ALTERNATIVE: Under the No Action Alternative, the 126 ARW would maintain their existing facilities and would not build any of the new facilities proposed. In general, the No Action Alternative would require that the 126 ARW continue to operate under unnecessarily inefficient conditions, and that anti-terrorism/force protection (AT/FP) requirements continue to be unmet. Deficiencies could continue to impair the 126 ARW's ability to successfully conduct their mission and to maintain wartime readiness and training.

SUMMARY OF FINDINGS:

Potential impacts associated with the Proposed Action and the Alternative Action have been assessed with regard to the following environmental resource areas:

Earth Resources. There will be 1.80 acres of surface disturbance over the course of the construction program associated with the Proposed Action (and 3.48 acres under the Alternative

Action). The grading of existing soil and placement of structural fill for new facilities will not substantially alter existing soil conditions at the installation because, to a large extent, the construction described above is planned for areas where surface disturbance has previously occurred. Best Management Practices (BMPs) will be used to limit soil movement, stabilize runoff, and control sedimentation. Impacts to earth resources are expected to be minimal.

Water Resources. There will be approximately 1.75 acres of new impervious surface that will result under the Proposed Action (and 3.43 acres under the Alternative Action). The 126 ARW will update their Storm Water Pollution Prevention Plan (SWPPP) to include these projects and will obtain, as appropriate, coverage under a National Pollutant Discharge Elimination System (NPDES) Construction Storm Water permit from the Illinois Environmental Protection Agency (Illinois EPA). Adherence to the requirements of the permit will include implementation of BMPs to minimize the potential for exposed soils or other contaminants from construction activities to reach nearby surface waters. It is expected that impacts to water resources will be minimal.

Biological Resources. In general, projects described under the Proposed Action and the Alternative Action are located at sites that are highly altered by man. The potential for any federally listed species to occur on the Base is low, with the exception of the Indiana Bat. Recent correspondence with the Illinois Department of Natural Resources (IDNR) revealed that no records exist of any state-listed threatened or endangered species on or adjacent to the Base. The 126 ARW will coordinate, as necessary, with the United States Fish and Wildlife Service (USFWS) prior to implementation of construction activities to ensure that impacts to sensitive species do not occur. Impacts to biological resources are expected to be minimal.

Air Quality. In general, combustive and fugitive dust emissions from proposed construction activities under the Proposed Action or the Alternative Action will produce localized, elevated air pollutant concentrations that will occur for a short duration and will not result in any long-term impacts on the air quality of St. Clair County or Air Quality Control Region (AQCR) 70. Impacts to air quality in the County and the AQCR are expected to be minimal.

Noise. Construction noise emanating off-site as a result of the Proposed Action will likely be noticeable in the immediate construction site vicinity, but is not expected to create adverse impacts. In addition, noise associated with the aircraft conversion and the associated construction activities described under the Alternative Action will similarly not be expected to create adverse impacts because the noise contours actually shrink under this alternative. The acoustic environment on and near the airfield property is expected to remain relatively unchanged from existing conditions. Impacts from noise are expected to be minimal.

2

Land Use/Visual Resources. The proposed construction and demolition projects described under the Proposed Action and the Alternative Action are expected to enhance overall installation planning and compatibility of functions on the 126 ARW installation in general. Some existing incompatibilities will be corrected. Visual resources are generally not expected to be impacted. Impacts to land use and visual resources are expected to be minimal.

Socioeconomics/Environmental Justice. There are no substantial long-term changes in population and/or employment as a result of implementation of the Proposed Action. Although there is an anticipated increase of 27 personnel under the Alternative Action, this is a minor increase and will not result in any appreciable impact. Additionally, the projects described under the alternatives are not expected to create adverse environmental or health effects, and therefore no disproportionately high or adverse impacts to minority, low-income, or youth populations are expected. Impacts to socioeconomics and environmental justice are expected to be minimal.

Cultural Resources. Activities associated with the Proposed Action or the Alternative Action are not expected to impact archaeological or traditional resources. Demolition of Building 5542 will be coordinated with the Illinois State Historic Preservation Officer (SHPO) to determine its eligibility to the National Register of Historic Places (NRHP) prior to demolition. Impacts to cultural resources are expected to be minimal.

Safety. Implementation of the Proposed Action or the Alternative Action does involve ground activities that may expose workers performing the required site preparation, grading, and building construction to some risk. Strict adherence to all applicable occupational safety requirements will minimize the relatively low risk associated with these construction activities. All projects have been sited outside any quantity-distance arcs, as appropriate. Additionally, the proposed projects will include measures to enhance and correct AT/FP shortfalls as part of the facility designs. Impacts to safety are expected to be minimal.

Hazardous Materials and Waste Management. The proposed construction and demolition projects associated with the Proposed Action or the Alternative Action will generate construction and demolition waste that will be recycled and/or taken to a local demolition landfill, as appropriate. There are no capacity issues with the existing landfills. Hazardous materials and wastes will be handled, stored, and disposed of in accordance with applicable regulations. Any asbestos-containing material (ACM), lead-based paint (LBP), or contaminated soils associated with Environmental Restoration Program (ERP) sites will be removed and disposed of per applicable regulations. Impacts as a result of hazardous materials and waste management are expected to be minimal.

Infrastructure. The proposed construction and demolition projects associated with the Proposed Action or the Alternative Action will result in some temporary interruption of utility services and

minor hindrance of transportation and circulation during construction activities. These impacts will be temporary, occurring only for the duration of the construction period. In general, infrastructure at the 126 ARW installation will improve under these actions. Impacts to infrastructure are expected to be minimal.

PUBLIC INVOLVEMENT: The National Environmental Policy Act (NEPA), 40 Code of Federal Regulations (CFR) 1500-1508, and 32 CFR 989 require public review of the Environmental Assessment (EA) before approval of the Finding of No Significant Impact (FONSI) and implementation of the Proposed Action or the Alternative Action. A notice of availability for public review of the Draft EA was published in the Belleville News-Democrat on Sunday, February 5 and Sunday, February 19, 2006. The Draft EA was available for public review from February 5 through March 10, 2006. No public comments were received. The EA was also coordinated with local, state, and federal agencies during the same time period. Two agency responses (USFWS and SHPO) were received. Both were in concurrence with the findings in the EA.

FINDING OF NO SIGNIFICANT IMPACT (FONSI): Based on my review of the facts and analysis in the Environmental Assessment, I conclude that neither the Proposed Action nor the Alternative Action will have a significant impact either by itself or considering cumulative impacts. Accordingly, the requirements of NEPA, the Council on Environmental Quality (CEQ), and 32 CFR 989, et seq. have been fulfilled, and an Environmental Impact Statement (EIS) is not necessary and will not be prepared.

RAYMOND J. KOTTMAN, Colonel, USAF Commander

IS MAY OC

Date

ACRONYMS AND ABBREVIATIONS

Jagin Introducting Ming Price 126 ARW 375 Ark releasing Wing FONSI Finding of No Significant Impact 375 AW 375 Ark rift Wing FONSI Finding of No Significant Impact 375 CDS/CEV Transformering Squadon, FV Tiscal year ACQS ambient air quality standards HAZMAT Hazardons Materials Pharmacy ACHP Advisory Council on Historic Hz Hertz Preservation I-64 Hinestate 64 ACHP Aris Expeditionary Force IDNR Illinois Department of Natural Resources AEEC Ari Force Instruction ILANG Illinois Linoimental Planning AFIB Ari Force Instruction ILANG Illinois Linoimental Planning AFIEKA Ari Force Instruction ILANG Illinois Linoimental Planning AGL Air Force Instruction ILANG Illinois Linoimental Planning AGL Air Force Instruct for Linoimental Planning Illinois Linoimental Planning Ari Force Instruct for Linoimental Planning Illinois Linoimental Planning Analysis AGL Ai	µg/m ³	mierograma nor aubia matar	FIFRA	Endered Incontinida, Europiaida, and
375 AW 375 ACHIR Wing FONSI Finding of No Significant Impact 375 CTS/CTF 757 Civi Findingering Squadron, Environmental Flight HAP High Accident Netential AAQS ambient air quality standards HAZMAT Hazrato ACHP Advisory Council on Historic Hz Herr Preservation L64 Interstate 64 ACM asbestos-containing material IAC Illinois Administrate Code AEF Ari Expeditionary Force IDNR Illinois Experiment of Natural Resources AEG Ari Force Institute for Environment, Environmental Planting Illinois FAR Illinois FAR AFFRA Air Force Institute for Environment, Analysis Illinois FAR Illinois Environmental Protection Agency AGL abver ground level Logian equivalent noise level over a 24-hour AGL abver ground level Logian equivalent noise level over a 24-hour AGL abver ground level Logian Maxional Actin Material Standards AGL Air Motility Command Lmax Maxional Ambient Air Quality Standards AGC Air Motility Command Lmax Maxional Ambient Air Quality Standards AGC Air Motility Command Lmax Maxional Ambient Air Quality Standards AGC Ai		micrograms per cubic meter	FIFKA	Federal Insecticide, Fungicide, and
375 CESCEV 37.5 Civil Engineering Squadron, FY fiscal year AAQS ambient air quality standards HAP High Accident Potential ACHP Advisory Connell on Historic HAP High Accident Potential ACM asbestos-containing material LAC Herz ACM asbestos-containing material LAC Illinois Advisory Concernation AFB Air Force Instruction IL Coordination for invironmental Planning AFI Air Force Instruction IL ANG Illinois Air National Planning AFIERA Air Force Instruction IL ANG Illinois Air National Planning AFIERA Air Force Instruction IL ANG Illinois Air National Planning ACUZ Air Force Policy Directive Logato Day-Night Average Sound Level ARFA Air Force Policy Directive Logato Day-Night Average Sound Level ARFA Air Mational Inseliguitos Planning Hard Lagato ACUZ Air Installation Compatible Use Zone Logato Day-Night Average Sound Level ARFA Air Mability Command LafA LagAtore Maximum Sound		375 th Airlift Wing	FONSI	
Environmental Flight - HAP High Acciden Potential AAQS ambient air quality standards HAZMAT Hazardous Materials Pharmacy HAZMAT Hazardous Materials Pharmacy Hz/A Advisory Council on Historic Hz/Hz/Hazardous Materials Pharmacy Hz/A Advisory Council on Historic Hz/Hz/Hz/Hz/Hz/Hz/Hz/Hz/Hz/Hz/Hz/Hz/Hz/H				
AAQS ambient air quality standards HAZMAT Hazardous Materials Pharmacy ACHP Advisory Council on Historic HZ Hertz CM asbestos-council on mistoric HZ Hertz ACM asbestos-council on guardiantical IAC Hinois Administrative Code AEDZ Airport Environs Overlay Zone IICEP Interagency and Intergovernmental AFB Air Force Instruction Hinois EPA Illinois Convented Planning AFIERA Air Force Instruction Hinois ENA Illinois Environmental Planning AFIERA Air Force Instruction Lag Daylot Air Mational Guard AFIERA Air Force Indian Religious Predom Act Lag Daylot Predom Act AGL above ground level Lag Daylot Predom Act Daylot Predom Act ARFA American Indian Religious Predom Act Daylot Predom Act Daylot Predom Act AMC Air Mobility Command Lag Maximum Sound Level Daylot Predom Act AQC Areiden Predinia Zone NAGPRA Native American Indian Religious Act AQC Air Mobility Control Region NAGPRA Native American Carses Protection and AST above ground storage tank Council on Environmental Preservation Act BMP Beid-	575 CE5/CE V			
ACHP Advisory Conneil on Historic Hz Hertz Preservation I-64 Interstate 64 ACM asbestos-containing material IAC Illinois Administrative Code AEF Air Expeditionary Force IDNR Illinois Department of Natural Resources AFI Air Force Instruction ILANG Illinois Environmental Planning AFI Air Force Instruction Environment, Illinois Environmental Protection Agency Safety, and Occupational Health Risk LBP lead-based paint Anil System L _{a6} Day-Night Average Sound Level AGL above ground level L _{a6} period AGCU Air Korei Policy Directive L _{a6} Maximum Sound Level ARFA American Indian Religious Freedom Act period Maximum Sound Level ANGC Air Kouliny Command Law Maximum Sound Level AQC Air Guality Control Region NAAQS National Ambient Air Quality Standards AQCR Air Quality Control Region NAAQS National Emvision Inventory BAI backup aircraft inventory NEC National Emvision Act AGC	AAOS			
PreservationI-64Interstate 64ACMasbestos-containing materialIACIllinois Administrative CodeAEDAirport Environs Overlay ZoneIICEPInteragency and IntergovernmentalAFBAir Force InstructionILLANGIllinois Environmental PlanningAFIAir Force InstructionIllinois Environmental PlanningAFIERAAir Force InstructionIllinois Environmental Protection AgencySafery, and Occupational Health RiskLBPIllinois Environmental Protection AgencyAFIERAAir Force Policy DirectiveLagDay-tight Average Sound LevelAGLabove ground levelequivalent noise level over a 24-hourARFAAmerican Indian Religious Fredom ActperiodARCAAir National GuardLTAlighter Than AirAOCArea of ConcernMSLabove man sea levelANGAir National GuardLTAlighter Than AirAQCRAir Quality Control RegionNAGPRANational Ambient Air Quality StandardsAQCAAir Quality Control RegionNAGPRANational Electrical CodeBAHbackup aircraft inventoryNFENational Electrical CodeBAHbackup aircraft inventoryNFENational Environmental Poicy ActBASHbefore presentNO,nitrogen oxidesCPQCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCRAClean Air ActNPDESNational Environmental Poicy ActAPZAccident Potential ResponseF				-
ACM asbestos-containing material IAC Illinois Administrative Code AEF Air Expeditionary Force IDN Illinois Department of Natural Resources AEOZ Airport Environs Overlay Zone IICEP Interagency and Intergovernmental Resources AFB Air Force Instruction Environment, IILANG Illinois Environmental Planning AFIEA Air Force Institute for Environment, IILInois ENV Illinois Environmental Protection Agency AGL Abore ground level Lag equivalent noise level over a 24-hour AGL above ground level Lag equivalent noise level over a 24-hour AGC Air Instainalino Compatible Use Zone Lag equivalent noise level over an eight-hour AREFA American Indian Religious Freedom Act period Maximum Sound Level ANC Air Mobinity Command Lag Maximum Sound Level APZ Accident Potential Zone NAAQS National Environmental Potection and AST aboveground storage tank Repatriation Act Repatriation Act ATFP anti-terrorism/force protection NEC National Environmental Poticy Act BAI backay aircraft inventor	nem			
AFFAir Expeditionary ForceIDNRIllinois Department of Natural ResourcesAEOZAirport Environs Overlay ZoneIICEPIntragency and IntergeorerumentalAFBAir Force BaseCoordination of Environmental PlanningAFIAir Force Institute for Environment,Illinois EPAAFIERAAir Force Institute for Environment,Illinois EPAAAIPOPAir Force Institute for Environment,Illinois EPAAGIabove ground levelLateAFPDAir Force Policy DirectiveLateAGICAir Motional GuardEraARFAAmerican Indian Religois Freedon ActperiodAMCAir Mobility CommandLateANGAir National GuardLateANGAir National GuardLateAQCRAir Quality Control RegionNAAQSAJUCAir Quality Control RegionNAAQSASTabove ground storage tarkRepatriation ActATFPanti-terrois force protectionNFCASIbevergound storage tarkNational Environmental Policy ActBASHberlow ground straceNO,BASHberlow ground straceNO,BASHberlow ground straceNO,CAClean Air ActCAClean Air ActCAConcil an Environmental ResourcesGardConcil an Environmental ResourcesGardConcil an Environmental ResourcesAGEAir Mational GuardASTberlow ground straceADCNPA <td< td=""><td>ACM</td><td></td><td></td><td></td></td<>	ACM			
AFDZ Airport Environs Overlay Zone IICEP Interagency and Intergovernmental Coordination for Environmental Planning AFI Air Force Base Coordination for Environmental Martine Servironmental Planning AFIERA Air Force Institute for Environment, AII and Occupational Health Risk LBP Iead-based paint AFIERA Air Force Policy Directive Lag(2) equivalent noise level over a 24-hour period AGL above ground level Lag(2) equivalent noise level over an eight-hour period AICUZ Air Instaliation Compatible Use Zone Lag(2) equivalent noise level over an eight-hour period AICUZ Air Mobility Command LTA Lighter Than Air AOC Area of Concern NAGPRA Native American Graves Protection and Repatriation Air Quality Standards AQCR Air Quality Control Region NAGPRA Native American Graves Protection and Repatriation Act AT/FP anti-terrorism/force protection NEC National Emission Inventory BASH Bird-Aircraft Strike Hazard NEPA National Environmental Policy Act bgs below ground surface NO ₂ nitrogen dixide BAP besto anization NGC national Environmental Policy Act BAF Bird-Aircraft Strike Hazard NEPA National Environmental Policy Act				
AFBAir Force BaseCoordination for Environmental PlanningAFIAir Force InstructionILANGIllinois Air National GuardAFIERAAir Force Institute for Environment,Illinois ENVironmental Protection AgencySafety, and Occupational Health RiskLaDay-Night Average Sound LevelAFPDAir Force Policy DirectiveLaDay-Night Average Sound LevelAFPDAir Force Policy DirectiveLaDay-Night Average Sound LevelAGLabove ground levelperiodAIRFAAmerican Indian Religious Freedom ActLog(8)equivalent noise level over an eight-hourAIRFAAmerican Indian Religious Freedom ActLog(8)equivalent noise level over an eight-hourANGAir Mobility CommandLawMaximum Sound LevelAOCArir Aubility Control RegionNAAQSNational Ambient Air Quality StandardsAQCRAir Quality Control RegionNAAGPRANative American Graves Protection and Repatriation ActAT/FPanti-terrorism/force protectionNECNational Electrical CodeBAIbackup aircraft InventoryNELNational Electrical CodeBASHBirde-Aircraft Strike HazardNEPANational Historic Preservation ActBMPBetfore presentNO,mitrogen oxideCAAClean Air ActNPDESNational Historic Preservation ScriceCAAClean Air ActNPDESNational Religetor of Historic PlacesCOcanoin onoxidePA/SIPreliminary Site Assessment/SiteCPRCode				
AFIAir Force InstructionILANGIllinois Environmental Protection AgencyAFIERAAir Force InstructionHinkikLDPIead-based paintARPDAir Force Policy DirectiveLag(2)equivalent noise level over a 24-hourAGLabove ground levelLag(2)equivalent noise level over a 24-hourAGLAbove ground levelLag(2)equivalent noise level over an eight-hourARFAAmerican Indian Religious Freedom ActLag(2)equivalent noise level over an eight-hourANGAir National GuardLTALighter Than AirAOCArea of ConcernNAAQSNational Air Quality StandardsAQCRAir Quality Control RegionNAAQRNative American Graves Protection and Reparitation ActAQCRAir Quality Control RegionNAGPRANative American Graves Protection and Reparitation ActASTaboverground storage tankRCCNational Envision InventoryASHBirl-Aircraft Strike HazardNFPANational Finission InventoryBASHBirl-Aircraft Strike HazardNFPANational Finission Protection ActBMPBest Management PracticeNO2nitrogen dioxideBMPBest Management PracticeNO2nitrogen oxidesCAAClean Air ActNPDESNational Pollutant Discharge EliminationCATSComprehensive Environmental Response, Comparator, and Lability ActOW2oitrogen oxidesCPRCode of Federal RegulationsP.LPublic LawCOcarefor proseed Action and 			neer	
AFIERAAir Force Institute for Environment, Safety, and Occupational Health Risk LBPIllinois EPA LBPIllinois environmental Protection Agency Lat Lat Day-Night Average Sound LevelAFPDAir Force Policy Directive above ground levelLat Lat Lat Lat Lat Lat Lat American Indian Religious Freedom Act Lat AMCEvel Over a 24-hour periodAICUZAir Installation Compatible Use Zone AIREA A merican Indian Religious Freedom Act Lat AMCLat Lat Lighter Than Air American Indian Religious Freedom Act Lat Lat Lat MCCmerican Indian Religious Freedom Act Lat Lat 			ILANG	
Safety, and Occupational Health RiskLBPlead-based paintAFPDAir Force Policy DirectiveLq(2+)oquivalent noise level over a 24-hourAGLabove ground levelrequivalent noise level over a 24-hourperiodAGLabove ground levelrequivalent noise level over a 24-hourperiodAIRTAAmerican Indian Religious Freedom Actrequivalent noise level over an eight-hourAIRTAAmerican Indian Religious Freedom Actrequivalent noise level over an eight-hourANCAir Mobility CommandLTALighter Than AirAOCArea of ConcernMSLabove mean sea levelAPZAccident Potential ZoneNAAQSNational Ambient Air Quality StandardsAQCRAir Quality Control RegionNAGPRANational Environmental Policy ActAT/FPanti-terrorism/force protectionNECNational Environmental Policy ActBASHbird-Aircraft Strike HazardNEPANational Environmental Policy ActBASHbird-Aircraft Strike HazardNPPANational Environmental Policy ActBMPbefore presentNO ₂ nitrogen oxideBMPbefore presentNO ₂ nitrogen oxideCEQCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCCRLACompensation, and Liability ActOyoroneCERCLACompensation, and Liability ActOyoroneCCRCLACompensation, and Liability ActOyoroneCCRCLACompensation ActPA/Aprimary assign			Illinois EPA	
Are FOR Policy DirectiveLanDay-Night Average Sound LevelAGLabove ground levelequivalent noise level over a 24-hourAGLabove ground levelequivalent noise level over a eight-hourAGLAmerican Indian Religious Freedom Actequivalent noise level over an eight-hourAIRFAAmerican Indian Religious Freedom Actmaxim Sound LevelAMCAir Mobility CommandLmaxMaximm Sound LevelANGAir National GuardLTALighter Than AirAOCArea of ConcernMSLabove means ea levelAPZAccident Potential ZoneNAAOSNational Ambient Air Quality StandardsAQCRAir Quality Control RegionNAAOPRANative American Graves Protection and Repatiation ActAT/FPanti-terroism/force protectionNECNational Electrical CodeBA1backup aircraft inventoryNEINational Electrical CodeBA1backup aircraft inventoryNEINational Electrical CodeBA1bedwag meraft inventoryNEFANational Historic Preservation ActBPbefore presentNO2nitrogen oxidesCAAClean Air ActNPDESNational Pollutant Discharge EliminationCAAClean Air ActNagNational Resources Conservation ServiceCCBCConneristive Environmental Response"Fdegreese FahrenheitCERCLAComprehensive Environmental Response"Fdegreese FahrenheitCCDcacid Air ActNational Respired aircraftDHDDDepa				
AFPDAir Force Policy DirectiveLog(2i)equivalent noise level over a 24-hour periodAGLabove ground levelLog(3i)equivalent noise level over an eight-hour periodAIRUZAir Installation Compatible Use ZoneLog(3i)equivalent noise level over an eight-hour periodAIRFAAmerican Indian Religious Freedom ActLog(3i)equivalent noise level over an eight-hour periodANGAir National GuardLmaxMaximum Sound LevelANGAir National GuardLTALighter Than AirAOCArea of ConcernMSL above mean sea levelApzAQCRAir Quality Control RegionNAGPRANative American Graves Protection and Repatriation ActAT/FPanti-terrorism/force protectionNECNational Emission InventoryBAIbackup aircraft inventoryNEINational Environmental Policy Actbgsbelow ground surfaceNIPANational Environmental Policy ActBMPBest Management PracticeNO2nitrogen dioxideBMPbefore presentNO2nitrogen oxidesCAAClean Air ActNPDESNational Resources Conservation ServiceCERCLACompensation, and Liability ActO3ozoreCCBRCode of Federal RegulationsP.L.Public LawCOcarlon monoxidePA/SIPreliminary Site Assessment/Site InvestigationCWAClean Water ActOWSoil/water separatorCERCLACompensation, and Liability ActO3ozoreCCRLA<				
AGL above ground level period AICUZ Air Installation Compatible Use Zone Log(3) equivalent noise level over an eight-hour AIRFA American Indian Religious Freedom Act period maximum Sound Level AMC Air Mobility Command Lmax Maximum Sound Level ANG Air National Guard LTA Lighter Than Air AOC Area of Concern MSL above mean sea level APZ Accident Potential Zone NAAQS National Ambient Air Quality Standards AQCR Air Quality Control Region NAAGPRA Native American Graves Protection and Repatriation Act AT/FP anti-terroism/force protection NEC National Emission Inventory BAI backup aircraft strike Hazard NEPA National Environmental Policy Act By below ground surface NIPA National Environmental Policy Act BMP Best Management Practice NO ₂ nitrogen oxides CAA Clean Air Act NPDES National Register of Historic Preservation Act CAA Clean Air Act NPDES National Register of Historic Places Comprehensive Environmental Quality<	AFPD			
AICUZ Air Installation Compatible Use Zone Lag(0) equivalent noise level over an eight-hour period AIRFA American Indian Religious Freedom Act period maximum Sound Level AMC Air National Guard LTA Lighter Than Air ANC Air National Guard LTA Lighter Than Air AOC Area of Concern MSL above mean sea level APZ Accident Potential Zone NAAQS National Ambient Air Quality Standards AQCR Air Quality Control Region NAGPRA Native American Graves Protection and Repatriation Act AT/FP anti-terrorism/force protection NEC National Electrical Code BA1 backup aircraft inventory NEI National Electrical Code BASH Bird-Aircraft Strike Hazard NEPA National Emission Inventory BMP Betst Management Practice NO ₂ nitrogen divide System CAA Clean Air Act NPDES National Register of Historic Preservation Service CAA Comprehensive Environmental Quality NRCS Natural Resource Conservation Service CATS Comprehensive Environmental Response T degre			cq(24)	
AIRFA American Indian Religious Freedom Act max priod AMC Air Mobility Command Lmax Maximum Sound Level ANG Air National Guard LTA Lighter Than Air AOC Area of Concern MSL above mean sea level APZ Accident Potential Zone NAAQS National Ambient Air Quality Standards AQCR Air Quality Control Region NAGPRA Native American Graves Protection and Repatriation Act AST aboveground storage tank Repatriation Act Rational Electrical Code BA1 backup aircraft inventory NEI National Emission Inventory BASH Bird-Aircraft Strike Hazard NEPA National Environmental Policy Act Bys below ground surface NO _Q nitrogen dixide BP best Management Practice NO _Q nitrogen dixide CAA Clean Air Act NPDES National Resistrice Conservation Service CAA Combat Arms training Simulator System System CEQ Conicil on Environmental Response °F degrees Fahrenheit Facilitation Act OWS Oil/water separator <td>AICUZ</td> <td></td> <td>$L_{eq(8)}$</td> <td>1</td>	AICUZ		$L_{eq(8)}$	1
AMCAir Mobility CommandLmaxMaximum Sound LevelANGAir National GuardLTALighter Than AirAOCArea of ConcernMSLabove mean sea levelAPZAccident Potential ZoneNAAQSNational Ambient Air Quality StandardsAQCRAir Quality Control RegionNAGPRANative American Graves Protection and Repatriation ActAT/FPanti-terrorism/force protectionNECNational Electrical CodeBAIbackup aircraft inventoryNEINational Environmental Policy Actbgsbelow ground surfaceNHPANational Environmental Policy Actbgsbelow ground surfaceNHPANational Environmental Policy ActBMPBest Management PracticeNO2nitrogen dioxideBPbefore presentNO2nitrogen dioxideCAAClean Air ActNPDESNational Pollutant Discharge EliminationCAAClean Air ActNO3ozoneCERCLAComprehensive Environmental Response"FCERCLACompensation, and Liability ActO3ozoneCCRCode of Federal RegulationsPL.Public LawCCOcarbon moxidePA/SIpreliminary assigned aircraftdBdecibelPAAprimary assigned aircraftdBdecibelPAAprimary assigned aircraftdBdecibelPAAprimary assigned aircraftdBdecibelPAAprimary assigned aircraftdBdecibelPAAprimary assigned aircra	AIRFA		eq(0)	
ANGAir National GuardLTALighter Than AirAOCArea of ConcernMSLabove mean sea levelAPZAccident Potential ZoneNAAQSNational Ambient Air Quality StandardsAQCRAir Quality Control RegionNAAGPRANative American Graves Protection and Repatriation ActASTaboveground storage tankNEPANational Emission InventoryBAIbackup aircraft inventoryNEINational Emission InventoryBASHBird-Aircraft Strike HazardNEPANational Emission InventoryBASHbelow ground surfaceNHPANational Emission InventoryBAPBest Management PracticeNO2nitrogen dioxideBPbefore presentNO2nitrogen dioxideCAAClean Air ActNPDESNational Pollutant Discharge EliminationCATSCombat Arms Training SimulatorSystemSystemCEQCouncil on Environmental Response, Compensation, and Liability ActO3ozoneCERFACode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationnicromeDDDDepartment of DefensePCIper apita incomeDOPAADescription of Proposed Action andPM10particulate matter less than or equal to 2.5 micromental mass ProcessEIAenvironmental Impact statementPOLpetrolution of Significant DeteriorationCFRCode of Federal RegulationsPL<	AMC		L _{max}	Maximum Sound Level
AOCArea of ConcernMSLabove mean sea levelAPZAccident Potential ZoneNAAQSNational Ambient Air Quality StandardsAQCRAir Quality Control RegionNAAQSNational Ambient Air Quality StandardsASTaboveground storage tankRepatriation ActAT/FPanti-terrorism/force protectionNECNational Environmental Policy ActBAIbackup aircraft inventoryNEINational Environmental Policy ActBASHBird-Aircraft Strike HazardNEPANational InventoryBASHBeird-Aircraft Strike HazardNEPANational InventoryBMPBest Management PracticeNO2nitrogen oxidesCAAClean Air ActNPDESNational Pollutant Discharge EliminationCATSCombat Arms Training SimulatorSystemCEQCouncil on Environmental Response,NRHPNational Register of Historic PlacesCERCLAComprehensive Environmental Response°Fdegrees FahrenheitCERCACommunity Environmental Response°Fdegrees FahrenheitCCRCode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteDDADepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and Register of Historic Placesmicrometers in diameterEAenvironmental Mapisi Processmicrometers in diameterEIAEnvironmental Restoration ProgramROIpertoleum, oil and lubricants <t< td=""><td>ANG</td><td></td><td></td><td>Lighter Than Air</td></t<>	ANG			Lighter Than Air
AQCRAir Quality Control RegionNAGPRANative American Graves Protection and Repartiation ActASTaboveground storage tankRepartiation ActAT/FPanti-terrorism/force protectionNECNational Electrical CodeBAIbackup aircraft inventoryNEINational Electrical CodeBASHBird-Aircraft Strike HazardNEPANational Environmental Policy Actbgsbelow ground surfaceNHPANational Historic Preservation ActBMPBest Management PracticeNO2nitrogen oxidesCAAClean Air ActNPDESNational Pollutant Discharge EliminationCATSCombat Arms Training SimulatorSystemCEQCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCERCLAComprehensive Environmental Response, Compensation, and Liability ActO3ozoneCERFACommunity Environmental Response°Fdegrees FahrenheitGCcarbon monoxidePL.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteDDPAADepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and AlternativesPM10particulate matter less than or equal to 2.5 micrometers in diameterEAenvironmental Impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparticulate matter less than or equal to 2.5 micrometers in diameterEAenvironmental Impact statementPOLpetrole	AOC	Area of Concern	MSL	
AQCRAir Quality Control RegionNAGPRANative American Graves Protection and Repartiation ActASTaboveground storage tankRepartiation ActAT/FPanti-terrorism/force protectionNECNational Electrical CodeBAIbackup aircraft inventoryNEINational Electrical CodeBASHBird-Aircraft Strike HazardNEPANational Environmental Policy Actbgsbelow ground surfaceNHPANational Historic Preservation ActBMPBest Management PracticeNO2nitrogen oxidesCAAClean Air ActNPDESNational Pollutant Discharge EliminationCATSCombat Arms Training SimulatorSystemCEQCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCERCLAComprehensive Environmental Response,NRHPNational Register of Historic PlacesCFRCode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAAprimary assigned aircraftDOPAADepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and Right-to-Know ActPM10particulate matter less than or equal to 2.5 micrometers in diameterEAenvironmental Impact statementPOLpetroluan, oil, and lubricantsDOPAADescription of Proposed Action and Right-to-Know ActPSDPrevention of Significant Deterioration <tr< td=""><td>APZ</td><td>Accident Potential Zone</td><td>NAAQS</td><td>National Ambient Air Quality Standards</td></tr<>	APZ	Accident Potential Zone	NAAQS	National Ambient Air Quality Standards
AT/FPanti-terrorism/force protectionNECNational Electrical CodeBA1backup aircraft inventoryNEINational Envision InventoryBASHBird-Aircraft Strike HazardNEPANational Environmental Policy Actbgsbelow ground surfaceNIPANational Environmental Policy ActBMPBest Management PracticeNO2nitrogen dioxideBPbefore presentNOnitrogen oxidesCAAClean Air ActNPDESNational Pollutant Discharge EliminationCATSCompet Arms Training SimulatorSystemCEQCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCBRCLAComprehensive Environmental Response,NRHPNational Register of Historic PlacesCERCode of Federal RegulationsP.LPublic LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAprimary assigned aircraftdBAA-weighted decibelPAprimary assigned aircraftDOPAADescription of Proposed Action andPM12.5particulate matter less than or equal to 2.5DOPAAEnvironmental Impact Analysis Processmicrometers in diameterEISenvironmental Restoration ProgramRO1Resource Conservation and Resource Conservation and Resource Conservation and Recovery ActERPEnvironmental Impact Analysis Processmicrometers in diameterEISenvironmental Impact Analysis Processmicrom	AQCR	Air Quality Control Region		
BAIbackup aircraft inventoryNEINational Emission InventoryBASHBird-Aircraft Strike HazardNEPANational Environmental Policy Actbgsbelow ground surfaceNHPANational Historic Preservation ActBMPBest Management PracticeNO2nitrogen dioxideBPbefore presentNOxnitrogen oxidesCAAClean Air ActNPDESNational Pollutant Discharge EliminationCATSCombat Arms Training SimulatorSystemCEQCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCERCLAComprehensive Environmental Response, Compensation, and Liability ActO3ozoneCERFACommunity Environmental Response°Fdegrees FahrenheitFacilitation ActOWSoil/water separatorCCCode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPbleadDoDDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action andPM2.5particulate matter less than or equal to 2.5Material assessmentPM10particulate matter less than or equal to 10EIAPEnvironmental Impact Analysis Processmicrometers in diameterEISenvironmental Impact Analysis Processppmparticulate matter less than or equal to 1.5EISenvironmental Restoration ProgramRCI	AST	aboveground storage tank		Repatriation Act
BASHBird-Aircraft Strike HazardNEPANational Environmental Policy Actbgsbelow ground surfaceNHPANational Historic Preservation ActBMPBest Management PracticeNO2nitrogen dioxideBPbefore presentNO2nitrogen oxidesCAAClean Air ActNPDESNational Pollutant Discharge EliminationCATSCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCEQCouncil on Environmental Response,NRHPNational Register of Historic PlacesCERCLAComprehensive Environmental Response,NRHPNatural Resources Conservation ServiceCERCLAComprehensive Environmental Response°Fdegrees FahrenheitCERCLACompensation, and Liability ActO3ozoneCERFACode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationInvestigationdBAA-weighted decibelPAprimary assigned aircraftDoDDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action andPM 2.5maicrometers in diameterEAenvironmental assessmentPM 10particulate matter less than or equal to 2.5EIAPEnvironmental Impact Analysis Processmicrometers in diameterEISenvironmental Impact Analysis Processmicrometers in diameterEISenvironmental Restoration ProgramROIRegi	AT/FP	anti-terrorism/force protection	NEC	National Electrical Code
bgsbelow ground surfaceNHPANational Historic Preservation ActBMPBest Management PracticeNO2nitrogen dioxideBPbefore presentNOxnitrogen oxidesCAAClean Air ActNPDESNational Pollutant Discharge EliminationCATSCombat Arms Training SimulatorSystemCEQCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCERCLAComprehensive Environmental Response,NRHPNational Register of Historic PlacesCERCLAComprehensive Environmental Response,NRHPNational Register of Historic PlacesCERCLACompensation, and Liability ActO3ozoneCERFACode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIpreliminary Site Assessment/SiteCWAClean Water ActInvestigationInvestigationdBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDODDepartment of DefensePCIparticulate matter less than or equal to 2.5Alternativesmicrometers in diametermicrometers in diameterEAenvironmental Impact statementPM10particulate matter less than or equal to 10EIAPEnvironmental Restoration ProgramROIRegion of Significant DeteriorationEVAEnvironmental Restoration ProgramROIRegion of InfluenceEAenvironmental Restoration ProgramROIRegion of InfluenceEISe	BAI	backup aircraft inventory	NEI	National Emission Inventory
BMPBest Management PracticeNO2nitrogen dioxideBPbefore presentNO4nitrogen oxidesCAAClean Air ActNPDESNational Pollutant Discharge EliminationCATSCombat Arms Training SimulatorSystemCEQCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCERCLAComprehensive Environmental Response,NRHPNational Register of Historic PlacesCERFACommunity Environmental Response°Fdegrees FahrenheitCERFACommunity Environmental Response°Fdegrees FahrenheitCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDOPAADescription of Proposed Action andPM 2.5micrometers in diameterEAenvironmental impact statementPOLpetrolutate matter less than or equal to 10EIAPEnvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparticulate matter less than or equal to 10EIAPEnvironmental Restoration ProgramRO1Resource Conservation and Recovery ActERPEnvironmental Impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEIAPEnvironmental Restoration ProgramRO1Resource Conservation and Recovery ActERP<	BASH	Bird-Aircraft Strike Hazard	NEPA	National Environmental Policy Act
BMPBest Management PracticeNO2nitrogen dioxideBPbefore presentNO3nitrogen oxidesCAAClean Air ActNPDESNational Pollutant Discharge EliminationCATSCombat Arms Training SimulatorSystemCEQCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCERCLAComprehensive Environmental Response,NRHPNational Register of Historic PlacesCERFACommunity Environmental Response°Fdegrees FahrenheitCERFACommunity Environmental Response°Fdegrees FahrenheitCOcarbon monoxidePL.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAprimary assigned aircraftdBAA-weighted decibelPbleadDOPAADescription of Proposed Action andPM 2,5particulate matter less than or equal to 2.5Alternativesmicrometers in diameterEAenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEVCRAEmergency Planning and CommunityPSDPrevention of Significant DeteriorationRight-to-Know ActRCRAResource Conservation and Recovery ActERPEnvironmental Restoration ProgramRO1Region of InfluenceESAEndangered Species ActSARASuperfund Amendments andFECRAEnvironm	bgs	below ground surface	NHPA	National Historic Preservation Act
BPbefore presentNOxnitrogen oxidesCAAClean Air ActNPDESNational Pollutant Discharge EliminationCATSCombat Arms Training SimulatorSystemCEQCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCERCLAComprehensive Environmental Response,NRHPNational Register of Historic PlacesCompensation, and Liability ActO3ozoneCERFACommunity Environmental Response°Fdegrees FahrenheitFacilitation ActOWSoil/water separatorCFRCode of Federal RegulationsP.L.Public LawCOcarbon monoxidePASIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDODDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action andPM 2.5micrometers in diameterEAenvironmental impact statementPOLpetroleur, oil, and lubricantsEOExecutive Orderppmparts per millionEVCRAEmergency Planning and CommunityPSDPrevention of Significant DeteriorationRight-to-Know ActRCRAResource Conservation and Recovery ActERPEnvironmental Restoration ProgramROIResioned conservation and Recovery ActERPEnvironmental Restoration ProgramRCIAResource Conservation and Recovery ActERP			NO_2	nitrogen dioxide
CATSCombat Arms Training SimulatorSystemCEQCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCERCLAComprehensive Environmental Response, Compensation, and Liability ActO3ozoneCERFACommunity Environmental Response°Fdegrees FahrenheitFacilitation ActOWSoil/water separatorCFRCode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDODDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and AlternativesPM_2.5particulate matter less than or equal to 2.5EAenvironmental impact Analysis Processmicrometers in diameterEISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and CommunityPSDPrevention of Significant DeteriorationRight-to-Know ActRCRAResource Conservation and Recovery ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments andFESOPFederally Enforceable State OperatingSELSound Exposure LevelFESOPFederall Emergency Management Agency	BP	before present	NO _x	
CEQCouncil on Environmental QualityNRCSNatural Resources Conservation ServiceCERCLAComprehensive Environmental Response, Compensation, and Liability ActNRHPNational Register of Historic PlacesCERFACommunity Environmental Response°Fdegrees FahrenheitFacilitation ActOWSoil/water separatorCFRCode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDoDDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and environmental impact Analysis Processmicrometers in diameterEISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and CommunityPSDPrevention of Significant Deterioration ngight-to-Know ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFESOPFederall Emergency Management AgencyFederally Enforceable State Operating SFSFFESOPFederall Emergency Management AgencySFSquare feet	CAA	Clean Air Act	NPDES	National Pollutant Discharge Elimination
CERCLAComprehensive Environmental Response, Compensation, and Liability ActNRHPNational Register of Historic PlacesCERFACommunity Environmental Response $^{\circ}$ Fdegrees FahrenheitFacilitation ActOWSoil/water separatorCFRCode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDODDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and AlternativesPM10particulate matter less than or equal to 2.5 micrometers in diameterEAenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and CommunityPSDPrevention of Significant Deterioration Right-to-Know ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFESOPFederal Hyenforceable State Operating PermitSELSound Exposure Level	CATS	Combat Arms Training Simulator		System
Compensation, and Liability ActO3ozoneCERFACommunity Environmental Response°Fdegrees FahrenheitFacilitation ActOWSoil/water separatorCFRCode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDoDDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and AlternativesPM10particulate matter less than or equal to 2.5 micrometers in diameterEAenvironmental assessmentPM10particulate matter less than or equal to 10 micrometers in diameterEISenvironmental Impact Analysis Processmicrometers in diameterEOExecutive Orderppmparts per millionEPCRAEmergency Planning and CommunityPSDPrevention of Significant Deterioration Right-to-Know ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFESOPFederal Rengency Management AgencyReauthorization ActFESOPFederal Emergency Management AgencySELSound Exposure Level	CEQ	Council on Environmental Quality	NRCS	
CERFACommunity Environmental Response Facilitation Act°Fdegrees FahrenheitCFRCode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDoDDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and AlternativesPM10particulate matter less than or equal to 2.5 micrometers in diameterEAenvironmental impact Analysis Processmicrometers in diameterEISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and CommunityPSDPrevention of Significant Deterioration Right-to-Know ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFESOPFederal IPurgency Management AgencySELSound Exposure LevelFESOPFederal IPurgency Management AgencySFsquare feet	CERCLA	Comprehensive Environmental Response,	NRHP	National Register of Historic Places
Facilitation ActOWSoil/water separatorCFRCode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDoDDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action andPM2.5particulate matter less than or equal to 2.5Alternativesmicrometers in diametermicrometers in diameterEAenvironmental assessmentPM10particulate matter less than or equal to 10EISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and CommunityPSDPrevention of Significant DeteriorationRight-to-Know ActRCRAResource Conservation and Recovery ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments andFESOPFederal Emergency Management AgencyReauthorization ActFESOPFederal Entergency Management AgencySELSound Exposure LevelPermitSFsquare feet				
CFRCode of Federal RegulationsP.L.Public LawCOcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDoDDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and AlternativesPM_2.5micrometers in diameterEAenvironmental assessmentPM10particulate matter less than or equal to 2.5 micrometers in diameterEISenvironmental Impact Analysis Processmicrometers in diameterEOExecutive Orderppmparts per millionEPCRAEmergency Planning and Community Right-to-Know ActPSDPrevention of Significant Deterioration Right-to-Know ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFESOPFederal Emergency Management AgencySELSound Exposure Level SF	CERFA	Community Environmental Response	°F	
COcarbon monoxidePA/SIPreliminary Site Assessment/SiteCWAClean Water ActInvestigationdBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDoDDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and AlternativesPM2.5particulate matter less than or equal to 2.5 micrometers in diameterEAenvironmental assessmentPM10particulate matter less than or equal to 10 micrometers in diameterEISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and Community Right-to-Know ActPSDPrevention of Significant Deterioration Right-to-Know ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFESOPFederal Emergency Management Agency PermitSFsquare feet				
CWAClean Water ActInvestigationdBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDoDDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and AlternativesPM_2.5particulate matter less than or equal to 2.5 micrometers in diameterEAenvironmental assessmentPM_10particulate matter less than or equal to 10 micrometers in diameterEISenvironmental Impact Analysis Processmicrometers in diameterEISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and CommunityPSDPrevention of Significant Deterioration Right-to-Know ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFESOPFederal Emergency Management AgencySELSound Exposure Level square feet				
dBdecibelPAAprimary assigned aircraftdBAA-weighted decibelPbleadDoDDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and AlternativesPM2.5particulate matter less than or equal to 2.5 micrometers in diameterEAenvironmental assessmentPM10particulate matter less than or equal to 10 micrometers in diameterEIAenvironmental Impact Analysis Processmicrometers in diameterEISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and Community Right-to-Know ActPSDPrevention of Significant Deterioration Right-to-Know ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFEMAFederal Emergency Management AgencySELSound Exposure Level square feet			PA/SI	-
dBAA-weighted decibelPbleadDoDDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and AlternativesPM2.5particulate matter less than or equal to 2.5 micrometers in diameterEAenvironmental assessmentPM10particulate matter less than or equal to 10EIAPEnvironmental Impact Analysis Processmicrometers in diameterEISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and Community Right-to-Know ActPSDPrevention of Significant Deterioration Right-to-Know ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFEMAFederal Emergency Management Agency PermitSELSound Exposure Level square feet				
DoDDepartment of DefensePCIper capita incomeDOPAADescription of Proposed Action and AlternativesPM2.5particulate matter less than or equal to 2.5 micrometers in diameterEAenvironmental assessmentPM10particulate matter less than or equal to 10 micrometers in diameterEIAenvironmental Impact Analysis Processmicrometers in diameterEISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and Community Right-to-Know ActPSDPrevention of Significant Deterioration Right-to-Know ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFEMAFederal Emergency Management Agency PermitSELSound Exposure Level square feet				primary assigned aircraft
DOPAADescription of Proposed Action and AlternativesPMparticulate matter less than or equal to 2.5 micrometers in diameterEAenvironmental assessmentPM10particulate matter less than or equal to 10 micrometers in diameterEIAPEnvironmental Impact Analysis Processmicrometers in diameterEISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and Community Right-to-Know ActPSDPrevention of Significant Deterioration Region of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFEMAFederal Emergency Management Agency PermitSELSound Exposure Level square feet				
Alternativesmicrometers in diameterEAenvironmental assessmentPM10particulate matter less than or equal to 10EIAPEnvironmental Impact Analysis Processmicrometers in diameterEISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and Community Right-to-Know ActPSDPrevention of Significant Deterioration Region of InfluenceERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFESOPFederal Emergency Management Agency PermitSELSound Exposure Level SF				
EAenvironmental assessmentPM10particulate matter less than or equal to 10EIAPEnvironmental Impact Analysis Processmicrometers in diameterEISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and CommunityPSDPrevention of Significant DeteriorationRight-to-Know ActRCRAResource Conservation and Recovery ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments andFEMAFederal Emergency Management AgencyReauthorization ActFESOPFederally Enforceable State OperatingSELSound Exposure LevelPermitSFsquare feet	DOPAA		PM _{2.5}	
EIAPEnvironmental Impact Analysis Processmicrometers in diameterEISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and CommunityPSDPrevention of Significant DeteriorationRight-to-Know ActRCRAResource Conservation and Recovery ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments andFEMAFederal Emergency Management AgencyReauthorization ActFESOPFederally Enforceable State OperatingSELSound Exposure LevelPermitSFsquare feet				
EISenvironmental impact statementPOLpetroleum, oil, and lubricantsEOExecutive Orderppmparts per millionEPCRAEmergency Planning and Community Right-to-Know ActPSDPrevention of Significant Deterioration RCRAERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFEMAFederal Emergency Management AgencyReauthorization ActFESOPFederally Enforceable State Operating PermitSELSound Exposure Level square feet			PM_{10}	
EOExecutive Orderppmparts per millionEPCRAEmergency Planning and Community Right-to-Know ActPSDPrevention of Significant Deterioration RCRAERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFEMAFederal Emergency Management AgencyReauthorization ActFESOPFederally Enforceable State Operating PermitSELSound Exposure Level square feet		· ·		
EPCRAEmergency Planning and Community Right-to-Know ActPSDPrevention of Significant Deterioration RCRAERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFEMAFederal Emergency Management AgencyReauthorization ActFESOPFederally Enforceable State Operating PermitSELSound Exposure Level square feet				
Right-to-Know ActRCRAResource Conservation and Recovery ActERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments andFEMAFederal Emergency Management AgencyReauthorization ActFESOPFederally Enforceable State OperatingSELSound Exposure LevelPermitSFsquare feet				
ERPEnvironmental Restoration ProgramROIRegion of InfluenceESAEndangered Species ActSARASuperfund Amendments andFEMAFederal Emergency Management AgencyReauthorization ActFESOPFederally Enforceable State OperatingSELSound Exposure LevelPermitSFsquare feet	EPCRA			
ESAEndangered Species ActSARASuperfund Amendments and Reauthorization ActFEMAFederal Emergency Management AgencyReauthorization ActFESOPFederally Enforceable State Operating PermitSELSound Exposure Level square feet				
FEMAFederal Emergency Management AgencyReauthorization ActFESOPFederally Enforceable State Operating PermitSEL SFSound Exposure Level square feet				
FESOP Federally Enforceable State Operating SEL Sound Exposure Level Permit SF square feet			SARA	
Permit SF square feet			() FI	
	FESOP			
FFCAFederal Facility Compliance ActSHPOState Historic Preservation Officer	FEG			
	FFCA	Federal Facility Compliance Act	SHPO	State Historic Preservation Officer

COVER SHEET

ENVIRONMENTAL ASSESSMENT FOR PROPOSED CONSTRUCTION, DEMOLITION, AND OTHER ACTIONS AT THE 126TH AIR REFUELING WING AT SCOTT AIR FORCE BASE, ILLINOIS

Responsible Agencies: United States Air Force (USAF), Headquarters Air Mobility Command (AMC), Scott Air Force Base (AFB), Illinois, and 375th (375 AW) Airlift Wing, Scott AFB, Illinois.

Affected Location: Scott AFB, St. Clair County, Illinois.

Proposed Action: Implementation of construction, demolition, and other actions in support of mission of the 126th Air Refueling Wing (126 ARW).

Report Designation: Environmental Assessment (EA).

Written comments and inquiries regarding this document should be directed to: Major Kevin Jacobs, 126 ARW, Illinois Air National Guard, 146 Air Guard Way, Scott Air Force Base, Illinois 62225-5503

Abstract: Under the Proposed Action, the 126 ARW proposes to implement construction projects associated with their five-year construction program that would include eight specific projects: construction of some new facilities, modifications to some existing facilities, and demolition of one facility. Under the Alternative Action, the eight construction and demolition projects associated with the Proposed Action would still occur, in addition to three additional construction projects. The 126 ARW would also release their eight assigned KC-135E model aircraft and would acquire 12 KC-135R model aircraft. The number of aircraft operations would increase by approximately 50 percent as a result of the four additional assigned aircraft. There would also be an increase of 27 personnel (firefighters) as a result of this alternative.

During the preparation of this EA, no significant impacts as a result of implementation of either the Proposed Action or the Alternative Action have been identified, and therefore, an Environmental Impact Statement is not required, and will not be prepared. Resources considered in the EA include earth resources, water resources, biological resources, air quality, noise, land use, socioeconomics and environmental justice, cultural resources, safety, solid and hazardous materials and wastes, and infrastructure. The EA was made available to the public for comment for a 30-day period (February 5 though March 10). No public comments were received. The EA was also coordinated with local, state, and federal agencies at the same time. There were only two agency responses (United States Fish and Wildlife Service [USFWS] and State Historic Preservation Office [SHPO]). Both were in concurrence with the findings in the EA.

FINDING OF NO SIGNIFICANT IMPACT FOR PROPOSED CONSTRUCTION, DEMOLITION, AND OTHER ACTIONS AT THE 126TH AIR REFUELING WING ILLINOIS AIR NATIONAL GUARD SCOTT AIR FORCE BASE, ILLINOIS

PURPOSE: The purpose of the Proposed Action is to provide the 126th Air Refueling Wing (126 ARW) with properly sized and configured facilities that are required to effectively accomplish their mission. The Proposed Action is necessary to reconfigure facilities to accommodate the continuously evolving mission of the 126 ARW. The new facilities would enhance the 126 ARW's ability to maintain a level of wartime readiness necessary to support the mission.

PROPOSED ACTION: Under the Proposed Action, the 126 ARW will implement construction projects in support of their air refueling mission that are planned for the next five-year period. Associated with this construction activity will be demolition of one facility that is obsolete, deteriorated, and in the footprint of one of the proposed facilities.

ALTERNATIVE ACTION: Under the Alternative Action, all construction and demolition associated with the Proposed Action would occur in addition to three other construction projects. Additionally, the 126 ARW would release their 8 Primary Assigned Aircraft (PAA) KC-135E model aircraft and would acquire 12 PAA KC-135R model aircraft. The number of aircraft operations would increase by approximately 50 percent as a result of the four additional PAA. In addition, there would be an increase of 27 personnel at the installation.

NO ACTION ALTERNATIVE: Under the No Action Alternative, the 126 ARW would maintain their existing facilities and would not build any of the new facilities proposed. In general, the No Action Alternative would require that the 126 ARW continue to operate under unnecessarily inefficient conditions, and that anti-terrorism/force protection (AT/FP) requirements continue to be unmet. Deficiencies could continue to impair the 126 ARW's ability to successfully conduct their mission and to maintain wartime readiness and training.

SUMMARY OF FINDINGS:

Potential impacts associated with the Proposed Action and the Alternative Action have been assessed with regard to the following environmental resource areas:

Earth Resources. There will be 1.80 acres of surface disturbance over the course of the construction program associated with the Proposed Action (and 3.48 acres under the Alternative

Action). The grading of existing soil and placement of structural fill for new facilities will not substantially alter existing soil conditions at the installation because, to a large extent, the construction described above is planned for areas where surface disturbance has previously occurred. Best Management Practices (BMPs) will be used to limit soil movement, stabilize runoff, and control sedimentation. Impacts to earth resources are expected to be minimal.

Water Resources. There will be approximately 1.75 acres of new impervious surface that will result under the Proposed Action (and 3.43 acres under the Alternative Action). The 126 ARW will update their Storm Water Pollution Prevention Plan (SWPPP) to include these projects and will obtain, as appropriate, coverage under a National Pollutant Discharge Elimination System (NPDES) Construction Storm Water permit from the Illinois Environmental Protection Agency (Illinois EPA). Adherence to the requirements of the permit will include implementation of BMPs to minimize the potential for exposed soils or other contaminants from construction activities to reach nearby surface waters. It is expected that impacts to water resources will be minimal.

Biological Resources. In general, projects described under the Proposed Action and the Alternative Action are located at sites that are highly altered by man. The potential for any federally listed species to occur on the Base is low, with the exception of the Indiana Bat. Recent correspondence with the Illinois Department of Natural Resources (IDNR) revealed that no records exist of any state-listed threatened or endangered species on or adjacent to the Base. The 126 ARW will coordinate, as necessary, with the United States Fish and Wildlife Service (USFWS) prior to implementation of construction activities to ensure that impacts to sensitive species do not occur. Impacts to biological resources are expected to be minimal.

Air Quality. In general, combustive and fugitive dust emissions from proposed construction activities under the Proposed Action or the Alternative Action will produce localized, elevated air pollutant concentrations that will occur for a short duration and will not result in any long-term impacts on the air quality of St. Clair County or Air Quality Control Region (AQCR) 70. Impacts to air quality in the County and the AQCR are expected to be minimal.

Noise. Construction noise emanating off-site as a result of the Proposed Action will likely be noticeable in the immediate construction site vicinity, but is not expected to create adverse impacts. In addition, noise associated with the aircraft conversion and the associated construction activities described under the Alternative Action will similarly not be expected to create adverse impacts because the noise contours actually shrink under this alternative. The acoustic environment on and near the airfield property is expected to remain relatively unchanged from existing conditions. Impacts from noise are expected to be minimal.

2

Land Use/Visual Resources. The proposed construction and demolition projects described under the Proposed Action and the Alternative Action are expected to enhance overall installation planning and compatibility of functions on the 126 ARW installation in general. Some existing incompatibilities will be corrected. Visual resources are generally not expected to be impacted. Impacts to land use and visual resources are expected to be minimal.

Socioeconomics/Environmental Justice. There are no substantial long-term changes in population and/or employment as a result of implementation of the Proposed Action. Although there is an anticipated increase of 27 personnel under the Alternative Action, this is a minor increase and will not result in any appreciable impact. Additionally, the projects described under the alternatives are not expected to create adverse environmental or health effects, and therefore no disproportionately high or adverse impacts to minority, low-income, or youth populations are expected. Impacts to socioeconomics and environmental justice are expected to be minimal.

Cultural Resources. Activities associated with the Proposed Action or the Alternative Action are not expected to impact archaeological or traditional resources. Demolition of Building 5542 will be coordinated with the Illinois State Historic Preservation Officer (SHPO) to determine its eligibility to the National Register of Historic Places (NRHP) prior to demolition. Impacts to cultural resources are expected to be minimal.

Safety. Implementation of the Proposed Action or the Alternative Action does involve ground activities that may expose workers performing the required site preparation, grading, and building construction to some risk. Strict adherence to all applicable occupational safety requirements will minimize the relatively low risk associated with these construction activities. All projects have been sited outside any quantity-distance arcs, as appropriate. Additionally, the proposed projects will include measures to enhance and correct AT/FP shortfalls as part of the facility designs. Impacts to safety are expected to be minimal.

Hazardous Materials and Waste Management. The proposed construction and demolition projects associated with the Proposed Action or the Alternative Action will generate construction and demolition waste that will be recycled and/or taken to a local demolition landfill, as appropriate. There are no capacity issues with the existing landfills. Hazardous materials and wastes will be handled, stored, and disposed of in accordance with applicable regulations. Any asbestos-containing material (ACM), lead-based paint (LBP), or contaminated soils associated with Environmental Restoration Program (ERP) sites will be removed and disposed of per applicable regulations. Impacts as a result of hazardous materials and waste management are expected to be minimal.

Infrastructure. The proposed construction and demolition projects associated with the Proposed Action or the Alternative Action will result in some temporary interruption of utility services and

minor hindrance of transportation and circulation during construction activities. These impacts will be temporary, occurring only for the duration of the construction period. In general, infrastructure at the 126 ARW installation will improve under these actions. Impacts to infrastructure are expected to be minimal.

PUBLIC INVOLVEMENT: The National Environmental Policy Act (NEPA), 40 Code of Federal Regulations (CFR) 1500-1508, and 32 CFR 989 require public review of the Environmental Assessment (EA) before approval of the Finding of No Significant Impact (FONSI) and implementation of the Proposed Action or the Alternative Action. A notice of availability for public review of the Draft EA was published in the Belleville News-Democrat on Sunday, February 5 and Sunday, February 19, 2006. The Draft EA was available for public review from February 5 through March 10, 2006. No public comments were received. The EA was also coordinated with local, state, and federal agencies during the same time period. Two agency responses (USFWS and SHPO) were received. Both were in concurrence with the findings in the EA.

FINDING OF NO SIGNIFICANT IMPACT (FONSI): Based on my review of the facts and analysis in the Environmental Assessment, I conclude that neither the Proposed Action nor the Alternative Action will have a significant impact either by itself or considering cumulative impacts. Accordingly, the requirements of NEPA, the Council on Environmental Quality (CEQ), and 32 CFR 989, et seq. have been fulfilled, and an Environmental Impact Statement (EIS) is not necessary and will not be prepared.

RAYMOND J. KOTTMAN, Colonel, USAF Commander

IS MAY OC

Date

ENVIRONMENTAL ASSESSMENT

PROPOSED CONSTRUCTION, DEMOLITION, AND OTHER ACTIONS AT THE 126TH AIR REFUELING WING ILLINOIS AIR NATIONAL GUARD

SCOTT AIR FORCE BASE, ILLINOIS

AIR NATIONAL GUARD ENVIRONMENTAL DIVISION

TABLE OF CONTENTS

ACRONYMS AND ABBREVIATIONS..... INSIDE FRONT AND BACK COVER

1.0	PUR	POSE A	ND NEEI)	1-1
	1.1	INTR	ODUCTIO	N	1-1
	1.2	PURP	OSE AND	NEED FOR THE ACTION	1-1
	1.3	LOCA	TION AN	D DESCRIPTION OF THE 126 ARW	1-4
	1.4			KEY ENVIRONMENTAL REQUIREMENTS	
		1.4.1	NATIONA	L ENVIRONMENTAL POLICY ACT	1-6
		1.4.2	Endange	ERED SPECIES ACT	1-6
		1.4.3	CLEAN A	IR ACT	1-6
		1.4.4	WATER R	ESOURCES REGULATORY REQUIREMENTS	1-7
		1.4.5	CULTURA	L RESOURCES REGULATORY REQUIREMENTS	1-7
		1.4.6	OTHER R	EGULATORY REQUIREMENTS	1-8
		1.4.7	ENVIRON	MENTAL COORDINATION	1-8
2.0				HE PROPOSED ACTION AND ALTERNATIVES	
	2.1			N	
	2.2			TION	
		2.2.1		ICT PUMPHOUSE SPILL CONTAINMENT	
		2.2.2	-	N TO MID-FIELD FIRE STATION	
		2.2.3		ICT PAVEMENTS AND GROUNDS FACILITY	
		2.2.4		ICT NEW SECURITY FORCES FACILITY	
		2.2.5		N/ALTERATION TO THE DEPLOYMENT PROCESSING CENTER	
		2.2.6		N/ALTERATION TO THE COMMUNICATION FACILITY	
		2.2.7		ICT TANKER AIRLIFT COMMAND TRAINING CENTER	
		2.2.8		DEMOLITION	2-6
		2.2.9		Y OF SURFACE DISTURBANCE ASSOCIATED WITH THE	2 (
	2.2			DACTION	2-6
	2.3			E ACTION #1: IMPLEMENTATION OF PROPOSED	
				ON AND DEMOLITION IN ADDITION TO AIRCRAFT D ASSOCIATED ACTIONS	20
		2.3.1		ICT AIRCRAFT APRON EXPANSION	
		2.3.2		N TO MID-FIELD FIRE STATION (FURTHER ADDITION)	
		2.3.3		N OF HYDRANT PIT AND BLAST DEFLECTORS ON THE PARKING	
		2.3.4		Y OF SURFACE DISTURBANCE ASSOCIATED WITH THE	
			ALTERNA	TIVE ACTION	2-10
	2.4	NO A	CTION AI	LTERNATIVE	2-11
3.0	EXIS	STING (CONDITI	ONS	3-1
	3.1	EART	H RESOU	IRCES	3-1
		3.1.1	DEFINITIO	ON OF THE RESOURCE	3-1
		3.1.2	Existing	CONDITIONS	3-1
			3.1.2.1	Geology	3-1
			3.1.2.2	Soils	3-2

		3.1.2.3	Topography	
3.2	WAT	ER RESO	URCES	
	3.2.1	Definiti	ON OF THE RESOURCE	
	3.2.2	Existing	G CONDITIONS	
		3.2.2.1	Surface Water	
		3.2.2.2	Floodplains	
		3.2.2.3	Groundwater	
3.3	BIOL	OGICAL	RESOURCES	
	3.3.1		ON OF THE RESOURCE	
	3.3.2		G CONDITIONS	
		3.3.2.1	Vegetation	
		3.3.2.2	Wildlife	
		3.3.2.3		
			Wetlands and Other Aquatic Habitat	
3.4	AIR (·····	
		 	ON OF THE RESOURCE	
			G CONDITIONS	
3.5				
	3.5.1	Definiti	ON OF RESOURCE	
		3.5.1.1	Maximum Sound Level	
		3.5.1.2	Sound Exposure Level	
		3.5.1.3	Time-Averaged Cumulative Noise Metrics	
	3.5.2	Existing	G CONDITIONS	
		3.5.2.1	Aircraft Activity	
		3.5.2.2	Ground-Based Activity	
3.6	LANI	DUSE AN	D VISUAL RESOURCES	
	3.6.1	Definiti	ON OF RESOURCE	
		3.6.1.1	Land Use	
		3.6.1.2	Visual Resources	
	3.6.2	Existing	G CONDITIONS	
		3.6.2.1	Land Use	
		3.6.2.2	Visual Resources	
3.7	SOCI	OECONO	MICS AND ENVIRONMENTAL JUSTICE	
	3.7.1	Definiti	ON OF THE RESOURCE	
	3.7.2	Existing	G CONDITIONS	
		3.7.2.1	Population and Employment	
		3.7.2.2	Environmental Justice	
3.8	CULT	URAL RI	ESOURCES	
	3.8.1	Definiti	ON OF THE RESOURCE	
	3.8.2	Existing	G CONDITIONS	
		3.8.2.1	Historical Setting	
		3.8.2.2	Identified Cultural Resources	
3.9	SAFE	ΤΥ		
	3.9.1	Definiti	ON OF THE RESOURCE	
	3.9.2	EXISTING	G CONDITIONS	
		3.9.2.1	Ground Safety	

			3.9.2.2	Explosives Safety	
			3.9.2.3	Anti-Terrorism/Force Protection	
			3.9.2.4	Flight Safety	
	3.10	SOLII	D AND HA	AZARDOUS MATERIALS AND WASTE	
		3.10.1	DEFINITI	ON OF THE R ESOURCE	
				CONDITIONS	
			3.10.2.1	Hazardous Materials and Petroleum Products	
			3.10.2.2		
			3.10.2.3	Environmental Restoration Program Sites	
			3.10.2.4		
	3.11	INFR/	ASTRUCT	URE	
		3.11.1	DEFINITI	ON OF THE R ESOURCE	
		3.11.2	Existing	CONDITIONS	
			3.11.2.1	Transportation and Circulation	
			3.11.2.2	Utilities	
4.0	ENV			CONSEQUENCES	
	4.1	EART	TH RESOU	JRCES	
		4.1.1		OLOGY	
		4.1.2			
			4.1.2.1	Proposed Action	
			4.1.2.2	Alternative Action	
			4.1.2.3	No Action Alternative	
	4.2			JRCES	
		4.2.1		OLOGY	
		4.2.2			
			4.2.2.1	Proposed Action	
			4.2.2.2	Alternative Action	
			4.2.2.3	No Action Alternative	
	4.3			RESOURCES	
		4.3.1		OLOGY	
		4.3.2			
				Proposed Action	
				Alternative Action	
				No Action Alternative	
	4.4		~		
				OLOGY	
		4.4.2			
			4.4.2.1	Proposed Action	
			4.4.2.2	Alternative Action	
			4.4.2.3	No Action Alternative	
	4.5				
		4.5.1		OLOGY	
		4.5.2			
			4.5.2.1	Proposed Action	
			4.5.2.2	Alternative Action	
			4.5.2.3	No Action Alternative	

4.6	LANI	D USE AN	D VISUAL RESOURCES	4-18
	4.6.1	Method	OLOGY	4-18
	4.6.2	IMPACTS.		4-19
		4.6.2.1	Proposed Action	4-19
		4.6.2.2	Alternative Action	4-20
		4.6.2.3	No Action Alternative	4-20
4.7	SOCI	OECONO	MICS AND ENVIRONMENTAL JUSTICE	4-21
	4.7.1	Method	OLOGY	4-21
	4.7.2	IMPACTS.		4-21
		4.7.2.1	Proposed Action	4-21
		4.7.2.2	Alternative Action	4-22
		4.7.2.3	No Action Alternative	4-22
4.8	CULT	TURAL RE	ESOURCES	4-22
	4.8.1	Method	OLOGY	4-22
	4.8.2	IMPACTS.		4-23
		4.8.2.1	Proposed Action	4-23
		4.8.2.2	Alternative Action	4-24
		4.8.2.3	No Action Alternative	4-25
4.9	SAFE	ТΥ		4-25
	4.9.1	Method	OLOGY	4-25
	4.9.2	IMPACTS.		4-25
		4.9.2.1	Proposed Action	4-25
		4.9.2.2	Alternative Action	4-26
		4.9.2.3	No Action Alternative	4-27
4.10	SOLI	D AND HA	AZARDOUS MATERIALS AND WASTE	4-27
	4.10.1	METHOD	OLOGY	4-27
	4.10.2	2 IMPACTS.		4-28
		4.10.2.1	Proposed Action	4-28
		4.10.2.2	Alternative Action	
		4.10.2.3	No Action Alternative	4-32
4.11	INFR	ASTRUCT	'URE	4-32
	4.11.1	METHOD	OLOGY	4-32
	4.11.2	2 IMPACTS.		4-33
		4.11.2.1	Proposed Action	4-33
		4.11.2.2	Alternative Action	4-33
		4.11.2.3	No Action Alternative	4-34
			ACTS AND IRREVERSIBLE AND IRRETRIEVABLE	5 1
			RESOURCES IMPACTS	
5.1				
			AND REASONABLY FORESEEABLE ACTIONS IN THE REGION . S OF CUMULATIVE IMPACTS	
5.0				3-3
5.2			E AND IRRETRIEVABLE COMMITMENT OF	E (
	KESC	UKCES		3-0

5.0

6.0	SPE		
	6.1	SPECIAL OPERATING PROCEDURES	
	6.2	MITIGATION	
7.0	PER	SONS AND AGENCIES CONTACTED	
8.0	LIST	Γ OF PREPARERS	
9.0	REF	ERENCES	

APPENDIX A INTERAGENCY AND INTERGOVERNMENTAL COORDINATION FOR ENVIRONMENTAL PLANNING (IICEP)

TABLES

2.2-1	List of Projects Included Under the Proposed Action	2-3
2.2-3	Pavements and Building Footprints – Net New Impervious Surface Under the	
	Proposed Action	2-7
2.3-1	Pavements and Building Footprints – Net New Impervious Surface Under the	
	Alternative Action	2-10
3.3-1	Threatened and Endangered Species Documented or Likely to Occur in St. Clair	
	County, with Assessment of Potential for Occurrence on the Installation.	3-11
3.4-1	National and Illinois Ambient Air Quality Standards	3-14
3.4-2	Baseline Emissions at Scott AFB, calendar year 2004	3-18
3.4.3	Air Emissions Inventory St. Clair County, Illinois, and AQCR 70 Calendar	
	Year 1999	3-19
3.5-1	Representative Maximum Sound Levels	3-21
3.5-2	Representative Sound Exposure Levels	3-21
3.5-3	Percentage of Population Highly Annoyed By Elevated Noise Levels	3-23
3.5-4	Average Daily Operations at Scott AFB/MidAmerica Airport	
3.5-5	Land Area Exposed To Indicated Sound Levels	3-25
3.6-1	Existing and Recommended Land Uses Surrounding Scott AFB	3-29
3.7-1	Population Changes in the Region	
3.7-2	Per Capita Income	3-32
3.7-3	Profile of Demographic Characteristics, Year 2000	3-33
3.7-4	Persons under Age 18 in the ROI in the Year 2000	3-33
3.7-5	Individuals in Poverty, Reported in the Year 2000	3-34
3.8-1	Buildings in ROI of 126 ARW Projects	3-39
3.10-1	Composition of Recycled Materials in Tons	3-53
4.4-1	Construction Emissions – Proposed Action	4-10
4.4-2	Construction Emissions Under the Alternative Action	4-11
4.4-3	Estimated Change in Aircraft Emissions at Scott AFB, as a Result of the	
	Alternative Action	4-12

4.4-4	Additional Indirect Emissions from Additional POV Commuting due to the	
	Alternative Action	
4.5-1	Typical Equipment Sound Levels	
4.5-2	Average Daily Operations at Scott and MidAmerica Airport with Aircraft	
	Modifications	
4.5-3	Land Area Exposed to Indicated Sound Levels Under Alternative Action	
4.8-1	Buildings Potentially Affected Under Action Alternatives	
5 1-1	Ongoing and Proposed Projects in the ROI	5-2

FIGURES

1.3-1	Regional Location of 126 ARW Installation, Scott Air Force Base, Illinois	1-5
2.1-1	Existing Facilities at the 126 ARW, Scott Air Force Base, Illinois	2-2
2.2-1	Projects Under the Proposed Action at the 126 ARW, Scott Air Force Base, Illinois	2-4
2.3-1	Projects Under Alternative Action #1 at the 126 ARW, Scott Air Force Base, Illinois	2-9
3.1-1	Soil Mapping Units on the 126 ARW Installation at Scott AFB, Illinois	3-3
3.5-1	Scott AFB/MidAmerica Airport Baseline Noise Contours	3-26
3.10-1	ERP Sites and AOCs on the 126 ARW Installation at Scott AFB, Illinois	3-49
4.2-1	Floodplains in Relation to Proposed Construction Activities at the 126 ARW	
	Installation at Scott AFB, Illinois	4-5
4.5-1	Scott AFB and MidAmerica Airport Noise Contours	4-17
4.10-1	ERP Sites and AOCs in Relation to Proposed Construction Activities on the	
	126 ARW Installation at Scott AFB, Illinois	4-31

1.0 PURPOSE AND NEED

1.1 INTRODUCTION

The 126th Air Refueling Wing (126 ARW) of the Illinois Air National Guard (ILANG) is a tenant at Scott Air Force Base (AFB), Illinois. The 126 ARW is tasked with providing air-to-air refueling and airlift capabilities for Department of Defense (DoD) assets worldwide under the Air Mobility Command (AMC). The Wing also supports state emergency missions. The 126 ARW currently has eight KC-135E refueler aircraft (eight primary assigned aircraft [PAA], no backup aircraft inventory [BAI]), and also provides support to numerous transient aircraft.

The 126 ARW proposes to implement construction projects associated with their five-year construction program that would include construction of several new facilities, modifications to some existing facilities, and demolition of one facility, including:

- Construct Pumphouse Spill Containment
- Addition to the Mid-Field Fire Station
- Construct New Pavements and Grounds Facility
- Construct New Security Forces Facility
- Addition/Alteration to the Deployment Processing Facility
- Addition/Alteration to the Communication Facility
- Construct New Tanker Airlift Command Training Center
- Demolish Parking Shed (Building 5042)

In accordance with the National Environmental Policy Act (NEPA) of 1969 (42 United States Code [USC] 4321-4347), Council on Environmental Quality (CEQ) Regulations for Implementing the Procedural Provisions of NEPA (40 Code of Federal Regulations [CFR] §§ 1500-1508), and 32 CFR 989, et seq., *Environmental Impact Analysis Process* (formerly promulgated as Air Force Instruction [AFI] 32-7061), the 126 ARW has prepared this Environmental Assessment (EA) that considers the potential consequences to the human and natural environment that may result from implementation of these projects.

1.2 PURPOSE AND NEED FOR THE ACTION

The purpose of the Proposed Action is to provide the 126 ARW with properly sized and configured facilities that are required to effectively accomplish their mission. The Proposed

Action is necessary to reconfigure facilities to accommodate the continuously evolving mission of the 126 ARW. The new facilities would enhance the 126 ARW's ability to maintain a level of wartime readiness necessary to support the mission. The following paragraphs describe the purpose and need for each of the listed projects.

Construct Pumphouse Spill Containment. The pumphouse must be constructed to contain all JP-8 fuel releases within the area. All equipment installed within any hazardous classified area must be constructed in accordance with the National Electrical Code (NEC). The existing pumphouse (Facility 5038) was not originally constructed in accordance with the Type III standard for hydrant fueling systems, as established by the DoD Fuel Facility Engineering Panel, as well as environmental guidelines, and therefore must be reconfigured to conform to the NEC Type III standard and environmental guidelines. All storm drainage from the site is collected and piped to an adjacent drainage ditch, which eventually flows off-Base. Should a spill occur at the pumphouse, it would flow directly to the drainage ditch, presenting substantial environmental concerns. Should fuel be inadvertently released into the existing spill containment area, it would drain into the existing electrical manhole and could lead to a potential fire hazard. The pumphouse must have a curb around the perimeter of the facility to provide containment in the event of a spill. The existing pumphouse slab is flush with the surrounding grade, allowing any spills within the facility to escape through the doorways.

Addition to Mid-Field Fire Station. The existing Mid-Field Fire Station (Building 3901) is operated by the host unit at Scott AFB, the 375th Airlift Wing (375 AW). As such, the 126 ARW has insufficient facilities for fire protection training. To meet the mission requirements for the 126 ARW, a facility that has the necessary office/administrative space, training facilities, and equipment storage is required. The 126 ARW currently conducts training exercises out of Fire Station #3 (Mid-Field Fire Station), during unit training assemblies (UTA's), which occur during one weekend per month; however, this is a temporary situation and permanent training facilities must be provided. These temporary measures are not conducive to an effective and efficient training environment. The 126 ARW will not be fully mission-ready until fire training facilities are available.

Construct Pavements and Grounds Facility. The existing pavements and grounds facility is comprised of 1950's era carports (Building 5542) that are dilapidated, rodent infested, and do not have adequate dimensions to support the necessary equipment. Additionally, there are numerous health, safety, and fire code deficiencies. The 126 ARW requires a pavements and grounds facility to support the civil engineering function that is properly sized and equipped for their current mission. Without the new facility, the 126 ARW civil engineering function will continue to operate in inefficient conditions, which will result in higher operating costs.

Construct New Security Forces Facility. The existing Security Forces Facility, which is located in Building 5046, is undersized to support training requirements. The 126 ARW Security Forces are currently allocated only approximately 50 percent of the authorized space required for this function. Currently the facility does not contain a Combat Arms Training Simulator (CATS); it has insufficient space for operations, administration, training, classrooms, and storage. The 126 ARW requires an adequately sized and properly configured facility to accommodate the Security Forces function, which include areas for command, administration, pass and identification production, law enforcement, an arms vault with an area for weapons maintenance, classroom, training and counseling, mobility storage and build-up, and locker rooms. Without sufficient space to conduct training, the Security Forces cannot efficiently and effectively conduct their mission, which includes ensuring that daily security on the installation is maintained. Additionally, without sufficient storage area, valuable mobility assets would continue to be exposed to weathering.

Addition/Alteration to the Deployment Processing Center. The existing Deployment Processing Center is located in Building 5028. A deployment processing center for mobility deployment currently does not exist. As a result of Air Expeditionary Force (AEF) requirements, the Air National Guard (ANG) is required to deploy at least every 30 months. This is in addition to any non-AEF flying requirements, contingencies, wars, and other mobilization deployments. Mobility Processing for the 126 ARW is currently being accomplished ad hoc in an available hangar. The ad hoc facilities are not compatible with a function that must be accomplished in a streamlined manner. Additionally, the use of a maintenance hangar for deployment processing detracts from the simultaneous maintenance activities that must occur so that forces can deploy. The 126 ARW requires a facility that can efficiently accommodate receiving and processing of personnel and their baggage; baggage pallet buildup; counseling; briefing; restrooms and a vending area. Adequate ingress and egress areas must also be provided to accommodate passenger buses, cargo trucks, and the handling of mobility and personnel bags. Without a properly designed Mobility Processing Center, the 126 ARW will be ineffective and inefficient during this extremely time-sensitive task.

Addition/Alteration to the Communication Facility. The space allocated for the Communications function of the 126 ARW is currently severely undersized, and the function is dispersed into three separate facilities to temporarily accommodate the space requirements. This results in an inefficient use of time and resources. Under the Proposed Action, the function would be consolidated into Building 5010 to create a more efficient environment for this function. The existing facilities do no have the necessary ventilation required for communications equipment, and therefore a temporary air conditioning unit is being used to keep the equipment properly ventilated. This results in a security risk to the equipment. Without a consolidated

communications function, the 126 ARW will continue to operate in an inefficient environment that could negatively impact command and control functions at the installation.

Construct Tanker Airlift Command Training Center. The United States Air Force (USAF) currently does not have a Tanker Airlift Command Training Center (TACTC) and basically relies on Tanker Airlift Command Center (TACC) training occurring at existing TACC locations (Scott AFB, MacDill AFB, and Travis AFB). The ANG has identified a need for a TACTC to provide training for TACCs. The 126 ARW TACC squadron must currently rely on availability at the Scott AFB TACC for training. The new facility would be used to provide TACC training USAF-wide. The facility would also serve as an alternate TACC facility to the Scott AFB facility, on an as-needed basis. On-site personnel are currently relocated to off-site facilities in California and Florida when the Scott AFB TACC becomes unavailable. The new facility would eliminate this operational down-time due to relocation by functioning as an alternate TACC.

Facility Demolition. Building 5542 would be demolished under the Proposed Action. This building is the existing carport that is used for pavements and grounds storage. It was constructed in 1951 and is dilapidated and functionally inadequate for the function it is performing. Additionally, it lies in the footprint of the proposed TACTC.

1.3 LOCATION AND DESCRIPTION OF THE 126 ARW

Scott AFB is located in St. Clair County, in the southwest portion of Illinois. The Base is approximately 100 miles south of Springfield, the state capital, and 20 miles east of St. Louis, Missouri. The Base comprises 3,589 acres and is located in a predominantly agricultural area. The base is immediately south of Interstate Highway 64 (Figure 1.3-1), near the cities of O'Fallon and Belleville. The 126 ARW is located in the northeast quadrant of Scott AFB, and west of Mid-America Airport, and covers approximately 133 acres.

The primary mission of the 126 ARW is to provide air refueling support to major commands of the USAF, as well as to other United States (U.S.) military forces and the military forces of allied nations. In addition to air-to-air refueling of military aircraft, the unit trains men and women to defend our nation anywhere in the world; to protect the State of Illinois through civil defense and disaster relief; and to serve in the surrounding communities.



Figure 1.3-1. Regional Location of 126 ARW Installation, Scott Air Force Base, Illinois

1.4 SUMMARY OF KEY ENVIRONMENTAL REQUIREMENTS

1.4.1 NATIONAL ENVIRONMENTAL POLICY ACT

NEPA requires federal agencies to take into consideration the potential environmental consequences of proposed actions in their decision-making process. The intent of NEPA is to protect, restore, and enhance the environment through well-informed federal decisions. The CEQ was established under NEPA to implement and oversee federal policy in this process. The CEQ subsequently issued the Regulations for Implementing the Procedural Provisions of the NEPA (40 CFR Sections 1500–1508) (CEQ 1978). These requirements specify that an EA be prepared to:

- Briefly provide sufficient evidence and analysis for determining whether to prepare an Environmental Impact Statement (EIS) or a Finding of No Significant Impact (FONSI).
- Aid in an agency's compliance with NEPA when an EIS is not necessary.
- Facilitate preparation of an EIS when one is necessary.

The activities addressed within this document constitute a federal action and therefore must be assessed in accordance with NEPA. To comply with NEPA, as well as other pertinent environmental requirements, the decision-making process for the Proposed Action includes the development of this EA to address the environmental issues related to the proposed activities. Each federal agency has their own procedures for implementing NEPA, and the USAF implementing procedures are contained in 32 CFR 989 et seq., *Environmental Impact Analysis Process*.

1.4.2 ENDANGERED SPECIES ACT

The Endangered Species Act (ESA) of 1973 (16 USC §§ 1531–1544, as amended) established measures for the protection of plant and animal species that are federally listed as threatened and endangered, and for the conservation of habitats that are critical to the continued existence of those species. Federal agencies must evaluate the effects of their proposed actions through a set of defined procedures, which can include the preparation of a Biological Assessment and can require formal consultation with the United States Fish and Wildlife Service (USFWS) under Section 7 of the Act.

1.4.3 CLEAN AIR ACT

The Clean Air Act (CAA) (42 USC §§ 7401–7671, as amended) provided the authority for the United States Environmental Protection Agency (USEPA) to establish nationwide air quality

standards to protect public health and welfare. Federal standards, known as the National Ambient Air Quality Standards (NAAQS), were developed for six criteria pollutants: ozone (O₃), nitrogen dioxide (NO₂), carbon monoxide (CO), sulfur dioxide (SO₂), particulate matter, and lead (Pb). The Act also requires that each state prepare a State Implementation Plan (SIP) for maintaining and improving air quality and eliminating violations of the NAAQS. Under the CAA Amendments of 1990, federal agencies are required to determine whether their undertakings are in conformance with the applicable SIP and demonstrate that their actions will not cause or contribute to a new violation of the NAAQS; increase the frequency or severity of any existing violation; or delay timely attainment of any standard, emission reduction, or milestone contained in the SIP.

1.4.4 WATER RESOURCES REGULATORY REQUIREMENTS

The Clean Water Act (CWA) of 1977 (33 USC § 1251 *et seq.*) regulates pollutant discharges that could affect aquatic life forms or human health and safety. Section 404 of the CWA, and Executive Order (EO) 11990, *Protection of Wetlands*, regulate development activities in or near streams or wetlands. Section 404 regulates development in streams and wetlands and requires a permit from the United States Army Corps of Engineers (USACE) for dredging and filling in wetlands. EO 11988, *Floodplain Management*, requires federal agencies to take action to reduce the risk of flood damage; minimize the impacts of floods on human safety, health, and welfare; and to restore and preserve the natural and beneficial values served by floodplains. Federal agencies are directed to consider the proximity of their actions to or within floodplains.

1.4.5 CULTURAL RESOURCES REGULATORY REQUIREMENTS

The National Historic Preservation Act (NHPA) of 1966 (16 USC § 470) established the National Register of Historic Places (NRHP) and the Advisory Council on Historic Preservation (ACHP), outlining procedures for the management of cultural resources on federal property. Cultural resources can include archaeological remains, architectural structures, and traditional cultural properties such as ancestral settlements, historic trails, and places where significant historic events occurred. NHPA requires federal agencies to consider potential impacts to cultural resources that are listed, nominated to, or eligible for listing on the NRHP; designated a National Historic Landmark; or valued by modern Native Americans for maintaining their traditional culture. Section 106 of NHPA requires federal agencies to consult with State Historic Preservation Officers (SHPO) if their undertakings might affect such resources. *Protection of Historic and Cultural Properties* (36 CFR 800 [1986]) provided an explicit set of procedures for federal agencies to meet their obligations under the NHPA, which includes inventorying of resources and consultation with SHPO.

The American Indian Religious Freedom Act (AIRFA) (42 USC § 1996) established federal policy to protect and preserve the rights of Native Americans to believe, express, and exercise their traditional religions, including providing access to sacred sites. The Native American Graves Protection and Repatriation Act (NAGPRA) (25 USC §§ 3001–3013) requires consultation with Native American tribes prior to excavation or removal of human remains and certain objects of cultural importance.

1.4.6 OTHER REGULATORY REQUIREMENTS

Additional regulatory legislation that potentially applies to the implementation of this proposal includes guidelines promulgated by EO 12898, *Federal Actions to Address Environmental Justice in Minority Populations and Low-Income Populations*, to ensure that citizens in either of these categories are not disproportionately affected by proposed actions. Additionally, potential health and safety impacts that could disproportionately affect children will be considered under the guidelines established by EO 13045, *Protection of Children from Environmental Health Risks and Safety Risks*.

1.4.7 Environmental Coordination

EO 12372, *Intergovernmental Review of Federal Programs*, requires intergovernmental notifications prior to making any detailed statement of environmental impacts. Through the process of Interagency and Intergovernmental Coordination for Environmental Planning (IICEP), the proponent must notify concerned federal, state, and local agencies and allow them sufficient time to evaluate potential environmental impacts of a Proposed Action. Comments from these agencies are subsequently incorporated into the Environmental Impact Analysis Process (EIAP). A list of federal, state, and local agencies that have been contacted regarding this proposal can be found in Appendix A, along with responses received.

In a recently formulated policy to address EO 13084, *Consultation and Coordination with Indian Tribal Governments*, the DoD has clarified its policy for interacting and working with federally recognized American Indian and Alaska Native governments. Under this policy guidance, proponents must provide timely notice to, and consult with, tribal governments prior to taking any actions that have the potential to affect protected tribal resources, tribal rights, or Indian lands. Tribal input must be solicited early enough in the planning process that it may influence the decision to be made.

2.0 DESCRIPTION OF THE PROPOSED ACTION AND ALTERNATIVES

2.1 INTRODUCTION

The 126 ARW relocated from Chicago, Illinois to Scott AFB in 1999. Nearly all of the 126 ARW facilities were newly constructed and were designed to accommodate the existing mission requirements at that time. Mission requirements for all the military has evolved since that time, and new mission requirements for the 126 ARW require that some of their facilities be updated, while there are other new facilities that are now required. The 126 ARW currently maintains 19 permanent facilities at Scott AFB (Figure 2.1-1). The Proposed Action is to implement construction and demolition projects that would accommodate the continuously evolving mission of the 126 ARW. These projects are described in more detail in Section 2.2.

2.2 PROPOSED ACTION

Under the Proposed Action, the 126 ARW would implement construction projects as described in Table 2.2-1, and a facility demolition as described in Section 2.2.8. These facilities would be sited as shown in Figure 2.2-1.

2.2.1 CONSTRUCT PUMPHOUSE SPILL CONTAINMENT

Under this project, the area on the southeastern and southwestern sides of the existing pumphouse (Facility 5038) would be graded to accommodate new piping and drainage inlets that would be installed to connect the new spill containment area into the existing spill containment system. A curbed slab would be poured to encompass all potential exits from the pumphouse, and to include above ground fuel piping between the pumphouse and the storage tank dikes. Additionally, the top elevation of the existing manholes within the containment area would be raised such that spills could not leak into these manholes. Approximately 3,000 square feet (SF) would be regraded to accommodate the new curb. There would be no additional paved areas as a result of this project.

2.2.2 Addition to Mid-Field Fire Station

An addition of 2,175 SF would be added to the existing Mid-Field Fire Station to accommodate the additional training requirements of the 126 ARW. The addition would primarily accommodate classrooms, lockers, and administrative areas for the 126 ARW. This would include a concrete foundation and floor slab; steel-framed masonry walls and floor structure; and all utilities. There would be no additional parking area required. The exterior would be architecturally compatible with the surrounding Base facilities.



Figure 2.1-1. Existing Facilities at the 126 ARW, Scott Air Force Base, Illinois

r	Table 2.2-1. List of Trojects included Onder the Troposed Action							
Action Item	Project Number	Project Title	Facility and/or Addition Size	Project Description				
1	VDYD042015	Construct Pumphouse Spill Containment	No addition	Grade area surrounding pumphouse, install piping, pour a curbed slab to encompass all exits from the pumphouse.				
2	VDYD992001	Addition to Mid-Field Fire Station	2,175 SF	Addition of 2,175 SF to the existing fire station. Includes classrooms, lockers, and administrative areas.				
3	VDYD049129	Construct Pavements and Grounds Facility	8,000 SF 2,000 SY (18,000 SF) new pavements	Construct an 8,000 SF facility to support Civil Engineering activities associated with maintenance. 2,000 SY of pavements would be required for the storage yard.				
4	VDYD069044	Construct New Security Forces Facility	13,900 SF	Construct a 13,900 SF facility to support Security Forces; to include a combat arms training simulator. No new pavements anticipated.				
5	VDYD059014 VDYD032079	Addition/Alteration to the Deployment Processing Center	5,600 SF 3,000 SY (27,000 SF) new pavements	Addition and alterations to Building 5028. Paved yard area will be expanded to accommodate mission.				
6	VDYD032096	Addition/Alteration to the Communications Facility	1,705 SF	Addition of 1,705 SF to Building 5010 to accommodate mission requirements.				
7	VDYD042033	Construct Tanker Airlift Command Training Center	16,090 SF 450 SY (4,050 SF) new pavements	Construct a 16,090 SF facility to accommodate new squadron mission.				
8		Demolition of Building 5542	2,304 SF	Demolish dilapidated building that is in the footprint of the proposed TACTC.				

Table 2.2-1. List of Projects Included Under the Proposed Action

Note: Building footprints are generally discussed in terms of square feet (SF), whereas pavement footprints are discussed in terms of square yards (SY).



Figure 2.2-1. Projects Under the Proposed Action at the 126 ARW, Scott Air Force Base, Illinois

2.2.3 CONSTRUCT PAVEMENTS AND GROUNDS FACILITY

Under the Proposed Action, a new 8,000 SF pavements and grounds facility would be constructed to support Base Engineering functions. The building would be constructed with a concrete foundation and floor slab; structural steel metal walls with a standing seam metal roof. A 2,000 square yard (SY) (18,000 SF) equipment yard would be included in this project to support outdoor storage of appropriate equipment. All utilities and site improvements would be included. The exterior would be architecturally compatible with the surrounding Base facilities.

2.2.4 CONSTRUCT NEW SECURITY FORCES FACILITY

Under the Proposed Action, a new 13,900 SF Security Forces facility would be constructed. The building would be constructed with a concrete foundation and floor slab, with masonry walls, an exterior brick veneer, and a metal seam roof. The exterior would be architecturally compatible with the surrounding Base facilities. All utilities and site improvements would be included. There would be no new pavements required.

2.2.5 ADDITION/ALTERATION TO THE DEPLOYMENT PROCESSING CENTER

Under this project, interior alterations would occur to Building 5028, which is currently 14,748 SF. There would also be an addition of approximately 5,600 SF to accommodate the undersized Deployment Processing Center. The addition would require a reinforced concrete foundation and floor slab with steel-framed metal siding, and metal standing seam roof. All utilities and site improvements would be included. The exterior would be architecturally compatible with the surrounding Base facilities. A 1,000 SY (9,000 SF) area of pavements would be added to an area of approximately 2,000 SY (18,000 SF) of pavements to accommodate outdoor storage.

2.2.6 ADDITION/ALTERATION TO THE COMMUNICATION FACILITY

Under this project, interior alterations would occur to 10,000 SF of Building 5010 to support the requirements of the computer equipment, and training facilities that would be located in this facility. There would also be an addition of approximately 1,705 SF to accommodate the undersized communication function. The addition would require a reinforced concrete foundation and floor slab with steel-framed masonry walls, and metal standing seam roof. All utilities and site improvements would be included. The exterior would be architecturally compatible with the surrounding Base facilities. No additional pavements would be required.

2.2.7 CONSTRUCT TANKER AIRLIFT COMMAND TRAINING CENTER

There is currently no TACTC at the 126 ARW, and therefore this would be a new 16,090 SF facility to support this function. The building would be designed as a single story masonry facility constructed on a reinforced concrete slab with steel-framed masonry walls and metal

standing seam roof. All utilities and site improvements would be included. The exterior would be architecturally compatible with the surrounding Base facilities. There is an existing parking area across the street from the proposed location that would accommodate the majority of parking requirements at this new facility. An additional 450 SY (4,050 SF) parking area would be added.

2.2.8 FACILITY DEMOLITION

Under the Proposed Action, one building would be demolished. Building 5542 is a multi-vehicle carport that was constructed in 1951 and is 2,304 SF. It used to be a carport within the old family housing area of Scott AFB. The building is in dilapidated condition and is in the footprint of the proposed TACTC.

2.2.9 SUMMARY OF SURFACE DISTURBANCE ASSOCIATED WITH THE PROPOSED ACTION

As a result of the Proposed Action, there would be approximately 45,166 SF of net new building footprint, and 31,050 SF of net new pavements, resulting in a total of approximately 1.75 acres of new impervious surface. Table 2.2-3 summarizes proposed surface disturbance as a result of the Proposed Action.

2.3 ALTERNATIVE ACTION #1: IMPLEMENTATION OF PROPOSED CONSTRUCTION AND DEMOLITION IN ADDITION TO AIRCRAFT INCREASE AND ASSOCIATED ACTIONS

Under this alternative, the eight construction and demolition projects associated with the Proposed Action, as described in Section 2.2 would still occur, in addition to other construction activities (described in more detail below), and non-construction related actions associated with the 126 ARW Master Plan. Under this alternative, the 126 ARW would release their eight PAA KC-135E model aircraft and would acquire 12 PAA KC-135R model aircraft. The number of aircraft operations would increase by approximately 50 percent as a result of the four additional PAA. The authorized flying time and manning would be based on 12 PAA. In general, aircraft maintenance activities at the 126 ARW would remain approximately the same as they currently are because the new model is more efficient and requires less maintenance. There would be an increase of 27 personnel (firefighters) as a result of this alternative. This alternative will be carried forward for further analysis.
Project	(a) New Pavement	(b) Demolished Pavement	(c) Net New Pavement (a – b)	(d) New Building Footprint	(e) Demolished Footprint	(f) Net New Building Footprint (d – e)	Total New Impervious Surface (c + f)
1 – Construct Pumphouse Spill Containment	0	0	0	0	0	0	0
2 – Addition to Mid-Field Fire Station	0	0	0	2,175 SF	0	2,175 SF	2,175 SF
3 – Construct Pavements and Grounds Facility	18,000 SF (2,000 SY)	0	18,000 SF	8,000 SF	0	8,000 SF	26,000 SF
4 – Construct a New Security Forces Facility	0	0	0	13,900 SF	0	13,900 SF	13,900 SF
5 – Addition/Alteration to Deployment Processing Center	9,000 SF (1,000 SY)	0	9,000 SF	5,600 SF	0	5,600 SF	14,600 SF
6 – Addition/Alteration to Communications Facility	0	0	0	1,705 SF	0	1,705 SF	1,705 SF
7 – Construct Tanker Airlift Command Training Center	4,050 SF (450 SY)	0	4,050 SF	16,090 SF	0	16,090 SF	20,140 SF
8 – Demolish Building 5542	0	0	0	0	2,304 SF	-2,304 SF	-2,304 SF
Totals	31,050 SF	0	31,050 SF	47,470 SF	-2,304 SF	45,166 SF	76,216 SF (1.75 acres)

Table 2.2-3. Pavements and Building Footprints – Net New Impervious Surface Under the Proposed Action

Under Alternative Action #1, additional construction activities required are described in the following sections.

2.3.1 CONSTRUCT AIRCRAFT APRON EXPANSION

Under this alternative, the existing aircraft parking apron would be expanded to accommodate the additional four KC-135R aircraft that would be beddown at Scott AFB for the 126 ARW. This expansion would be located as shown in Figure 2.3-1, at the northwest end of the existing apron. The expansion would be approximately 8,000 SY (72,000 SF) of pavements and would be constructed of reinforced concrete that is two to three feet thick.

2.3.2 ADDITION TO MID-FIELD FIRE STATION (FURTHER ADDITION)

Under this alternative, in addition to the expansion of the Mid-Field Fire Station under the Proposed Action, an additional expansion of 1,200 SF would occur to the existing Mid-Field Fire Station to accommodate the additional space and training requirements for an additional 27 fire fighters for the 126 ARW. The structural addition would primarily accommodate classrooms, lockers, and administrative areas for these additional fire fighters. The addition would include a concrete foundation and floor slab; steel-framed masonry walls and floor structure; and all utilities. There would be no additional parking area required. The exterior would be architecturally compatible with the surrounding Base facilities.

2.3.3 ADDITION OF HYDRANT PIT AND BLAST DEFLECTORS ON THE PARKING APRON

Under this alternative, a new fuel hydrant pit and blast deflector would be located on the aircraft parking apron, as shown in Figure 2.3-1. The hydrant pit would allow for more efficient aircraft fueling on the apron, and would comply with the military Type III hydrant fueling system requirements. The proposed hydrant pit would require approximately 400 linear feet of trenching, and the stainless steel pipe connecting the proposed hydrant pit to the existing hydrant pits would be located approximately 5 feet below ground surface (bgs). The pipe would be connected to the existing eight hydrant pumps on the apron that are directly connected to the Petroleum, Oil, and Lubricants (POL) facility. The hydrant pit itself would be located within a subterranean concrete vault that would be approximately 320 cubic feet (8 feet long, 5 feet wide, and 8 feet deep). The blast deflector would deflect heat and noise from engine run-ups on the apron. The deflector would be located as shown in Figure 2.3-1. It would be constructed of concrete.



Figure 2.3-1. Projects Under Alternative Action #1 at the 126 ARW, Scott Air Force Base, Illinois

2.3.4 SUMMARY OF SURFACE DISTURBANCE ASSOCIATED WITH THE ALTERNATIVE ACTION

As a result of the Alternative Action, there would be approximately 46,366 SF of net new building footprint, 103,050 SF of net new pavements, resulting in a total of approximately 3.43 acres of new impervious surface. Table 2.3-1 summarizes proposed surface disturbance as a result of the Alternative Action.

Project	(a) New Pavement	(b) Demolished Pavement	(c) Net New Pavement (a – b)	(d) New Building Footprint	(e) Demolished Footprint	(f) Net New Building Footprint (d – e)	Total New Impervious Surface (c + f)
1 – Construct Pumphouse Spill Containment	0	0	0	0	0	0	0
2 – Addition to Mid-Field Fire Station	0	0	0	2,175 SF	0	2,175 SF	2,175 SF
3 – Construct Pavements and Grounds Facility	18,000 SF (2,000 SY)	0	18,000 SF	8,000 SF	0	8,000 SF	26,000 SF
4 – Construct a New Security Forces Facility	0	0	0	13,900 SF	0	13,900 SF	13,900 SF
5 – Addition/Alteration to Deployment Processing Center	9,000SF (1,000 SY)	0	9,000 SF	5,600 SF	0	5,600 SF	14,600 SF
6 – Addition/Alteration to Communications Facility	0	0	0	1,705 SF	0	1,705 SF	1,705 SF
7 – Construct Tanker Airlift Command Training Center	4,050 SF (450 SY)	0	4,050 SF	16,090 SF	0	16,090 SF	20,140 SF
8 – Demolish Building 5542	0	0	0	0	2,304 SF	-2,304 SF	-2.304 SF
9 – Construct Aircraft Apron Expansion	72,000 SF (8,000 SY)	0	72,000 SF	0	0	0	72,000 SF
10 – Addition to Mid-Field Fire Station	0	0	0	1,200 SF	0	1,200 SF	1,200 SF
11 – Construct Hydrant Pit and Blast Deflector	0	0	0	0	0	0	0
Totals	103,050 SF	0	103,050 SF	48,670 SF	-2,304 SF	46,366 SF	149,416 SF (3.43 acres)

Table 2.3-1. Pavements and Building Footprints – Net New Impervious Surface Under the
Alternative Action

2.4 NO ACTION ALTERNATIVE

Under the No Action alternative, the 126 ARW would maintain their existing facilities and would not construct, renovate, or demolish facilities, as proposed. The 126 ARW would continue to have insufficient space and inefficient configuration to meet mission requirements.

- The Pumphouse would continue to be noncompliant with the NEC's Type III standard for hydrant fueling systems, and fuel accidentally released would eventually flow into the storm drainage system. Additionally, fuel could drain into the existing electrical manhole leading to a potentially dangerous fire situation.
- The Mid-field Fire Station would continue to be undersized, which would continue to impact proper training.
- The Pavements and Grounds facility would not be constructed and the 126 ARW would continue to use the dilapidated, insufficiently equipped 1950's carport for this function. Insufficient storage space would continue to require additional maintenance on the equipment that is forced to be stored in the elements.
- The Security Forces facility would continue to be undersized by approximately 50 percent and training and storage would continue at dispersed locations throughout the installation, resulting in inefficiencies.
- The Deployment Processing Center would also continue to be undersized and operated out of several separate facilities, resulting in inefficient processing of personnel through this time-sensitive operation.
- The Communications facility would continue to operate in a facility that was never designed for state-of-the-art computer equipment, which would ultimately require unnecessary maintenance on this equipment.
- The TACTC would not be constructed and this training capacity would remain unfulfilled.

Under the No Action alternative, these deficiencies would continue to impair the 126 ARW's ability to successfully conduct their mission and to maintain wartime readiness and training.

THIS PAGE INTENTIONALLY LEFT BLANK.

3.0 EXISTING CONDITIONS

Chapter 3.0 describes the existing environmental and socioeconomic conditions potentially affected by the Proposed Action. This section provides information to serve as a baseline from which to identify and evaluate environmental and socioeconomic changes likely to result from implementation of the Proposed Action. Baseline conditions represent current conditions. The potential environmental and socioeconomic impacts of implementing the Proposed Action or its alternatives are described in Chapter 4.0.

In compliance with NEPA, CEQ guidelines, and 32 CFR Part 989, et seq., the description of the affected environment focuses on those resources and conditions potentially subject to impacts. These resources and conditions include: earth resources, water resources, biological resources, air quality, noise, land use and visual resources, socioeconomics and environmental justice, cultural resources, safety, hazardous materials and wastes, and infrastructure.

3.1 EARTH RESOURCES

3.1.1 DEFINITION OF THE RESOURCE

Earth resources include geology, soils, and topography. Geologic resources of an area typically consist of surface and subsurface materials and their inherent properties. The term "soils" refers to unconsolidated materials formed from the underlying bedrock or other parent material. Soils play a critical role in both the natural and human environment. Soil drainage, texture, strength, shrink/swell potential, and erodibility all determine the suitability of the ground to support manmade structures and facilities. Topography refers to an area's surface features including its vertical relief. These resources may have scientific, historical, economic, and recreational value. The region of influence (ROI) for earth resources in this EA includes the 126 ARW installation at Scott AFB, Illinois.

3.1.2 EXISTING CONDITIONS

3.1.2.1 Geology

The geologic units of St. Clair County include Paleozoic sedimentary rocks and Cenozoic unconsolidated materials. Pennsylvanian Age bedrock lies approximately 85 feet below the surface and includes layers of shale, siltstone, sandstone, limestone, claystone, and coal. The Pennsylvanian strata are approximately 265 feet thick. Water-yielding Chesterian Series sandstones lie beneath the Pennsylvanian strata. Wells in these sandstones yield 20 to 50 gallons per minute (Scott AFB 2004a; 375th Civil Engineering Squadron, Environmental Flight [375 CES/CEV] 2003).

The Herron No. 6 coal bed, with an average thickness of 6 to 7 feet, lies 90 to 200 feet below the surface of Scott AFB and extends out several miles to the west and south. Abandoned subsurface mines are located about 1 mile southwest of the Scott AFB runway and about 2 miles northwest of the Base. There has been no mining under the Base itself, and therefore there is no subsidence risk from such activity (Scott AFB 2004a; 375 CES/CEV 2003).

Scott AFB lies within Seismic Zone IX, which contains the New Madrid Fault Zone, that extends from Cairo, Illinois, on the Ohio River southward through New Madrid, Missouri. The New Madrid Fault Zone is the most active seismic area east of the Rocky Mountains, with almost weekly tremors and on rare occasions, small earthquakes measuring 3.0 to 4.0 or more on the Richter scale. The last major earthquake along this fault was in 1812 and measured more the 8.0 on the Richter scale (United States Geological Survey 2005).

Glacial and alluvial deposits ranging in thickness from 50 feet to 125 feet dominate surficial geology at the Base. The Base lies on the Springfield Plain subdivision of the Till Plains section of the Central Lowlands Physiographic Province and is located on the west end of the Silver Creek Valley Basin (Scott AFB 2004a; 375 CES/CEV 2003).

3.1.2.2 Soils

Upland soils in St. Clair County consist primarily of a glacial till plain or glacial outwash plain covered by wind-deposited soils, called loess. The thickness of the loess in the County ranges from 10 feet in the eastern portions to 100 feet in the western portions. The predominant soils at Scott AFB are silt loams and silty clay loams that occur to a depth of 16 inches. These soils typically have a moderately high water holding capacity, moderate to high shrink-swell ratios, and a moderate to high corrosive potential (Scott AFB 2004a; 375 CES/CEV 2003). According to the Natural Resources Conservation Service (NRCS) 2002 *Soil Survey for St. Clair County*, there are six soil mapping units that occur on the 126 ARW installation at Scott AFB. These mapping units are described below.

Bethalto Silt Loam, 0-2 percent slopes. The soil is derived from alluvial parent material, such as sand, silt, or clay that was deposited by a riverine system. This soil is typically found in alluvial fans. These are somewhat poorly drained soils (NRCS 2002). The Bethalto Silt Loam is the predominant soil type on the 126 ARW installation. As shown in Figure 3.1-1, this soil type comprises approximately 85 percent of the installation.

Caseyville Silt Loam, 0-2 percent slopes. This soil type is found on interfluvial uplands on till plains that are nearly flat to gently rolling. The parent material of this soil is loess, which is a fine-grained, silty material that was wind deposited. This is a somewhat poorly drained soil (NRCS 2002). This soil type is found on the most eastern portion of the 126 ARW installation.



Figure 3.1-1. Soil Mapping Units on the 126 ARW Installation at Scott AFB, Illinois

Wakeland Silt Loam, 0-2 percent slopes, frequently flooded. This soil type is found in floodplains, and is comprised of alluvial material, such as sand, silt, or clay, that was deposited by a riverine system. This is a somewhat poorly drained soil that is frequently flooded (NRCS 2002). This soil type is found along the drainage system on the northern portion of the 126 ARW installation.

Mascoutah Silty Clay Loam, 0-2 percent slopes. This soil type is found on interfluvial uplands on till plains that are nearly flat to gently rolling. The parent material of this soil is loess, which is a fine-grained, silty material that was wind deposited. This is a poorly drained soil (NRCS 2002). This soil is found on the western portion of the 126 ARW parking apron.

Edwardsville Silt Loam, 0-2 percent slopes. This soil type is similar to the Mascoutah Silty Clay Loam in that it is also found on interfluvial uplands on till plains that are nearly flat to gently rolling. The parent material of this soil is also loess. This is a somewhat poorly drained soil (NRCS 2002). This soil is found in a very small pocket on the western portion of the 126 ARW installation where East Drive bends due north.

Menfro Silt Loam, 10-18 percent slopes, eroded. This soil type is typically found on side slopes of till plains. The parent material of this soil is also loess. This is a well-drained soil that is prone to erosion due to the slope (NRCS 2002). There is just one small pocket of this soil type on the eastern portion of the 126 ARW installation along East Drive.

Because of the relatively flat topography on Scott AFB, soil erosion is not a widespread problem. The primary development constraints associated with the soils at the 126 ARW are poor drainage and moderate to severe shrink-swell potential for the upland soils. Subsurface construction, such as basements, can be problematic due to the poor drainage characteristics and from periodic flooding (Scott AFB 2004a; 375 CES/CEV 2003).

3.1.2.3 Topography

Scott AFB is located on the west end of the Silver Creek Valley Basin, which is generally characterized by flat to gently rolling hills. The Base land surface is generally flat. The maximum surface elevation at the Base is 510 feet above mean sea level (MSL) at a till ridge north of the Base golf course. The lowest surface elevation is approximately 420 feet MSL along the eastern boundary of the Base within the Silver Creek floodplain. The elevation of Silver Creek east of the Base is about 405 feet MSL (Scott AFB 2004a; 375 CES/CEV 2003).

3.2 WATER RESOURCES

3.2.1 DEFINITION OF THE RESOURCE

Water resources analyzed in this EA include surface water and groundwater quantity and quality. Surface water resources include lakes, rivers, and streams and are important for a variety of reasons, including economic, ecological, recreational, and human health. Groundwater includes the subsurface hydrologic resources of the physical environment and is an essential resource. Groundwater properties are often described in terms of depth to aquifer or water table, water quality, and surrounding geologic composition.

Other issues relevant to water resources include the downstream water and watershed areas affected by existing and potential runoff, and hazards associated with 100-year floodplains. Floodplains are defined by EO 11988, *Floodplain Management*, as "the lowland and relatively flat areas adjoining inland and coastal waters including flood-prone areas of offshore islands, including at a minimum, the area subject to a one percent or greater chance of flooding in any given year" (that area inundated by a 100-year flood). Floodplain values include natural moderation of floods, water quality maintenance, groundwater recharge, as well as habitat for many plant and animal species.

The ROI for water resources in this EA comprises the area of the 126 ARW installation and airfield at Scott AFB, underlying aquifers, and their downstream drainages.

3.2.2 EXISTING CONDITIONS

3.2.2.1 Surface Water

The ROI is located in the Lower Kaskaskia Watershed, which drains an area of 1,060,900 acres (NRCS 2004). Silver Creek, a tributary of the Kaskaskia River (which is a tributary of the Mississippi River), is located on the east side of Scott AFB. It drains approximately 60 percent of surface runoff from the Base including surface runoff from the 126 ARW installation (375 CES/CEV 2003; Scott AFB 2004a). The Illinois Environmental Protection Agency (Illinois EPA) rates water quality in Silver Creek as "fair" in the vicinity of Scott AFB. Nutrients and siltation from agricultural operations are the primary non-point sources of water pollution (375 CES/CEV 2003). Ash Creek drains the remainder of Scott AFB, but does not receive surface runoff from the 126 ARW installation.

Surface water features on Scott AFB include North Ditch, South Ditch, and Mosquito Creek, all of which are tributaries of Silver Creek, as well as Scott Lake, Cardinal Lake, and the Base golf course ponds (Scott AFB 2004a). The only surface water feature on the 126 ARW installation is

Cardinal Creek and an open drainage ditch that flows east to west through the northern portion of the installation eventually discharging into Silver Creek (375 AW 2005).

A substantial percentage of the 126 ARW installation consists of surfaces that are impervious to water infiltration, such as asphalt, concrete, or buildings/facilities. Drainage from these areas is directed by surface topography and perimeter curbing to enclosed storm sewers and open channels (Scott AFB 2004b). Scott AFB is divided into 12 defined drainage basins. Surface runoff generated within the 126 ARW installation is associated with portions of Basins N1 and T1. Basin N1 includes the aircraft ramp area of the 126 ARW, and Basin T1 includes the 126 ARW aircraft hangars, flightline, and bulk fuel storage. Surface water runoff from Basin N1 flows east to Silver Creek via the North Ditch. Surface water runoff from Basin T1 is conveyed via underground piping to an open channel, which is Cardinal Creek. This surface water then drains into the wetland area located in the eastern portion of Scott AFB before reaching Silver Creek (Scott AFB 2004b).

Storm water runoff generated by the 126 ARW is permitted under an industrial storm water permit issued to Scott AFB by the Illinois EPA. Runoff is managed in accordance with the Scott AFB *Final Storm Water Pollution Prevention Plan* (SWPPP), which is a requirement of the permit (Scott AFB 2004b). The 375 AW SWPPP is an engineering and management strategy prepared specifically for the 375 AW to improve the quality of the storm water runoff and thereby improve the quality of the receiving waters. The SWPPP also works to minimize storm water runoff thereby enhancing infiltration and subsequent ground water recharge. This plan ensures implementation of best management practices (BMPs) and delineates monitoring, training, and documentation requirements of the 375 AW's National Pollutant Discharge Elimination System (NPDES) storm water permit. The plan includes notification, permit application, and erosion control requirements for any construction activity that will disturb through clearing, grading, or excavating greater than one acre at the installation.

3.2.2.2 Floodplains

According to the Scott AFB Final Floodplain Survey (2005) and the Federal Emergency Management Agency (FEMA) Flood Insurance Rate Map (2003) associated with the ROI, portions of the 100-year and 500-year floodplain associated with Silver Creek and Cardinal Creek are located within the 126 ARW installation (375 AW 2005; FEMA 2003). The 100-year and 500-year floodplains associated with Cardinal Creek are located in the northern portion of the installation, while a small portion of the 100-year and 500-year floodplains associated with Silver Creek are located in the eastern portion of the installation. The remainder of the 126 ARW installation is located within an area defined by FEMA as Zone X, which are "areas determined to be outside 500-year floodplain." The area immediately east of the 126 ARW installation boundary is defined as "special flood hazard areas inundated by 100-year flood," Zone AE on the FEMA map, where base flood elevations are determined. This area includes the 100-year floodplain associated with Silver Creek, as well as the Silver Creek floodway, which is defined by FEMA as "... the channel and adjacent overbank areas necessary to effectively convey floodwaters" (FEMA 2003; FEMA 2004).

3.2.2.3 Groundwater

Groundwater underlying Scott AFB ranges from 20 feet bgs on the western side of the Base to less than 1 foot bgs on the eastern side of the Base. Ground water flow is generally west to east towards Silver Creek (375 CES/CEV 2003). Because the 126 ARW installation is located in the eastern portion of the Base, it is assumed that groundwater underlying the 126 ARW installation is closer to 1 foot bgs.

3.3 BIOLOGICAL RESOURCES

3.3.1 DEFINITION OF THE RESOURCE

Biological resources consist of native or naturalized plants and animals, and their habitats, including wetlands. Although the existence and preservation of biological resources are both intrinsically valuable, these resources also provide essential aesthetic, recreational, and socioeconomic benefits to society. This section focuses on plant and animal species and vegetation types that typify or are important to the function of the ecosystem, are of special societal importance, or are protected under federal or state law or statute. For purposes of this assessment, sensitive biological resources are defined as those plant and animal species listed by the USFWS or the Illinois Department of Natural Resources (IDNR) as species of concern. Three categories of protection status are included in this section including, 1) federally listed threatened and endangered species, 2) state listed threatened and endangered species, and 3) other sensitive species (i.e., federal candidate, federal species of concern, proposed threatened, proposed endangered species, and state species in greatest need of conservation).

Federally Listed Threatened and Endangered Species. The ESA of 1973 provides protection to species federally listed as endangered or threatened. Endangered species are those species that are at risk of extinction in all or a significant portion of their range. Threatened species are those that could be listed as endangered in the near future.

State Listed Threatened and Endangered Species. The Illinois Endangered Species Protection Act (IDNR 2004) protects threatened and endangered species in the State of Illinois. Endangered species are those species in danger of extinction as a breeding species within the state. Illinois-threatened species are breeding species likely to become a state-listed endangered species within the foreseeable future.

Other Sensitive Species. Taxa under this heading are those federally listed as candidate, proposed endangered, proposed threatened, and state species in greatest need of conservation. Candidate species are those for which the USFWS has sufficient information on biological vulnerability and threats to support proposals to list them as endangered or threatened, but issuance of proposed rules for these species is precluded by higher priority listing actions. Proposed endangered and threatened species are those proposed for listing as endangered and threatened, respectively, and for which formal ruling is in progress. At present, candidate and proposed species do not receive legal protection under the ESA.

In addition, EO 13186, *Responsibilities of Federal Agencies to Protect Migratory Birds* (2001), recognizes the ecological and economic importance of migratory birds to this and other countries. It requires federal agencies to evaluate the effects of their actions and plans on migratory birds (with an emphasis on species of concern) in their NEPA documents. Species of concern are those identified in, 1) the report "Migratory Nongame Birds of Management Concern in the United States" (USFWS 1995), 2) priority species identified by established plans such as those prepared by Partners in Flight, or 3) listed species in 50 CFR 17.11 *Endangered and Threatened Wildlife*.

Wetlands. Wetlands are defined by the USACE and USEPA as "those areas that are inundated or saturated by surface or groundwater at a frequency and duration sufficient to support, and that under normal circumstances do support, a prevalence of vegetation typically adapted for life in saturated soil conditions. Wetlands generally include marshes, bogs, and similar areas" (33 CFR 328.3[b] 1984). Wetlands provide a variety of functions including groundwater recharge and discharge; flood attenuation; sediment stabilization; sediment and toxicant retention; nutrient removal and transformation; aquatic and terrestrial diversity and abundance; and aesthetic values. Three criteria are necessary to define wetlands: vegetation (hydrophytes), soils (hydric), and hydrology (frequency of flooding or soil saturation). Jurisdictional wetlands are those subject to regulatory authority under Section 404 of the CWA and EO 11990, *Protection of Wetlands*.

The ROI for biological resources includes Scott AFB and adjacent properties that exhibit contiguous tracts of land for migration of species.

- 3.3.2 EXISTING CONDITIONS
- 3.3.2.1 Vegetation

Scott AFB lies on the Springfield Plain subdivision of the Till Plains section of the Central Lowlands Physiographic Province and is located on the west end of the Silver Creek Valley Basin, an area characterized by generally flat to gently rolling hills (Scott AFB 2004a). The

Base land surface is generally level. The Central Lowlands Physiographic Province is the result of several glaciations. Glaciers covered most of the area in recent geologic history and left deep soil deposits which are now some of the richest agricultural lands in the province. The topography is flat to slightly rolling and the drainage pattern has been significantly altered from its original, prior-to-glaciation condition. Natural vegetative communities within the Base have been largely modified by past Base operations (USAF 1991). Scott AFB is located on relatively flat topography with low hills to the west and north. Land use around the Base is mainly agricultural with natural vegetation between fields, along roads and streams and near residences.

Upland vegetation on Scott AFB has been characterized into four different community types: Urban Upland, Upland Forest, Nonforested Upland, and Riparian Forest (Scott AFB 2005a). The Urban Upland community type covers approximately 80 percent of Scott AFB and typically consists of manicured lawns and associated landscaping and trees along streets. Other areas included in this community type are the ILANG portion of the Base, the golf course, the driving range, the unpaved areas of the airfield, and the former housing area located east of the golf course.

The Upland Forest community areas are dominated by upland trees such as white oak, northern red oak, sassafras, black walnut, hickories, black cherry, and hackberry. Various species of pine trees have also been planted in these areas. Amur honeysuckle is established in the upland forest and is a prominent component of the understory. Nearly all of the upland forested areas occur between the Silver Creek floodplain and the Family Camp area which is located in the northeast portion of Scott AFB. Other fragments of this community occur as narrow strips along steep fill slopes adjacent to the floodplain (Scott AFB 2005a).

The Nonforested Upland areas are dominated by grass species such as fescue, bluegrass and lovegrass, and typical open-field vegetation such as goldenrod with some invasion of smaller trees and shrubs in areas that are not maintained by mowing. The Nonforested Upland areas are found around the Family Camp area, various locations around Scott Lake, an area at the southern end of the airfield and one other area in the southern portion of the Base around the former landfill (Scott AFB 2005a).

The Riparian Forest areas contain many vegetative species common to wetlands, including ash, elm, cottonwood, pin oak, and silver maple. Two species, the hackberry and the shagbark hickory tend to differentiate the poorly drained riparian forest areas from the poorly drained wetlands. The Riparian Forest has a canopy that is approximately 30 to 40 percent open (Scott AFB 2005b). The understory of the Riparian Forest is relatively sparse; however, stinging nettle and white heath aster dominate a dense herbaceous layer in this community. The Riparian Forest have been classified as jurisdictional wetlands. These areas were intermixed with jurisdictional and

non-jurisdictional areas and were therefore mapped as a complex of wetland and non-wetland areas.

3.3.2.2 Wildlife

This section focuses on terrestrial and aquatic wildlife living in a natural, undomesticated setting. Numerous wildlife species occur or have the potential to occur at Scott AFB including more than 40 species of herpetofauna (USAF 1991) and over 230 species of birds (USACE 2002). Mammal species that may occur in the area include Eastern cottontail, woodchuck, gray and fox squirrels, white tailed deer, Virginia opossum, beaver, muskrat, raccoon, and coyote (Hoffmiester 2002).

Wildlife species found at Scott AFB are generally limited to species that have adapted to existence in a developed, semi-urban, and industrial setting. Of the 83 bird species detected during the 2001 study, the most common species observed were the Common Grackle, Downy Woodpecker, Wood Duck, Red-bellied Woodpecker and White Breasted Nuthatch. During migratory periods, the Indigo Bunting, Blue-gray Gnatcatcher and the Prothonotary Warbler were the most common species. Of special note was the presence of a Little Blue Heron and a Snowy Egret, both of which are listed as state endangered species (Martin et al. 2002).

3.3.2.3 Rare, Threatened, and Endangered Species

Table 3.3-1 lists special status species that could potentially occur in the project vicinity. It is the policy of the USAF to treat any state-listed species with the same protection afforded to the federal-listed species whenever practicable. Although not required by the federal ESA, the USAF will provide similar conservation measures for species protected by Illinois state law, when such protection is not in direct conflict with the military mission.

Recent correspondence with the IDNR revealed that no records exist of any state-listed threatened or endangered species on or adjacent to the Base. For birds, a population record typically indicates a site of known breeding or nesting because birds can occur incidentally at many sites during migration (Scott AFB 2005a). The Illinois Natural Heritage Society database does not provide a conclusive statement on the presence, absence, or condition of a listed species and does not preclude the need for field surveys. The information contained in the database is based only on the best available information at the time the database was provided.

Common Name	Scientific Name	Status	Potential for Occurrence
PLANTS			
Decurrent false aster	Boltonia decurrens	FT	Low. Not known from project area. Occurs on sunlit floodplains and open wetlands.
Buffalo clover	Trifolium reflexum	ST	Low. Not known from project area. Occurs on dry mesic savannas, flatwoods, and prairies.
Green trillium	Trillium viride	SE	Low. Not known from project area. Occurs in bottomland forests.
BIRDS		_	
Bald Eagle	Haliaeetus leucocephalus	FT	Low. No known nests or roost sites within 15 miles and no open water in the project vicinity. Potential for flyovers and perching.
Short-eared Owl	Asio flammeus	SE	Low. No known nests or sightings in project area. Ground nesting prefers prairies.
Little Blue Heron	Egretta caerulea	SE	Possible. Sighted in the 2001 bird survey as migrant through the area.
Snowy Egret	Egretta thula	SE	Possible. Sighted in the 2001 bird survey as migrant through the area.
Common Moorhen	Gallinula chloropus	ST	Low. Not known from project area. Prefer open wetlands.
Loggerhead Shrike	Lanius ludovicianus	ST	Low. Not known from project area. Prefer open areas with windrows of trees and brush.
Yellow-crowned Night Heron	Nyctanassa violacea	SE	Low. Not known from project area. Prefer open water wetlands.
Black-crowned Night Heron	Nycticorax nycticorax	SE	Low. Not known from project area. Prefer open water wetlands.
MAMMALS	-	•	
Indiana bat	Myotis sodalis	FE	Possible. One individual was captured in a 2001 survey; however, no maternity colonies are known from the project area. Presence or absence of this species will be determined by future studies.
REPTILES			
Eastern massasauga FE = Federal Endangered, FT =	Sistrurus catenatus catenatus	FC	Low. Possible fragmented habitat has been documented on Scott AFB but it is unlikely that this species would occur in this fragmented habitat.

Table 3.3-1. Threatened and Endangered Species Documented or Likely to Occur in St.Clair County, with Assessment of Potential for Occurrence on the Installation.

FE = Federal Endangered, FT = Federal Threatened, FC= Federal Candidate, ST = State Threatened Sources: IDNR 2004; INHS Database, Correspondence November 4, 2004.

3.3.2.4 Wetlands and Other Aquatic Habitat

Section 404 of the CWA established a program to regulate the discharge of dredge and fill material into waters of the U.S., including wetlands. Activities in waters of the U.S. that are regulated under this program include fills for development, water resource projects (such as dams and levees), infrastructure development (such as highways and airports), and conversion of wetlands to uplands for farming and forestry. EO 11990, *Protection of Wetlands*, requires federal agencies, including the USAF, to minimize the destruction, loss, or degradation of wetlands, and to preserve and enhance the natural and beneficial values of wetlands.

In July 2004, in coordination with the St. Louis District USACE, the 375 CES/CEV conducted a formal wetland delineation of all areas on Scott AFB (Scott AFB 2005b). Twenty-one wetlands were observed and delineated during the 2004 field activities. These wetlands were generally characterized as palustrine forested wetland communities, located within the forested floodplain of Silver Creek. Additional wetlands observed included isolated palustrine emergent (i.e., marshes) areas in various developed and undeveloped areas of the Base. The USACE determined that man-made surface waters such as Scott Lake, Cardinal Lake, irrigation ponds, and golf course ponds that were constructed in mapped upland soils should be considered nonjurisdictional waters. Additionally, isolated wetlands would also be considered nonjurisdictional. This designation extends to nearly all of these waters and wetlands not located within the Silver Creek floodplain, except the pond and wetlands along both sides of Golf Course Road in front of the clubhouse.

With regard to streams, it was determined that essentially every stream on Scott AFB has been channelized. The USACE indicated that these channels would be considered "waters of the U.S." if historical stream flow could be identified on historical aerial photography. As part of the delineation, historical aerial photographs were evaluated and it was determined that all swales and streams would meet the jurisdictional criteria with the exception of one man-made swale located south of MidAmerica Airport Taxiway within the flight line (Scott AFB 2005b).

3.4 AIR QUALITY

3.4.1 DEFINITION OF THE RESOURCE

This section discusses air quality considerations and conditions in the area surrounding Scott AFB in St. Clair County, Illinois. It addresses air quality standards and describes current air quality conditions in the region.

Federal Air Quality Standards. Air quality is determined by the type and concentration of pollutants in the atmosphere, the size and topography of the air basin, and local and regional meteorological influences. The significance of a pollutant concentration in a region or

geographical area is determined by comparing it to federal and/or state ambient air quality standards. Under the authority of the CAA, the USEPA has established nationwide air quality standards to protect public health and welfare, with an adequate margin of safety.

These federal standards, known as the NAAQS, represent the maximum allowable atmospheric concentrations and were developed for six "criteria" pollutants: O_3 , NO_2 , CO, respirable particulate matter less than or equal to 10 micrometers in diameter (PM_{10}), SO_2 , and Pb. The NAAQS are defined in terms of concentration (e.g., parts per million [ppm] or micrograms per cubic meter [μ g/m³]) determined over various periods of time (averaging periods). Short-term standards (1-hour, 8-hour, or 24-hour periods) were established for pollutants with acute health effects and may not be exceeded more than once a year. Long-term standards (annual periods) were established for pollutants with chronic health effects and may never be exceeded. Primary standards, as shown in Table 3.4-1, set limits to protect public health, including the health of sensitive populations, such as asthmatics, children, and the elderly. Secondary standards set limits to protect public welfare, including protection against decreased visibility, damage to animals, vegetation, and buildings.

Based on measured ambient criteria pollutant data, the USEPA designates areas of the U.S. as having air quality equal to or better than the NAAQS (attainment) or worse than the NAAQS (nonattainment). Upon achieving attainment, areas are considered to be in maintenance status for a period of 10 or more years. Areas are designated as unclassifiable for a pollutant when there is insufficient ambient air quality data for the USEPA to form a basis of attainment status. For the purpose of applying air quality regulations, unclassifiable areas are treated similar to areas that are in attainment of the NAAQS.

On April 15, 2004, the USEPA promulgated attainment designations for the newly established 8hour O₃ standard effective as of June 15, 2004. Meanwhile, states must continue to implement existing plans developed under the 1-hour standard during the transition to the new 8-hour standard. On December 17, 2004, the USEPA designated areas as attainment or nonattainment for the newly developed standard for respirable particulate matter less than or equal to 2.5 micrometers in diameter ($PM_{2.5}$), which are fine particulates that have not been previously regulated (USEPA 2005).

State Air Quality Standards. Under the CAA, state and local agencies may establish ambient air quality standards (AAQS) and regulations of their own, provided that these are at least as stringent as the federal requirements. The State of Illinois has air quality standards that are virtually identical to the federal standards, except that the new federal $PM_{2.5}$ and 8-hour O_3 standards do not have an Illinois equivalent. A summary of the NAAQS that apply to the proposed project area is presented in Table 3.4-1.

	NAAOS						
	Averaging	NAA					
Air Pollutant	Time	Primary	Secondary				
Carbon Monoxide	8-hour	9 ppm (10 μ g/m ³)					
(CO)	1-hour	35 ppm (40 μ g/m ³)					
Nitrogen Dioxide (NO ₂)	AAM	$0.053 \text{ ppm} (100 \ \mu\text{g/m}^3)$	0.053 ppm (100 μg/m ³)				
Sulfur Dioxide (SO ₂)	AAM	$0.03 \text{ ppm} (80 \ \mu\text{g/m}^3)$					
	24-hour	0.14 ppm (365 μg/m ³)					
	3-hour		$0.5 \text{ ppm} (1,300 \mu\text{g/m}^3)$				
Particulate Matter	AAM	$50 \ \mu g/m^3$	$50 \ \mu g/m^3$				
(PM_{10})	24-hr	$150 \ \mu g/m^3$	$150 \ \mu g/m^3$				
Particulate Matter	AAM	15 μg/m ³	$15 \mu\text{g/m}^3$				
$(PM_{2.5})^{(a)}$	24-hour	$65 \mu\text{g/m}^3$	$65 \mu g/m^3$				
Ozone $(O_3)^{(b)}$	1-hour	0.12 ppm	0.12 ppm				
	8-hour	0.08 ppm	0.08 ppm				
Lead (Pb) & Lead Compounds	3-month	$1.5 \ \mu g/m^3$	1.5 µg/m ³				

Table 3.4-1. National and Illinois Ambient Air Quality Standards

Notes: AAM = Annual Arithmetic Mean; ppm = parts per million; $\mu g/m3$ = micrograms per cubic meter.

(a) The PM2.5 standard (particulate matter with a 2.5 μm diameter or smaller) were promulgated in December 2004 and are effective as of 5 April 2005. The standard will be implemented over the next years. The State of Illinois has not yet adopted this standard.

(b) The 8-hour O3 standard will replace the 1-hour standard in June 2005, one year after the effective date of EPA's recent nonattainment designations. Meanwhile, states must continue to implement existing plans developed under the 1-hour standard during the transition to the new 8-hour standard. The State of Illinois has not yet adopted this standard.

Sources: 40 Code of Federal Regulations 50; Illinois Administrative Code (IAC) 1992.

State Implementation Plan. For non-attainment regions, the states are required to develop an SIP designed to eliminate or reduce the severity and number of NAAQS violations, with an underlying goal to bring state air quality conditions into (and maintain) compliance with the NAAQS by specific deadlines. The SIP is the primary means for the implementation, maintenance, and enforcement of the measures needed to attain and maintain the NAAQS in each state.

Prevention of Significant Deterioration (PSD). Section 162 of the CAA further established the goal of prevention of significant deterioration (PSD) of air quality in all international parks; national parks which exceeded 6,000 acres; and national wilderness areas and memorial parks which exceeded 5,000 acres if these areas were in existence on August 7, 1977. These areas were defined as mandatory Class I areas, while all other attainment or unclassifiable areas were defined as Class II areas. Under CAA Section 164, states or tribal nations, in addition to the

few

federal government, have the authority to redesignate certain areas as (non-mandatory) PSD Class I areas, e.g., a national park or national wilderness area established after August 7, 1977, which exceeds 10,000 acres. PSD Class I areas are areas where any appreciable deterioration of air quality is considered significant. Class II areas are those where moderate, well-controlled growth could be permitted. Class III areas are those designated by the governor of a state as requiring less protection than Class II areas. No Class III areas have yet been so designated. The PSD requirements affect construction of new major stationary sources in the PSD Class I, II, and III areas and are a pre-construction permitting system.

Visibility. CAA Section 169A established the additional goal of prevention of further visibility impairment in PSD Class I areas. Visibility impairment is defined as a reduction in the visual range and atmospheric discoloration. Determination of the significance of an activity on visibility in a PSD Class I area is typically associated with evaluation of stationary source contributions. The USEPA is implementing a Regional Haze rule for PSD Class I areas that will address contributions from mobile sources and pollution transported from other states or regions.

Emission levels are used to qualitatively assess potential impairment to visibility in PSD Class I areas. Decreased visibility may potentially result from elevated concentrations of PM_{10} and SO_2 in the lower atmosphere.

General Conformity. CAA Section 176(c), General Conformity, established certain statutory requirements for federal agencies with proposed federal activities to demonstrate conformity of the proposed activities with each state's SIP for attainment of the NAAQS. Federal activities must not:

- (a) cause or contribute to any new violation;
- (b) increase the frequency or severity of any existing violation; or
- (c) delay timely attainment of any standard, interim emission reductions, or milestones in conformity to a SIP's purpose of eliminating or reducing the severity and number of NAAQS violations or achieving attainment of NAAQS.

General conformity applies only to nonattainment and maintenance areas. If the emissions from a federal action proposed in a nonattainment area exceed annual thresholds identified in the rule, a conformity determination is required of that action. The thresholds become more restrictive as the severity of the nonattainment status of the region increases.

Stationary Source Operating Permits. In Illinois, the Illinois EPA, Bureau of Air, is responsible for identifying air pollution problems, proposing appropriate regulations, conducting inspections, and reviewing permit applications. Title V of the CAA Amendments of 1990

requires states to issue Federal Operating Permits for major stationary sources. A major stationary source in an attainment or maintenance area is a facility (i.e., plant, base, or activity) that emits more than 100 tons per year (TPY) of any one criteria air pollutant, 10 TPY of a hazardous air pollutant, or 25 TPY of any combination of hazardous air pollutants. Thresholds are lower for pollutants for which a region is in nonattainment status. The purpose of the permitting rule is to establish regulatory control over large, industrial activities and to monitor their impact upon air quality. Illinois's Title V program and other air program laws, including licensing (i.e., permitting) are found in IAC Title 35, Subtitle B.

3.4.2 EXISTING CONDITIONS

Regional Air Quality. Federal regulations at 40 CFR 81 delineate certain air quality control regions (AQCRs), which were originally designated based on population and topographic criteria closely approximating each air basin. The potential influence of emissions on regional air quality would typically be confined to the air basin in which the emissions occur. Therefore, the ROI for the Proposed Action is the Metropolitan St. Louis Interstate AQCR (AQCR 70), which includes Bond, Clinton, Madison, Monroe, Randolph, St. Clair, and Washington Counties in Illinois; and Franklin, Jefferson, St. Charles, and St. Louis Counties, plus the city of St. Louis in Missouri (40 CFR 81.18; 40 CFR 81, Appendix A).

Attainment Status. A review of federally published attainment status for Illinois in 40 CFR 81.314 indicated that St. Clair County is designated as attainment (i.e., meeting national standards) for CO, NO₂, SO₂, PM₁₀, and Pb, and nonattainment for the new PM_{2.5} and 8-hour O₃ standards. For O₃, the region was in maintenance status for the old 1-hour standard, having achieved attainment on 13 May 2003. Although the 1-hour O₃ standard was revoked on 15 June 2005, certain control measures remain in place until the 8-hour standard can be fully implemented (personal communication, Kaleel 2005). Control measures in the St. Louis metropolitan area include gasoline vapor recovery systems; controls on industry; centralized inspection of car emissions; the use of cleaner fuels throughout the region; and a range of transportation control measures, which include traffic flow improvement projects, intelligent transportation system and regional ridesharing program (East-West Gateway Council of Governments 2005).

PSD Class I Areas. No mandatory federal PSD Class I areas are located within the ROI. The nearest PSD Class I area is the Mingo National Wildlife Refuge, which is located in southeast Missouri near the town of Puxico, along the Mississippi River, 107 miles south of St. Clair County. Sensitive air quality related values in the bottomland hardwood swamp, which is administered by the USFWS, include vegetation, wildlife, soils, water quality, visibility, odor, and cultural and archaeological resources (National Park Service 2005). Additional PSD Class I areas in the region include the Hercules-Glade Wilderness, which is 210 miles to the southwest

of Scott AFB; the Upper Buffalo Wilderness, which is 216 miles to the south-southwest; and Mammoth Cave National Park, which is 222 miles east of Scott AFB.

Climate. Both the warm moist air from the Gulf of Mexico and the cold air masses originating in Canada affect the climate in southern Illinois, including St. Clair County and the St. Louis metropolitan area. Summers are warm and humid, with temperatures of 90 degrees Fahrenheit (°F) or higher occurring 35 to 40 days per year (with at most five days of 100°F temperatures or more per year). Winter temperatures drop below 0 °F only two or three days per year with temperatures below freezing occurring approximately 25 days per year. Snowfall averages 18 inches per winter. Normal precipitation is approximately 34 inches per year. Winter months are the driest, with March through May being the wettest months of the year. Thunderstorms occur 40 to 50 days per year, with a few each year producing large hail and damaging winds. Winds in St. Clair County average 10 to 12 miles per hour from the south during May through October (National Oceanic and Atmospheric Administration 1998; National Weather Service 2005).

Current Emissions. Air emissions at Scott AFB include those from stationary and mobile sources. The stationary sources include combustion sources, fuel storage and transfer, and operational sources. The mobile sources include vehicles and aircraft operations. Baseline emissions for the Base are presented in Table 3.4-2. In this table, nitrogen oxides (NO_x) include NO₂ and other nitrogen compounds; and sulfur oxides (SO_x) include SO₂ and other sulfur compounds. Because volatile organic compounds (VOCs) and NO_x are precursors to the formation of O₃ in the atmosphere, control of these pollutants is the primary method of reducing O₃ concentrations in the atmosphere. PM₁₀ includes PM_{2.5} and may be used as an upper limit for PM_{2.5} emissions. Scott AFB is a synthetic minor source of air pollution, with a Federally Enforceable State Operating Permit (FESOP) maintaining its potential emissions from stationary sources below major source levels. The permit covers five jet fuel storage tanks equipped with internal floating roofs, diesel emergency power generator and natural gas-fired equipment, a jet engine test cell, 11 gasoline storage tanks, one ethylene glycol storage tank, an indoor shooting range controlled by a bag house (a fabric filter system configured in cylindrical "bags," which remove dust from a gas stream), and one sulfur dioxide generator.

Regional Air Emissions. The previous section lists on-Base emissions for Scott AFB in St. Clair County, Illinois. The NEPA process, however, must also consider impacts from indirect emissions from stationary and mobile sources related to the project, some of which (for example, commuting of new employees to and from the facility) occur outside of the installation. For comparison purposes, Table 3.4.3 lists county-wide emissions for St. Clair County, Illinois, and for AQCR 70 (which includes St. Clair County), as compiled by the USEPA in its National Emissions Inventory (NEI), which was last updated in 1999 (USEPA 2003). The 1999 NEI

contains estimates of annual emissions for stationary and mobile sources of air pollutants in each county.

		ANNUAL EMISSIONS (TPY)				
	СО	VOC	NO _x	SO_x	PM ₁₀	
Abrasive blasting	-	-	-	-	< 1	
Aerospace ground equipment	2	< 1	5	< 1	< 1	
Aircraft operations	411	240	54	16	34	
Asphalt paving operations	-	2	-	-	-	
Degreasing	-	< 1	-	-	-	
External combustion	7	< 1	8	< 1	1	
Fire training	< 1	< 1	< 1	-	< 1	
Fuel cell maintenance	-	< 1	-	-	-	
Fuels dispensing/loading	-	29	-	-	-	
Internal combustion	3	1	13	< 1	< 1	
Jet engine testing	< 1	< 1	< 1	< 1	< 1	
Landfill	< 1	-	-	-	-	
Munitions and firearms	< 1	-	-	-	-	
Paint gun cleaning	-	< 1	-	-	-	
Storage tanks	-	8	-	-	-	
Surface coating	-	< 1	-	-	< 1	
Vehicle emissions	116	10	15	1	1	
Woodworking	-	-	-	-	< 1	
Wet cooling towers	-	-	-	-	< 1	
TOTAL	539	291	96	17	36	

Table 3.4-2. Baseline Emissions at Scott AFB, calendar year 2004

Source: Scott AFB 2005c

	POLLUTANTS (IN TPY)				
СО	VOC	NO _x	SO ₂	PM_{10}	
4,480	7,289	1,369	3,457	14,201	
89,407	6,982	11,260	663	439	
81,499	67,595	144,330	475,626	207,382	
774,394	64,744	104,351	6,133	4,199	
	4,480 89,407 81,499	CO VOC 4,480 7,289 89,407 6,982 81,499 67,595	CO VOC NOx 4,480 7,289 1,369 89,407 6,982 11,260 81,499 67,595 144,330	CO VOC NOx SO2 4,480 7,289 1,369 3,457 89,407 6,982 11,260 663 81,499 67,595 144,330 475,626	

Table 3.4.3. Air Emissions Inventory St. Clair County, Illinois, and AQCR 70Calendar Year 1999

Source: USEPA 2003

3.5 NOISE

3.5.1 DEFINITION OF RESOURCE

Noise is considered to be unwanted sound that interferes with normal activities or otherwise diminishes the quality of the environment. It may be intermittent or continuous, steady or impulsive. It may be stationary or transient. Stationary sources are normally related to specific land uses, e.g., housing tracts or industrial plants. Transient noise sources move through the environment, either along relatively established paths (e.g., highways, railroads, and aircraft flight tracks around airports), or randomly. There is wide diversity in responses to noise that not only vary according to the type of noise and the characteristics of the sound source, but also according to the sensitivity and expectations of the receptor, the time of day, and the distance between the noise source (e.g., an aircraft) and the receptor (e.g., a person or animal).

The physical characteristics of noise, or sound, include its intensity, frequency, and duration. Sound is created by acoustic energy, which produces minute pressure waves that travel through a medium, like air, and are sensed by the ear drum. This may be likened to the ripples in water that would be produced when a stone is dropped into it. As the acoustic energy increases, the intensity or amplitude of these pressure waves increase, and the ear senses louder noise. The unit used to measure the intensity of sound is the decibel (dB). Sound intensity varies widely (from a soft whisper to a jet engine) and is measured on a logarithmic scale to accommodate this wide range. The logarithm, and its use, is nothing more than a mathematical tool that simplifies dealing with very large and very small numbers. For example, the logarithm of the number 1,000,000 is 6, and the logarithm of the number 0.000001 is -6. Obviously, as more zeros are added before or after the decimal point, converting these numbers to their logarithms greatly simplifies calculations that use these numbers.

The frequency of sound is measured in cycles per second, or hertz (Hz). This measurement reflects the number of times per second the air vibrates from the acoustic energy. Low frequency sounds are heard as rumbles or roars, and high frequency sounds are heard as screeches. Sound measurement is further refined through the use of "A-weighting." The normal human ear can detect sounds that range in frequency from about 20 Hz to 15,000 Hz. However, all sounds throughout this range are not heard equally well. Therefore, through internal electronic circuitry, some sound meters are calibrated to emphasize frequencies in the 1,000 to 4,000 Hz range. The human ear is most sensitive to frequencies in this range, and sounds measured with these instruments are termed "A-weighted," and are shown in terms of A-weighted decibels (dBA).

The duration of a noise event, and the number of times noise events occur, are also important considerations in assessing noise impacts.

As a basis for comparison when noise levels are considered, it is useful to note that at distances of about 3 feet, noise from normal human speech ranges from 63 to 65 dB, operating kitchen appliances range from about 83 to 88 dB, and rock bands approach 110 dB.

The word "metric" is used to describe a standard of measurement. As used in environmental noise analysis, there are many different types of noise metrics. Each metric has a different physical meaning or interpretation and each metric was developed by researchers attempting to represent the effects of environmental noise.

The metrics supporting the assessment of noise from aircraft operations associated with the proposals assessed in this document are the maximum sound level (L_{max}), the Sound Exposure Level (SEL), and Time-Averaged Sound Levels. Each metric represents a "tier" for quantifying the noise environment, and is briefly discussed below.

3.5.1.1 Maximum Sound Level

The L_{max} metric defines peak noise levels. L_{max} is the highest sound level measured during a single noise event (e.g., an aircraft overflight), and is the sound actually heard by a person on the ground. For an observer, the noise level starts at the ambient noise level, rises up to the maximum level as the aircraft flies closest to the observer, and returns to the ambient level as the aircraft recedes into the distance. Maximum sound level is important in judging a noise event's interference with conversation, sleep, or other common activities.

This document considers noise from aircraft operating around airfields. Around airfields, the primary operational modes of aircraft are departures (take-offs) and arrivals (landings). Table 3.5-1 shows L_{max} values at various distances associated with typical military and civilian aircraft operating at Scott AFB and MidAmerica Airport.

	L _{MAX} VAL	L _{max} Values (in dBA) At Varying Distances (In Feet)				
Aircraft and Power Type	500	1,000	2,000	5,000	10,000	
KC-135E Takeoff	110.3	101.9	92.3	79.0	68.9	
KC-135E Landing	108.6	100.3	90.1	71.7	58.1	
KC-135R Takeoff	93.9	87.1	79.8	68.9	59.1	
KC-135R Landing	90.4	83.4	75.8	64.4	54.2	
B-727 Takeoff	112.8	106.0	98.8	88.0	78.7	
B-727 Landing	86.6	79.4	71.9	60.8	51.1	
Lear 35 Takeoff	96.6	89.4	81.6	69.7	59.0	
Lear 35 Landing	81.9	74.3	66.1	54.0	44.0	

Table 3.5-1. Representative Maximum Sound Levels

Source: OMEGA108

3.5.1.2 Sound Exposure Level

 L_{max} alone may not represent how intrusive an aircraft noise event is because it does not consider the length of time that the noise persists. The SEL metric combines intensity and duration into a single measure. It is important to note, however, that SEL does not directly represent the sound level heard at any given time, but rather provides a measure of the total exposure of the entire event. Its value represents all of the acoustic energy associated with the event, as though it was present for one second. Therefore, for sound events that last longer than one second, the SEL value will be higher than the L_{max} value. The SEL value is important because it is the value used to calculate other time-averaged noise metrics. Table 3.5-2 shows SEL values corresponding to the aircraft and power settings reflected in Table 3.5-1.

	SEL VALUES (IN DBA) AT VARYING DISTANCES (IN FEET)				(IN FEET)
Aircraft and Power Type	500	1,000	2,000	5,000	10,000
KC-135E Takeoff	113.2	106.6	98.8	87.9	79.6
KC-135E Landing	110.6	104.1	95.7	79.7	67.8
KC-135R Takeoff	97.2	92.2	86.7	78.2	70.2
KC-135R Landing	96.0	90.8	85.0	76.0	97.6
B-727 Takeoff	117.0	112.1	106.7	98.3	90.8
B-727 Landing	92.1	86.8	81.1	72.3	64.5
Lear 35 Takeoff	102.5	97.1	91.1	81.6	72.7
Lear 35 Landing	87.6	81.8	75.4	65.7	57.5

 Table 3.5-2. Representative Sound Exposure Levels

Source: OMEGA108

3.5.1.3 Time-Averaged Cumulative Noise Metrics

The number of times noise events occur during given periods is also an important consideration in assessing noise impacts. The "cumulative" noise metric supporting the analysis of multiple time-varying noise events is the Day-Night Average Sound Level (L_{dn}).

Day-Night Average Sound Level

This metric sums the individual noise events and averages the resulting level over a specified length of time. Thus, it is a composite metric which considers the maximum noise levels, the duration of the events, the number of events that occur, and the time of day during which they occur. This metric adds 10 dB to those events that occur between 10:00 p.m. and 7:00 a.m. to account for the increased intrusiveness of noise events that occur at night when ambient noise levels are normally lower than during the day time. This cumulative metric does not represent the variations in the sound level heard. Nevertheless, it does provide an excellent measure for comparing environmental noise exposures when there are multiple noise events to be considered.

Finally, it should be noted that ambient background noise is not considered in the noise calculations that are presented below. There are two reasons for this. First, ambient background noise, even in wilderness areas, varies widely, depending on location and other conditions. For example, studies conducted in an open pine forest in the Sierra National Forest in California have measured up to a 10 dBA variance in sound levels simply due to an increase in wind velocity (Harrison 1973). Therefore, assigning a value to background noise would be arbitrary. Secondly, and probably most important, is that it is reasonable to assume that ambient background noise in the project's ROI would have little or no effect on the calculated L_{dn} . In calculating noise levels, louder sounds dominate the calculations, and overall, aircraft and other transportation-related noise would be expected to be the dominant noise sources characterizing the acoustic conditions in the region.

Using measured sound levels as a basis, the USAF developed several computer programs to calculate noise levels resulting from aircraft operations. Sound levels calculated by these programs have been extensively validated against measured data, and have been proven to be highly accurate.

In this document, the sound levels calculated for aircraft operations in an airfield environment are all L_{dn} . L_{dn} metrics are the preferred noise metrics of the Department of Housing and Urban Development, the Department of Transportation, the Federal Aviation Administration, the USEPA, and the Veteran's Administration.

Ignoring the night-time penalty for the moment, L_{dn} may be thought of as the continuous or cumulative dBA that would be present if all of the variations in sound level that occur over the

given period were smoothed out so as to contain the same total sound energy. While L_{dn} does provide a single measure of overall noise impact, it is fully recognized that it does not provide specific information on the number of noise events or the specific individual sound levels that occur. For example, an L_{dn} of 65 dB could result from a very few noisy events, or a large number of quieter events. Although it does not represent the sound level heard at any one particular time, it does represent the total sound exposure. Scientific studies and social surveys have found the L_{dn} to be the best measure to assess levels of community annoyance associated with all types of environmental noise. Therefore, its use is endorsed by the scientific community and governmental agencies (American National Standards Institute 1980, 1988; USEPA 1974; Federal Interagency Committee on Urban Noise 1980; Federal Interagency on Noise 1992).

The ROI for the noise analysis are the areas around Scott AFB and MidAmerica Airport that are exposed to aviation-related noise resulting from activities in the region.

3.5.2 EXISTING CONDITIONS

Public annoyance is the most common concern associated with exposure to elevated noise levels. When subjected to L_{dn} levels of 65 dBA, approximately 12 percent of the persons so exposed will be "highly annoyed" by the noise. At levels below 55 dBA, the percentage of annoyance is significantly lower (less than 3 percent), and at levels above 70 dBA, it is significantly higher (greater than 25 percent) (Finegold et al. 1994). Table 3.5-3 shows the percentage of the population expected to be highly annoyed at a range of noise levels.

Noise Exposure (L _{dn} in dBA)	Percent Highly Annoyed
< 65	< 12
65 - 70	12-21
70 - 75	22-36
75 - 80	37 - 53
80 - 85	54 - 70
> 85	> 71

 Table 3.5-3. Percentage of Population Highly Annoyed By Elevated Noise Levels

Source: Finegold et al. 1994

3.5.2.1 Aircraft Activity

The following terms are defined to provide a better understanding of how data are developed for input to the various noise models used to calculate noise.

Around an airfield, *aircraft operations* are categorized as takeoffs, landings, or closed patterns (which could include activities referred to as touch-and-gos or low approaches). Each takeoff or

landing constitutes one operation. A *closed pattern* occurs when the pilot of the aircraft approaches the runway as though planning to land, but then applies power to the aircraft and continues to fly as though taking off again. The pilot then flies a circular or rectangular track around the airfield, and again approaches for landing. In some cases, the pilot may actually land on the runway before applying power, or in other cases the pilot simply approaches very close to the ground. In either event, since a closed pattern operation essentially consists of a landing and a takeoff, it is considered two operations.

Scott AFB and MidAmerica Airport are co-located aviation facilities located near Belleville, Illinois. Scott AFB and its associated runway are situated in the western portion of the complex; MidAmerica Airport is situated to the east. Under current conditions, the two facilities support military and civil aviation activity. Together, the two facilities support approximately 125 daily aviation operations. Considering all types of flight activities, a scenario representing an "average day's" operations was developed. The operations considered include arrivals (landings), departures (takeoffs), and closed patterns. Noise calculations consider the frequency of flight operations, runway utilization, and the flight tracks and flight profiles flown by each aircraft. The numbers and types of representative operations considered are shown in Table 3.5-4.

	Arri	VALS	DEPAF	RTURES	CLOSED P	ATTERNS ²
Aircraft	Day	Night	Day	Night	Day	Night
KC-135E	4.061	0.451	4.505	0.007	12.183	1.354
Other Based Military	7.185	0.526	7.638	0.077	9.914	0
Transient Military	5.007	0.039	5.007	0.039	0	0
Air Carrier / Air Taxi	2.443	0	2.443	0	0	0
General Aviation	5.524	0	5.515	0	13.662	0
Total	24.220	1.016	25.108	0.123	35.759	1.354

 Table 3.5-4. Average Daily Operations at Scott AFB/MidAmerica Airport¹

Notes: 1. Daily operations are based on averages of annual operations; therefore, numbers do not round.

2. Because closed patterns consist of a landing and a takeoff (two aviation operations), the 37.113 closed patterns shown equate to 74.226 aviation operations.

Sources: USAF 2001; personal communication, Newman 2004.

These levels and types of activity are then combined with information on climatology, maintenance activities, and aircraft flight parameters, and processed through the Air Force's BASEOPS/NOISEMAP (Moulton 1990) computer models to calculate L_{dn} . Once noise levels are calculated, they are plotted on a background map in 5-dB increments from 65 dBA to 85 dBA, as applicable. Noise contours associated with current activities at Scott AFB/MidAmerica

Airport are shown in Figure 3.5-1. The land area (in acres) encompassed by each contour is shown in Table 3.5-5.

Sound Level (In L _{dn})	Acres of Land ¹
65 - 70	1736.10
70 - 75	850.09
75 - 80	410.52
80 - 85	193.73
> 85	81.26

Table 3.5-5. Land Area Exposed To Indicated Sound Levels

Note: 1. Land areas exposed to indicated sound levels. Total area exposed to L_{dn} 65 or greater is approximately 3,271.7 acres.
 Source: Wasmer and Maunsell 2002.

3.5.2.2 Ground-Based Activity

Some additional noise results from day-to-day activities associated with operations, maintenance, and the industrial functions associated with the operation of the airport. These noise sources include the operation of ground-support equipment, and other transportation noise from vehicular traffic. However, this noise is generally localized in industrial areas on or near the airfield, or on established routes supporting traffic to-and-from the airfield. Noise resulting from aircraft operations remains the dominant noise source in the airfield region.

3.6 LAND USE AND VISUAL RESOURCES

3.6.1 DEFINITION OF RESOURCE

3.6.1.1 Land Use

Land use classifications reflect either natural or human activities occurring at a given location. Land use resulting from human activities includes residential, commercial, industrial, airfield, recreational, agriculture, and other developed areas. Natural uses include resource production such as forestry, mining, or agriculture, and resource protection such as conservation areas, wildlands, and parks. Management plans, policies, and regulations regulate the type and extent of land use allowable in specific areas and protection specially designated for environmentally sensitive areas. The ROI for land use for the Proposed Action includes the lands of Scott AFB and the 126 ARW installation and adjacent properties in St. Clair County.



Figure 3.5-1. Scott AFB/MidAmerica Airport Baseline Noise Contours

3.6.1.2 Visual Resources

Visual resources refer to the natural and constructed features that give a particular environment its aesthetic qualities. In undeveloped areas, landforms, water surfaces, and vegetation are the primary components that characterize the landscape. Constructed elements such as buildings, fences, and streets may also be visible. These may dominate the landscape or be relatively unnoticeable. Attributes used to describe the visual resource value of an area include landscape character, perceived aesthetic value, and uniqueness.

In developed areas, the natural landscape is likely to provide a background for more obvious constructed features. The size, forms, materials, and functions of buildings, structures, roadways, and infrastructure, along with surrounding landscape features, define the visual context of an area. These features form the overall impression that an observer receives of its visual character. Some urbanized areas or developments prescribe standards or goals for achieving or preserving visual quality. In urban areas, there may be ordinances or zoning provisions that guide physical development.

In non urban contexts, laws (such as the Wilderness Act or the National Wild and Scenic Rivers Act) and management objectives protect scenic quality of some special areas. Federal land managers also clarify the scenic value of lands in accordance with federal land management regulations.

The ROI for visual resources includes all locations within the viewshed (i.e., with line-of-sight) to or from the project areas on the 126 ARW installation at Scott AFB.

- 3.6.2 EXISTING CONDITIONS
- 3.6.2.1 Land Use

Scott AFB and MidAmerica Airport

Scott AFB is 3,589 acres in areal extent, located in a predominantly agricultural portion of western Illinois. The Base is located immediately south of Interstate 64 (I-64) (Figure 1.3-1), near the cities of O'Fallon and Belleville. The Base is adjacent to the MidAmerica Airport. The airport was built as a cooperative effort with Scott AFB, and has a 10,000-foot runway that serves some military customers, in addition to functioning as a commercial passenger and cargo airport. Scott AFB and MidAmerica Airport share runways through Joint Use Agreements, effectively providing a parallel runway system. The 126 ARW is located in the northeastern quadrant of Scott AFB, and generally west of MidAmerica Airport. The 126 ARW installation occupies about 133 acres, and is entirely surrounded by Scott AFB property.

The functional land uses within the 126 ARW parcel include airfield, aircraft operations and maintenance, administrative, industrial, and open space. Immediately to the west is the golf course, to the north is former family housing (now open space) and open recreational uses focused around Scott Lake. Directly to the east are administrative and mission-related aircraft facilities. The airfield is located along the southwest boundary of the 126 ARW parcel. None of the 126 ARW land or facilities is within airfield or munitions safety setback zones or buffers.

The General Plan for Scott AFB indicates that the golf course will be reconfigured using part of the former housing area to the north of the 126 ARW parcel. This area will also be developed with a new golf club house and other outdoor recreational facilities, and with administrative facilities for a new Network Operations facility. Immediately to the east of the 126 ARW, a new headquarters facility will be constructed. All these planned changes are compatible with uses on the 126 ARW parcel (375 AW 2004a).

The 126 ARW parcel is easily accessible from I-64 via the Shiloh or Belleville Gates. Within the Base, Scott School Road leads directly into the 126 ARW installation. The 126 ARW is also on the bus route linking to the light rail system on the south side of Scott AFB. The rail line provides access to St. Louis, 20 miles to the west.

Areas Surrounding the Scott AFB and MidAmerica Airport

Areas surrounding Scott AFB and MidAmerica Airport were historically tall grass prairie. Most of the surrounding suitable land has been converted for agricultural use for several decades. Agricultural land is interspersed with wetlands and wooded areas, and small rural communities. Employment opportunities at the Base and in St Louis have expanded the economic base for these communities, and supported continued growth (375 AW 2004a).

Most of the land immediately adjacent to the Base and airport is within the County of St. Clair. Surrounding municipalities include the City of O'Fallon, the Village of Shiloh, and the City of Mascoutah. The City of O'Fallon has a Comprehensive Plan and growth is being directed toward the northwest, away from the airport area. The Village of Shiloh, to the west, considers airport activities in its zoning process. The City of Mascoutah, to the southeast, has incorporated recommended compatible land use concepts into its zoning. Some residential development on the north edge of Mascoutah (south of MidAmerica Airport) may be a future concern. Growth is being encouraged along the I-64 corridor (375 AW 2004a).

Airport Land Use Planning

The Air Installation Compatible Use Zone (AICUZ) program has established land use compatibility guidelines that are similar to those used by the Federal Aviation Administration. The guidelines have been used by the County and local jurisdictions in planning and zoning to

prevent future incompatible development around the airport complex. Currently, there is some existing residential use within the Base's Accident Potential Zone (APZ) I and residential and quasi-public use in the APZ II (Scott AFB 2004a). Some residences on the northwest side of the airfield complex are exposed to noise levels above 65 L_{dn} .

Because of the economic importance of the Base and MidAmerica Airport, St. Clair County, in cooperation with the Base and surrounding communities, has developed an Airport Environs Overlay Zone (AEOZ) to guide and limit the development of incompatible land uses around the airfield (Scott AFB 2004c). St. Clair County owns the land immediately north of the airport and Base, and serves as a buffer from future encroachment. The County has actively pursued legislation to enable the County to acquire land through eminent domain in order to preclude encroachment on airport facilities and uses. St. Clair County's Future Land Use Plan Sub-Area plan for the Scott-Joint Use Area defines compatible uses for lands outside airport and the military-owned lands. Table 3.6-1 provides information on future land use recommendations from the Sub-Area plan.

Orientation from Scott AFB	Existing Land Use	St. Clair County Future Land Use Recommendation
North	MidAmerica Airport aviation facilities	Maintain aviation use
South	Sparsely populated City of Mascoutah influences land use patterns south of Scott AFB	Continue rural residential, recreational, and industrial uses
East	MidAmerica Airport aviation facilities Highway related commercial uses within I-64/Illinois Route 4 interchange	Continue commercial highway land uses for area around highway interchange
West	Agricultural use immediately west of Air Mobility Drive Further west, a mix of residential and commercial uses associated with Village of Shiloh	Implement/allow regional commercial land use for the area along the west side of Air Mobility Drive

Table 3.6-1. Existing and Recommended Land Uses Surrounding Scott AFB

3.6.2.2 Visual Resources

Scott AFB is located in a rural area in central Illinois. The surrounding land has slightly undulating terrain. The area is characterized by agricultural land, with interspersed natural areas

of wetlands and woods, mostly along natural waterways. The surrounding natural context is typical of the region.

The Base is located on essentially flat terrain, and has the characteristic appearance of an airfield with wide open spaces for the runways and clear zones. Along the airfield are large industrial structures (hangars and aircraft maintenance facilities), and smaller-scaled commercial and administrative facilities behind the "flightline" areas. The administrative and community areas are distinguished by a consistent architectural style of brick facades. A portion of Scott AFB has been designated as an historic district, which provides a unique visual component to the Base; however, this district is well away from the ANG cantonment area. Landscaping is generally informal with mostly mowed grassy areas with trees, and a well-maintained appearance. The overall visual quality of the built environment is cohesive and attractive. Views to and from the airport are limited due to intervening structures, small changes in elevation, and wooded areas.

3.7 SOCIOECONOMICS AND ENVIRONMENTAL JUSTICE

3.7.1 DEFINITION OF THE RESOURCE

Socioeconomic resources are defined as the basic attributes associated with the human environment, particularly population and economic activity. Population is described by the change in magnitude, characteristics, and distribution of people. Economic activity is typically composed of employment distribution, personal income, and business growth. Any impact to these two fundamental socioeconomic indicators can have ramifications for secondary considerations, like housing availability and public service provision.

The ANG's implementing regulation for NEPA is 32 CFR Part 989, et seq. Environmental Impact Analysis, (formerly known as AFI 32-7061). To comply with NEPA, the planning and decision making process for actions proposed by federal agencies involves a study of other relevant environmental statutes and regulations, including EO 12898, Federal Actions to Address Environmental Justice in Minority Populations and Low-Income Populations. The essential purpose of EO 12898 is to ensure the fair treatment and meaningful involvement of all people regardless of race, color, national origin, or income with respect to the development, implementation, and enforcement of environmental laws, regulations, and policies. Fair treatment means that no group of people, including racial, ethnic, or socioeconomic groups, should bear a disproportionate share of the negative environmental consequences resulting from industrial, municipal, and commercial operations or the execution of federal, state, tribal, and local programs and policies. Also included with environmental justice are concerns pursuant to EO 13045, Protection of Children from Environmental Health Risks and Safety Risks. This EO directs federal agencies to identify and assess environmental health and safety risks that may disproportionately affect children under the age of 18. These risks are defined as 'risks to health
or to safety that are attributable to products or substances that the child is likely to come in contact with or ingest.'

The socioeconomic and environmental justice analysis that follows is a component of NEPA compliance. Socioeconomic data are presented for the ROI of St. Clair County, the State of Illinois, and the nation, where information is available. Baseline trends for the County are compared to those at the state and national scale. Consequently, data in this section are presented for the county, state, and national levels. Existing conditions for environmental justice were analyzed through demographic characterization, particularly ethnicity and poverty status for the ROI.

3.7.2 EXISTING CONDITIONS

3.7.2.1 Population and Employment

Scott AFB is located in a rural area in St. Clair County, Illinois. Table 3.7-1 compares the differences in population in the ROI between the 1990 Census, the 2000 Census, and most recent estimates from 2004. St Clair County is growing at a slower rate than the state and nation as a whole, and even had a decline in population between 1990 and 2000. The closest communities to the Base where military employees may live or buy goods and services are O'Fallon, Mascoutah, Shiloh, and Belleville. O'Fallon has a current population of about 26,000. Belleville has a slightly higher population with 41,410 persons in 2000 (United States Census Bureau [USCB] 2000). Belleville is a bedroom suburb to the City of St. Louis. Both Shiloh and Mascoutah are small communities with populations under 10,000 persons.

Location	1990	2000	2004	% change 1990-2000	% change 2000-2004
United States	248,709,873	281,421,906	293,655,404	13.1	4.3
Illinois	11,430,602	12,419,293	12,713,634	8.6	2.3
St Clair County	262,852	256,082	259,132	-2.5	1.2

 Table 3.7-1. Population Changes in the Region

Source: USCB 2000a, 2000b, 2004, 2005

Data from the USCB (Table 3.7-2) indicate that in St. Clair County, which is the smallest geographic area for which labor statistics were analyzed, there were about 129,500 persons (16 years and older) in the labor force in 2004 (USCB 2004). Table 3.7-2 compares the per capita income (PCI) in the ROI, showing that it is almost 20 percent lower in St. Clair County than the State of Illinois.

Geographic area	Per Capita Income, In Dollars, 2003	No. in labor force 2004	% pop. in labor force	
U.S.	31,472	145,437,824	65.9	
Illinois	32,965	6,384,492	67.1	
St Clair County	27,324	129,448	67.0	

Table 3.7-2. Per Capita Income

Source: USCB 2004; Bureau of Economic Analysis 2003

There are about 14,000 persons employed at Scott AFB, including the 126 ARW and other tenants. About 60 percent of these jobs are military and 40 percent civilian. The total Scott AFB community, which is comprised of civilian and military personnel, their dependents, and military retirees, is about 39,500 persons, mostly residing in St. Clair County (Scott AFB 2005d). Total payroll for Scott AFB and tenants in fiscal year (FY) 2003 was about \$970 million with about \$412 million in annual expenditures. This included expenditures of \$42 million for construction projects, \$191 million on materials, equipment and supplies, and \$112 million for services (Scott AFB 2005d).

The 126 ARW (included in the estimates above) currently has a work force of 283 active guard technicians/reserves (less than 1 percent of the St. Clair County work force) and a monthly unit training assembly part-time force of 564 guardsmen (personal communication, Fogarty 2005). The current total annual payroll for the 126 ARW for full-time technicians and reserves is about \$17.4 million. Active and inactive part-time guardsmen payroll is about \$7.6 million.

3.7.2.2 Environmental Justice

In order to provide a thorough environmental justice evaluation, this EA gives particular attention to the distribution of race, poverty, and legal status (under age 18) in areas potentially impacted by implementation of the Proposed Action.

Demographics

Table 3.7-3 displays the comparative statistics for race and Hispanic identification for the ROI. The Black and African American population is higher than the state or nation as a whole, while the Hispanic or Latino population (of any race) is lower.

				RACE				
	ONE RACE							
Geographic area	White	Black or African American	American Indian and Alaska Native	Asian	Native Hawaiian and Other Pacific Islander	Some other race	Two or more races	Hispanic or Latino (of any race)
U.S.	211,460,626	34,658,190	2,475,956	10,242,998	398,835	15,359,073	6,826,228	35,305,818
%	75.1	12.3	0.9	3.6	0.1	5.5	2.4	12.5
Illinois	9,125,471	1,876,875	31,006	423,603	4,610	722,712	235,016	1,530,262
%	73.5	15.1	0.2	3.4	<0.1	5.8	1.9	12.3
St Clair County	173,970	73,666	665	2,322	116	2,040	3,303	5,604
%	67.9	28.8	0.3	0.9	<0.1	0.8	1.3	2.2

Table 3.7-3. Profile of Demographic Characteristics, Year 2000

Note: Percent of total population (column 2) that each group represents is given in parenthesis. Only the percentages under the 'Race' heading will total 100 percent. Hispanic or Latino can be part of any race, and therefore the percent of Hispanic or Latino is percent of total population.

Source: USCB 2000a, 2000b

Poverty and Legal Status

Table 3.7-4 compares the percent of persons under the age of legal consent (age 18). The areas within the ROI have populations of persons under age 18 that mimic the state and national means.

Geographic area	Percent Under Age 18, 2000
U.S.	25.7
Illinois	26.1
St Clair County	27.7
C	

Table 3.7-4. Persons under Age 18 in the ROI in the Year 2000

Source: USCB 2000a, 2000b

Table 3.7-5 compares poverty at all geographic levels for both individuals and persons under age 18. Poverty in the year 2000 was defined as an income of \$8,794 in a household of one individual, or \$17,603 for a family of four (USCB n.d.). St Clair County has a slightly higher percentage of individuals living below the poverty level (14.5 percent) than the nation as a whole (12.4 percent) and the state of Illinois (10.7 percent). This trend is somewhat greater for persons under the age of 18, with 21.6 percent of this population living below poverty compared to 16.1 percent for the nation and 14.0 percent for the State of Illinois.

Geographic area	Percent Individuals Below Poverty Level	Percent Persons Under Age 18 Below Poverty Level		
U.S.	12.4	16.1		
Illinois	10.7	14.0		
St Clair County	14.5	21.6		

 Table 3.7-5. Individuals in Poverty, Reported in the Year 2000

Source: USCB 2000a, 2000b

3.8 CULTURAL RESOURCES

3.8.1 DEFINITION OF THE RESOURCE

Cultural resources are historic districts, sites, buildings, structures, or objects considered important to a culture, subculture, or community for scientific, traditional, religious or other purposes. They include archaeological resources, historic architectural/engineering resources, and traditional resources. Cultural resources that are eligible for listing in the NRHP are called historic properties. Historic properties are evaluated for potential adverse impacts from an action. In addition, some cultural resources such as American Indian sacred sites or traditional resources may not be historic properties but they are also evaluated under NEPA for potential adverse effects from an action. These resources are identified through consultation with appropriate American Indian or other interested groups. In 1999, the DoD promulgated its American Indian and Alaska Native Policy emphasizing the importance of respecting and consulting with tribal governments on a government-to-government basis. The Policy requires an assessment, through consultation, of the effects of proposed DoD actions that may have the potential to significantly affect protected tribal resources, tribal rights, and Indian lands before decisions are made by the armed services.

The ROI for cultural resources is the area within which the Proposed Action has the potential to affect existing or potentially occurring archaeological, architectural, or traditional cultural resources. For the Proposed and Alternative actions, the ROI is defined as each project's footprint, including any areas that could be used temporarily for staging or other project-related activities.

3.8.2 EXISTING CONDITIONS

3.8.2.1 Historical Setting

Native Americans have likely inhabited south central Illinois since the continental glaciers began receding northward at the end of the last glacial period, approximately 12,000 years before present (BP) (Illinois State Geological Survey 2005; Fagan 1991). Based on the archaeological record, this time span is divided into temporal periods: Paleoindian (12,000-10,000 years BP); Early, Middle, and Late Archaic (10,000-2,800 years BP); Early, Middle, and Late Woodland (2,800-900 years BP); Early and Late Mississippian (1,100-500 years BP); and Protohistoric (500-330 years BP) (Scott AFB 2003a).

The Paleoindians are thought to have been big game hunters, ranging throughout large regions, subsisting on now-extinct Pleistocene fauna such as mammoth and mastodon (Scott AFB 2003a). Although they no doubt supplemented their diet by gathering various plant species, such organic items are not often well preserved in the archaeological record. Instead, these hunters are best known through the non-organic artifacts they left behind, principally, projectile points. There are technological distinctions among these projectile points that are likely indicative of cultural divisions and possibly the specialization toward hunting, particular large game animals (Fagan 1991). Paleoindian populations of the area seem to have preferred uplands, unlike the later Native American cultural manifestations that centered around the bottom lands of the Mississippi River (Scott AFB 2003a). Due to factors such as antiquity and rising sea levels, the remains of these cultures are sparse, but are found contemporaneously throughout North and South America. These early cultures remain enigmatic, fueling contention about the initial inhabitants of the New World (Fagan 1991).

As the North American climate became warmer and drier, the large mammals the Paleoindians relied upon became extinct (Scott AFB 2003a). As a result, inhabitants focused on different game species and increased their reliance on plant resources. This was the start of the Archaic Period throughout much of North America (Fagan 1991). Archaeological evidence suggests that Archaic groups engaged in a mobile way of life, living in lower elevation camps along the Mississippi floodplain in the summer and moving between a series of upland hunting camps following game movements during the rest of the year (Scott AFB 2003a). The Archaic Period is divided into early, middle, and late sub-periods that should be viewed as a transitional continuum between the Paleoindian Period and the following Woodland Period (Fagan 1991).

Like the Archaic, the Woodland Period is also divided into early, middle, and late sub periods. The beginning of the Woodland Period (Early) is generally marked by the appearance of ceramics and is by and large indistinguishable from the Late Archaic. The Middle Woodland Period is witness to the rise of incipient agriculture, and a shift toward larger population centers with possible burial mounds (Scott AFB 2003a). This was the time of the Hopewell culture that centered in two main areas: along the Ohio Valley in the east, and along the Mississippi and Illinois rivers in the west. Both areas appear to have been centers to well developed trade networks that eventually grew together (Fagan 1991). This is evidenced by artifacts of exotic materials such as shark teeth and sea shells from the Gulf and East coasts and obsidian from the Yellowstone area. The Hopewell tradition is also known for elaborate mortuary customs and enigmatic earthworks, some of which contain burials. For reasons probably related to carrying capacity and unchecked population growth, Hopewell organization faltered approximately 1,600 years BP (Fagan 1991). Trade networks collapsed and art styles diversified, lacking the previous regional continuity. By the end of the Late Woodland Period, Native American groups transitioned to bow and arrow technology and were living in large central villages, relying substantially on maize agriculture.

Unlike the preceding Archaic Period, the Mississippian Period is only divided into Late and Early sub periods. The Early Mississippian was a continuation of the Late Archaic with an intensified reliance on maize agriculture. The earlier development of ceramics now allowed for the storage of crops that previously had to be consumed within a relatively narrow window after harvest (Scott AFB 2003a). Square, single-post structures were adopted which began to transition to wall trench structures toward the end of the Early Mississippian. Settlements grew in size, tending to cluster around a central plaza.

The Late Mississippian is best known through the Cahokia Site, less than 20 miles from Scott AFB (Scott AFB 2003a). It is the largest known archaeological site within the borders of the U.S. and likely represents the most complex social and political organization in North America outside of the cultures of Mesoamerica. Cahokia was the center of a vast civilization that included specialized outlying camps, villages, and major and minor chiefdoms that flourished for nearly 600 years. At its zenith (between 950 – 750 years BP), it occupied more than five square miles and had a population believed to be in excess of 30,000. Like the previous Hopewell tradition, the Mississippian tradition focused on agriculture in the fertile river floodplain, but to a more diverse and intensive level (Fagan 1991). When European explorers reached Cahokia, they found it unoccupied.

The Protohistoric Period refers to the time between the indirect influence of non-native trade goods and diseases, and actual direct contact with European groups. The duration of the Protohistoric period varies greatly throughout North America, lasting as little as five minutes to as long as 250 years (Fagan 1991). Epidemics and new technologies such as firearms and horses disrupted local populations to the extent that it is difficult to discern tribal territories. In the 1700s Illinois contained representatives from numerous tribes, including the Miami, Illinois, Fox, Kickapoo, Shawnee, and Mascouten (Scott AFB 2003a).

The Historic Period (in the area of Scott AFB) began with the arrival of French explorers in 1673 when the region was controlled by the Illinois Confederacy (Scott AFB 2003a). It took a little more than 150 years to relocate local Native American populations to the Oklahoma reservations (Scott AFB 2003a).

After first being settled by the French, the area passed to British control after the 1763 Treaty of Paris, stripping the French of all possessions in North America. Later, after the U.S. gained independence from Great Britain, Cahokia was established as the county seat of St. Clair County in 1790 (Scott AFB 2003a). Eight miles south of Scott AFB, the town of Belleville was founded in 1814. Settlers to the area were largely Virginian slave holders as well as a sizable number of German immigrants (Scott AFB 2003a). The City of Belleville grew steadily through local agriculture and industries such as coal mining, beer brewing, flour milling, and stove manufacturing, a few of which remain important to the current local economy (Scott AFB 2003a).

Named for Corporal Frank S. Scott, the first enlisted man to die in an aircraft accident, Scott AFB started as a Midwest training base and flying field in 1917 (Scott AFB 2005e). Flying instruction began on September 11, 1917, with most training using the eight-cylinder Curtiss JN-3D "Jennie" (Global Security 2005a). Given the dangerous nature of early aviation and the need to attend to and transport injured aviators, Scott personnel modified two Jennies into air ambulances that were first used on August 24, 1918. The current mission of Scott AFB has strong ties to its history of aero medical evacuation. In November of 1918, World War I ended and activity at the field decreased substantially.

Initially leased from Shiloh Valley Township, the area of Scott Field was purchased by the War Department in 1919 for \$119,285.84 (Scott AFB 2003a). Two years later, Scott Field became a Lighter Than Air (LTA) station for the research and development of the technology. This was accompanied by the construction of several associated structures including an airship hangar, a helium storage and repurification plant, and a 176-foot high mooring mast (Scott AFB 2003a). Given the explosive nature of helium, several airship disasters turned public opinion as well as military thoughts against LTA aviation. As a result, the Air Corps Balloon and Airship School was deactivated in 1928. With a lack of funds needed to maintain existing airships, LTA activities at Scott Field officially ended in 1937 (Scott AFB 2003a).

Between 1938 and 1942, the Base area was increased to almost three times its original size. Adding numerous facilities, Scott Field was one of the primary training facilities for the Army Air Corps. Training mostly radio operator mechanics, Scott had produced more than 77,000 technicians by mid 1945 (Scott AFB 2003a). Scott Field was renamed Scott AFB in 1947 with the separation and redesignation of the Army Air Force as the USAF. During the next two decades, Scott AFB continued its emphasis on training and began a collaborative relationship

with the ANG. During this time it also accepted increased aeromedical duties, becoming the U.S. headquarters for aeromedical evacuation (Scott AFB 2003a).

Today the Base occupies 3,589 acres, with over 39,000 people living and/or working on Base (Scott AFB 2005d). The primary mission of Scott AFB is global mobility. It is responsible for all air mobility operations as well as providing U.S. aeromedical evacuation. Additionally, Scott AFB provides operational support for airlift and air refueling capabilities (Scott AFB 2005d).

A relatively recent tenant of the Base, the 126 ARW relocated to Scott AFB from Chicago's O'Hare International Airport in 1999. The mission of the 126 ARW is to provide refueling support to U.S. military forces and those of allied nations, as well as airlift missions (Global Security 2005b).

3.8.2.2 Identified Cultural Resources

Archaeological Resources

Cultural resource management formally began at Scott AFB in 1975, but it was not until 1986 that a cultural resources program was established. Since that time, numerous cultural resources surveys have been performed as well as test excavations at five historic archaeological sites. These efforts have identified 12 archaeological sites and two historic cemeteries. All of the sites are historic, although Native American artifacts are represented at only two of the sites. Only one of the 12 sites was evaluated as NRHP-eligible; following impact mitigation, it was destroyed through construction. As a result, there are no known NRHP-eligible archaeological resources at Scott AFB. Most of the Base has been surveyed or is known to be heavily disturbed through the construction and demolition efforts related to the growth period during and after WWII. The Integrated Cultural Resources Management Plan (2003) has identified parcels where more archaeological work may be required, but none are near any projects of the Proposed Action or Alternative Action.

Historic Architectural Resources

Scott AFB is home to the Scott Field Historic District that is made up of 104 contributing historic buildings and structures. The buildings and structures were inventoried and evaluated in 1992 by Thomason and Associates of Nashville, Tennessee (Scott AFB 2003a). Thomason and Associates also completed the NRHP district nomination that was approved by the Illinois SHPO in 1993 and the National Park Service in 1994. A 1994 evaluation of potentially eligible historic Cold War era resources examined 59 structures. None of the evaluated structures were recommended as eligible at the time; however it was recommended that Building 3200 (ANG Alert Hangar) be reevaluated in 2002, on the 50th anniversary of its build date. In December of 2002, Building 3200 was evaluated as eligible for the NRHP under Criterion C.

Resources in the Vicinity of the 126 ARW Installation

The Proposed Action and Alternative Action involve up to 11 projects at specified locations (Section 2.2). Six of the 11 projects described in Sections 2.2 and 2.3 would undergo alteration or demolition (Table 3.8-1).

Building Number, Function	Build Date	NRHP Eligibility	Project/Action
5542 Multi-vehicle Carport	1951	Unevaluated	Facility Demolition
5010 Communication Facility	2000	Not eligible	Addition/Alteration to Communication Facility
3901 Fire Station	2001	Not eligible	Addition/Alteration to Mid-Field Fire Station
3901 Fire Station	2001	Not eligible	Further Addition to Mid- Field Fire Station
5028 Deployment Processing Center	2001	Not eligible	Addition/Alteration to Deployment Processing Center
5038 Pump House	2001	Not eligible	Construct Pump House Spill Containment

Table 3.8-1. Buildings in ROI of 126 ARW Projects

Traditional Resources

No traditional resources or Native American issues have been identified at Scott AFB (Scott AFB 2003a). Although there are no reservations in the State of Illinois, the Kaskaskia and the Kickapoo have judicially established lands near Scott AFB.

3.9 SAFETY

3.9.1 DEFINITION OF THE RESOURCE

This section addresses ground, explosive, and flight safety associated with activities conducted by the 126 ARW installation. Ground safety considers issues associated with human activities and operations and maintenance activities that support unit operations. A specific aspect of ground safety addresses anti-terrorism/force protection (AT/FP) considerations. Explosive safety addresses the management and use of ordnance or munitions associated with installation operations and training activities. Flight safety considers aircraft flight risks such as aircraft accidents. The ROI for safety is the aviation facilities that support 126 ARW operations and the lands immediately adjacent to these facilities.

3.9.2 EXISTING CONDITIONS

3.9.2.1 Ground Safety

Day-to-day operations and maintenance activities conducted by the 126 ARW are performed in accordance with applicable USAF safety regulations, published Air Force Technical Orders, and standards prescribed by Air Force Occupational Safety and Health requirements.

The 375 AW (host unit) fire department responds to all emergencies on Scott AFB. All required emergency response equipment is available, and no waivers are in effect (personal communication, Berridge 2005). All ILANG facilities are equipped with required automatic fire suppression systems, and no waivers are in effect (personal communication, Berridge 2005). There are no unidentified Clear Zone, APZs, or other airfield encroachments on the installation or MidAmerica Airport (personal communication, Pulse 2005). There are 15 airfield clearance violations that require waivers. All have been granted or are pending. There are eight other clearance violations for which permanent exemptions have been granted (USAF 2004).

3.9.2.2 Explosives Safety

The 126 ARW stores, maintains, and uses a range of munitions required for day-to-day performance of their mission. All ordnance is handled and stored in accordance with USAF explosive safety directives (AFI 91-201), and all munitions maintenance is carried out by trained, qualified personnel using USAF-approved technical procedures. All other ordnance is stored by the 375 AW. There are no explosive safety waivers in effect. If explosive ordnance disposal support is required, it is provided by active duty units on Scott AFB (personal communication, Berridge 2005).

3.9.2.3 Anti-Terrorism/Force Protection

As a result of terrorist activities, the DoD and the USAF have developed a series of AT/FP guidelines for military installations. These guidelines address a range of considerations that include access to the installation, access to facilities on the installation, facility siting, exterior design, interior infrastructure design, and landscaping (Unified Facilities Criteria 2003; USAF n.d.). The intent of this siting and design guidance is to improve security, minimize fatalities, and limit damage to facilities in the event of a terrorist attack.

Many military installations, such as the 126 ARW facilities, were developed before such considerations became a critical concern. Thus, under current conditions, many units are not

able to comply with all present AT/FP standards. However, as new construction occurs, it would incorporate these standards, and as facilities are modified, AT/FP standards would be incorporated to the maximum extent practicable.

3.9.2.4 Flight Safety

The primary public concern with regard to flight safety is the potential for aircraft accidents. Such mishaps may occur as a result of mid-air collisions, collisions with manmade structures or terrain, weather-related accidents, mechanical failure, pilot error, or bird-aircraft collisions. Flight risks apply to all aircraft; they are not limited to the military. Flight safety considerations addressed include aircraft mishaps and bird-aircraft strikes.

Aircraft Mishaps

The USAF defines four major categories of aircraft mishaps: Classes A, B, C, and E, which includes High Accident Potential (HAP). Class A mishaps result in a loss of life, permanent total disability, a total cost in excess of \$1 million, or destruction of an aircraft. Class B mishaps result in total costs of more than \$200,000 but less than \$1 million, and result in permanent partial disability or inpatient hospitalization of three or more personnel. Class C mishaps involve reportable damage of more than \$20,000, but less than \$200,000; an injury resulting in any loss of time from work beyond the day or shift on which it occurred, or occupational illness that causes loss of time from work at any time; or an occupational injury or illness resulting in permanent change of job. HAP events are any hazardous occurrence that has a high potential for becoming a mishap. Class C mishaps and HAP, the most common types of accidents, represent relatively unimportant incidents because they generally involve minor damage and injuries, and rarely affect property or the public (USAF 2004). This EA will focus on Class A mishaps because of their potentially catastrophic results.

Based on historical data on mishaps at all installations, and under all conditions of flight, the military services calculate Class A mishap rates per 100,000 flying hours for each type of aircraft in the inventory. It should be noted that these mishap rates do not consider combat losses due to enemy action. In evaluating this information, it should be emphasized that data presented are only statistically predictive. The actual causes of mishaps are due to many factors, not simply the amount of flying time of the aircraft.

The 126 ARW operates KC-135E aircraft. Since entering the USAF inventory, C-135-type aircraft have flown more than 12,347,700 hours. During this time, C-135-type aircraft have experienced 79 Class A mishaps. These data reflect a Class A mishap rate per 100,000 flying hours of 0.64 (USAF Safety Center 2005).

Considering this rate, the probability of a C-135-type aircraft being involved in a Class A mishap is 0.000006. In fact, the 126 ARW has only experienced one Class A mishap since the unit began flying these aircraft. The mishap occurred when the unit was flying out of Chicago, Illinois in 1982 (personal communication, Pulse 2005).

Bird-Aircraft Strike Hazards

Bird-aircraft strikes constitute a safety concern because of the potential for damage to aircraft or injury to aircrews or local populations if an aircraft crash should occur in a populated area. Aircraft may encounter birds at altitudes of 30,000 feet MSL or higher. However, most birds fly close to the ground. Over 94 percent of reported bird strikes occur below 3,000 feet above ground level (AGL). Approximately 50 percent of bird strikes happen in the airport environment, and almost 15 percent occur during low-altitude flight training and use of weapons ranges (USAF Bird-Aircraft Strike Hazard [BASH] Team 2005).

Migratory waterfowl (e.g., ducks, geese, and swans) are the most hazardous birds to low-flying aircraft because of their size and their propensity for migrating in large flocks at a variety of elevations and times of day. Waterfowl vary considerably in size, from one to two pounds for ducks, five to eight pounds for geese, and up to 20 pounds for most swans. There are two normal migratory seasons, fall and spring. Waterfowl are usually only a hazard during migratory seasons. These birds typically migrate at night and generally fly between 1,500 to 3,000 feet AGL during the fall migration and from 1,000 to 3,000 feet AGL during the spring migration.

Along with waterfowl, raptors, shorebirds, gulls, herons, and songbirds also pose a hazard. In considering severity, the results of bird-aircraft strikes in restricted areas show that strikes involving raptors result in the majority of Class A and Class B mishaps related to bird-aircraft strikes. Raptors of greatest concern are vultures and red-tailed hawks. Peak migration periods for raptors, especially eagles, are from October to mid-December and from mid-January to the beginning of March. In general, flights above 1,500 feet AGL would be above most migrating and wintering raptors.

The potential for bird-aircraft strikes is greatest in areas used as migration corridors (flyways) or where birds congregate for foraging or resting (e.g., open water bodies, rivers, and wetlands).

While any bird-aircraft strike has the potential to be serious, many result in little or no damage to the aircraft, and only a minute portion result in a Class A mishap. During the years 1985 to 2004, the USAF BASH Team documented 62,536 bird strikes. Of these, 25 resulted in Class A mishaps where the aircraft was destroyed. These occurrences constituted approximately 0.04 percent of all reported bird-aircraft strikes (USAF BASH Team 2005).

A wildlife strike hazard does exist at Scott AFB/MidAmerica Airport. The facilities are surrounded by habitat attractive to both avian and large and small mammalian species Avian species of notable concern include a variety of waterfowl, due to the installation's proximity to the Mississippi River flyway, large flocks of wintering blackbirds, and starlings. Mammalian species that may be present on the aviation facilities include deer, fox, coyotes, rabbits, and rodents (375 AW 2004b).

The management and control of wildlife hazards on Scott AFB and MidAmerica Airport is considered a joint responsibility, shared by the 375 AW and the MidAmerica Airport Director of Operations. Wildlife risk is mitigated through avoidance, environmental and habitat control, dispersal, removal, and depredation as required. A federal depredation permit from the USFWS is required prior to killing any protected birds, with the exception of European Starlings, House Sparrows, Rock Doves, and domestic pigeons which are not federally protected. It should be noted that the inadvertent "taking" of a protected species as a result of a bird-aircraft strike is exempt from any permit requirements (375 AW 2004b).

When the presence of wildlife increases in the vicinity of the runways, certain flight activities may be modified, or completely curtailed until the risk associated with wildlife presence is reduced to acceptable levels (375 AW 2004b). Since the 126 ARW has been operating from Scott AFB/MidAmerica Airport, the unit has experienced an average of less than eight bird-strikes per year. None have been serious (personal communication, Pulse 2005).

3.10 SOLID AND HAZARDOUS MATERIALS AND WASTE

3.10.1 DEFINITION OF THE RESOURCE

This section describes the affected environment associated with hazardous materials and petroleum products, hazardous and petroleum wastes, Environmental Restoration Program (ERP) sites, and solid waste at the construction, renovation, and demolition areas.

The terms "hazardous materials" and "hazardous waste" refer to substances defined as hazardous by the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA) and the Solid Waste Disposal Act (SWDA), as amended by the Resource Conservation and Recovery Act (RCRA). In general, hazardous materials include substances that, because of their quantity, concentration, or physical, chemical, or infectious characteristics, may present substantial danger to public health or the environment when released into the environment. Hazardous wastes that are regulated under RCRA are defined as any solid, liquid, contained gaseous, or semisolid waste, or any combination of wastes that either exhibit one or more of the hazardous characteristics of ignitability, corrosivity, toxicity, or reactivity, or are listed as a hazardous waste under 40 CFR Part 261. Petroleum products include petroleum-based fuels, oils, and their wastes. The ERP is a DoD program to identify, characterize, and remediate environmental contamination from past activities at DoD installations.

Issues associated with hazardous material and waste typically center around waste streams, underground storage tanks (USTs), aboveground storage tanks (ASTs), and the storage, transport, use, and disposal of pesticides, fuels, lubricants, and other industrial substances. When such materials are improperly used in any way, they can threaten the health and well being of wildlife species, habitats, and soil and water systems, as well as humans. This section also considers solid waste.

The management of hazardous materials and hazardous waste is governed by specific environmental statutes. The key regulatory statutes include:

Comprehensive Environmental Response, Compensation and Liability Act of 1980 (42 USC 9601–9675) as amended by the Superfund Amendments and Reauthorization Act (SARA) of 1986. CERCLA/SARA regulates the prevention, control, and compensation of environmental pollution.

Community Environmental Response Facilitation Act of 1992 (CERFA) (42 USC 9620). This act amended CERCLA to require that, prior to termination of federal activities on any real property owned by the federal government, agencies must identify real property where hazardous substances were stored, released, or disposed of.

Emergency Planning and Community Right-to-Know Act (EPCRA) of 1986 (42 USC 11001–11050). EPCRA requires emergency planning for areas where hazardous materials are manufactured, handled, or stored and provides citizens and local governments with information regarding potential hazards to their community.

Resource Conservation and Recovery Act of 1976 (42 USC 6901–6992). RCRA established standards and procedures for handling, storage, treatment, and disposal of hazardous waste.

Federal Facility Compliance Act (FFCA) of 1992 (Public Law [P.L.] 102-426). This act provides for a waiver of sovereign immunity on the part of federal agencies with respect to federal, state, and local requirements relating to RCRA solid and hazardous waste laws and regulations.

Federal Insecticide, Fungicide, and Rodenticide Act (FIFRA) of 1996 (7 USC 136 et seq.). FIFRA provides federal control of pesticide distribution, sale, and use. It also provides certification criteria for pesticide applicators, including contractors. *Pollution Prevention Act of 1990 (42 USC 13101–13109).* This act encourages minimization of pollutants and waste through changes in production processes.

USEPA Regulation on Identification and Listing of Hazardous Waste (40 CFR Part 261). This regulation identifies solid wastes subject to regulation as hazardous and to notification requirements under RCRA.

USEPA Regulation on Standards for the Management of Used Oil (40 CFR Part 279). This regulation delineates requirements for storage, processing, transport, and disposal of oil that has been contaminated by physical or chemical impurities during use.

USEPA Regulation on Designation, Reportable Quantities, and Notification (40 CFR Part 302). This regulation identifies reportable quantities of substances listed in CERCLA and sets forth notification requirements for releases of those substances. It also identifies reportable quantities for hazardous substances designated in the CWA.

Air Force Policy Directive (AFPD) 32-70, Environmental Quality, establishes the policy that the USAF is committed to environmentally sound practices. These include the following:

- Cleaning up environmental damage resulting from its past activities.
- Meeting all environmental standards applicable to its present operations.
- Planning its future activities to minimize environmental impacts.
- Managing responsibly the irreplaceable natural and cultural resources it holds in public trust.
- Eliminating pollution from its activities wherever possible.

AFPD 32-70 and AFI 32-7000 series incorporate the requirements of all Federal regulations, other AFIs, and DoD directives for the management of hazardous materials, hazardous wastes, and special hazards.

The ROI for hazardous materials, hazardous waste, and petroleum products encompasses areas that could be exposed to an accidental release of hazardous substances from the construction, renovation, or demolition activities. Therefore, the ROI for this section is defined as the boundary of the 126 ARW parcel at Scott AFB.

3.10.2 EXISTING CONDITIONS

The 375 CES/CEV is responsible for the implementation of hazardous material and waste plans at Scott AFB. In conformance with the policies established by AFPD 32-70, the 375 CES/CEV has developed procedures and plans to manage hazardous wastes, hazardous materials, special wastes, and environmental restoration sites on Scott AFB.

3.10.2.1 Hazardous Materials and Petroleum Products

Throughout the USAF, hazardous materials are managed in accordance with AFI 32-7086. This instruction establishes procedures and standards that govern the management of hazardous materials. It applies to all USAF personnel who authorize, procure, issue, use, or dispose of hazardous materials, and to those who manage, monitor, or track any of those activities. The 375 CES/CEV manages hazardous materials in accordance with AFI 32-7086.

Hazardous materials and petroleum products are used throughout the installation for various functions, including aircraft refueling, maintenance, and washing; vehicle maintenance and washing; POL distribution and management; facilities maintenance and repair; maintenance of ground support equipment; and aircraft support operations. Hazardous materials used in these functions include fuels and lubricating oils, solvents, paints and thinners, antifreeze, deicing compounds, and acids. At Scott AFB, hazardous materials are managed through a centralized Base Hazardous Material (HAZMAT) Pharmacy using an Environmental Management Information System, which tracks the inventory and acquisition of hazardous materials along with hazardous waste disposal and health and safety information (Air Force Institute for Environment, Safety, and Occupational Health Risk Analysis [AFIERA] 2002).

The Base *Spill Prevention Control and Countermeasure (SPCC) Plan* (Scott AFB 2001a) provides guidance on hazardous material and petroleum storage, spill prevention measures, and contingency procedures including spill containment and cleanup. This plan establishes responsibilities for handling fuels and other hazardous fluids, containing and recovering spills, spill training, and spill reporting procedures. Potential pollutants stored at the installation include JP-8 aviation fuel, #2 fuel oil, gasoline, and diesel fuel, which are all stored in aboveground storage tanks and associated distribution systems. In addition, smaller amounts of paints, thinners, lubricants and other industrial chemicals are stored and handled in various buildings. The fuel storage facility is located in the southwest portion of the installation and constitutes the major fuel storage capacity at the installation. However, the 126 ARW maintains a fuel storage facility within the ROI which is covered by the Scott AFB SPCC.

3.10.2.2 Hazardous and Petroleum Wastes

Hazardous wastes are managed through the Base level Hazardous Waste Management Plan. This Plan is currently being revised by the 375 CES/CEV in accordance with AFI 32-7042, Solid and Hazardous Waste Compliance (AFIERA 2002). The Hazardous Waste Management Plan provides guidance to Scott AFB personnel (including tenants such as the 126 ARW) on the handling, storage, and disposal of hazardous materials and this plan will implement the "cradle-to-grave" management control of hazardous waste as mandated by USEPA.

Some of the hazardous wastes generated at Scott AFB include spent solvents, photofixer, waste oils, waste cleaning compounds, and various forms of waste paint. The Base Hazardous Waste Management Program covers the handling of universal wastes such as batteries, pesticides, mercury thermostats, and mercury-containing lamps and various special wastes including potentially infectious medical wastes, industrial process wastes, and pollution control wastes.

The 126 ARW is regulated as a large quantity generator and maintains USEPA identification number IL7570024177. There are approximately 50 different satellite accumulation points where hazardous wastes are collected on Scott AFB. Of the 50 points, 23 are managed by the 126 ARW. Building 3306 serves as the central accumulation site for all wastes generated on Scott AFB. This central accumulation site is managed and operated by the 375 CES/CEV. The 126 ARW does not operate a hazardous waste storage facility on Scott AFB and it is the policy of the 126 ARW to ship hazardous wastes offsite as expeditiously as possible.

Asbestos-containing materials (ACM) are managed by the USAF through the implementation of AFI 32-1052, Facilities Asbestos Management. This instruction provides direction for the management of asbestos and ACM at USAF installations. This instruction requires installations to develop an asbestos management plan for the purpose of maintaining a permanent record of the condition and status of ACM in buildings and other facilities on the installation, as well as documenting asbestos management efforts. In addition, the instruction requires the development of an asbestos operating plan. This plan describes how the installation maintains compliance with the AFI for asbestos-related projects. However, the plan further notes that USEPA policy is to leave asbestos in place if disturbance or removal could pose a threat to human health or the environment.

The 126 ARW maintains compliance with the requirements of AFI 32-1052 through the Scott AFB Asbestos Management Plan (Scott AFB 2000a) and the asbestos operations plan (Scott AFB 2000b). This management plan describes procedures for the removal, encapsulation, enclosure, and repair activities associated with ACM-abatement projects. The objective of the plan is to reduce the potential of exposure to potentially hazardous levels of airborne asbestos fibers and assist in maintaining compliance with all Federal, state, and local asbestos regulations.

Lead-based paint (LBP) is regulated through the residential Lead-Based Paint Hazard Reduction Act of 1992. Subtitle B, Section 408 regulates the use and disposal of LBP on federal facilities. Federal agencies are required to comply with applicable federal, state, and local laws and regulations relating to LBP activities and hazards.

USAF policy (USAF 1993) requires each installation to develop and implement a facility management plan for identifying, evaluating, managing, and abating LBP hazards. The *Lead-Based Paint Management Plan* (Scott AFB 1996) provides a basic approach to LBP management. The Plan covers designation of responsibilities, identification of hazards, testing procedures, abatement methods, training requirements, and protection of citizens and workers. The Plan also addresses lead exposure from other sources such as lead soldered fittings used in the potable water system and occupational exposure to lead through corrosion control, welding, and cable maintenance operations. The mitigation and monitoring of LBP, disposal, and other hazards are also discussed.

3.10.2.3 Environmental Restoration Program Sites

The ERP, formerly known as the Installation Restoration Program, is a subcomponent of the Defense Environmental Restoration Program that was promulgated as law under SARA. The ERP requires each DoD installation to identify, investigate, and remediate hazardous waste release and disposal sites.

In 1995, Preliminary Assessments/Site Investigations completed at Scott AFB identified 16 ERP sites (Parsons 1995). Two of these have been closed with no further site remediation planned. However, 17 Areas of Concern (AOCs) are currently under investigation as potential ERP sites (personal communication, McCoy 2005).

The ROI covered by this EA includes the area defined as the boundary of the 126 ARW parcel. Because none of the actions or alternatives involves areas outside of this ROI, ERP sites outside of this area will not be discussed. Within the ROI for this Proposed Action, there are two ERP sites (OT-07 and SS-12) and five AOCs (AOC 6, AOC 14, AOC 18, AOC 22 and AOC 23) (Figure 3.10-1).

Site OT-07 is a former sludge weathering lagoon located in the central to eastern portion of the Base near the northern end of Taxiway "G" (formerly Taxiway "H"). It was constructed in the mid-1970s and used for two years. It was primarily used as a disposal site for tank bottom sludge, and on occasion, other industrial wastes had been disposed in the lagoon. In 1981, the lagoon was closed and the soils were removed to a depth of two feet. The site was backfilled to grade with sand and gravel. A No Further Response Action Planned Decision Document was signed in 1992 by the Base and forwarded to the Illinois EPA. To date, Scott AFB is awaiting concurrence from the Illinois EPA on closure of the site.



Figure 3.10-1. ERP Sites and AOCs on the 126 ARW Installation at Scott AFB, Illinois

Site SS-12 (Building 1197) is the former golf course maintenance/entomology shop located adjacent to the golf course at the corner of Grover Avenue and Golf Course Road in the northern part of the Base. This building was constructed in 1972. Its primary function was mixing pesticides and herbicides for application to the golf course grounds and other areas of the Base. The mixing occurred next to the east fence of the facility, and any spillage would have drained southeast toward a storm drain in the southeast corner of the storage yard. Building 1197 was demolished in 2001 and no investigation or remediation has occurred at this site (Scott AFB 2003b).

The site listed as AOC 6 refers to 20 oil water separators (OWS) on Scott AFB, two of which occur in the ROI near Buildings 3675 (OWS 15) and 3674 (OWS 14). These OWS were in operation from the 1960s to 1994. All of the OWS were replaced and or removed in May 1996 (Scott AFB n.d.). A Preliminary Site Assessment/Site Investigation (PA/SI) has recently been conducted on all 20 of the OWS sites (Scott AFB 2005f).

OWS-15 – This former OWS was situated off the northeast side of Building 3675 (Army Reserve Motor Pool). A drain from the vehicle wash rack flowed to the OWS. During the 1994 PA/SI field effort, three soil borings were completed around the former OWS excavation area; two soil samples and two groundwater samples were collected at discrete depths in each boring. Although several polycyclic aromatic hydrocarbons were detected, all concentrations were below Illinois EPA regulatory limits. No other compounds were detected in groundwater (Parsons 1995). During the recent reconnaissance in 2005, an OWS was not observed and the area of the OWS indicated in the 1994/1995 PA/SI report has been overlain with a relatively new concrete pad. No samples were proposed to be obtained from this site (Scott AFB 2005f).

OWS-14 – This OWS was situated adjacent to former Building 3674. The 1994/1995 PA/SI report stated that Building 3674 was used as a jet engine test facility until 1985 (Parsons 1995). Fuels and solvents, used to flush fuels from the engines, flowed from a floor drain in the building to the OWS. In 1988, the building was converted to a mobile command center equipment storage and maintenance area, and for the storage and maintenance of lawn maintenance equipment. According to the 1994/1995 PA/SI report, at the time of the 1994 PA/SI field activities, the OWS was still in place and had been serviced/emptied once between 1981 and 1994. Six soil borings were completed during the 1994 PA/SI field activities. Two subsurface soil samples and two groundwater samples were collected from each boring. The results of the soil samples indicated the presence of several VOCs. Two groundwater samples collected indicated the presence of trichloroethylene and cis-dichloroethene at concentrations below the USEPA maximum contaminant level. As observed during the recent reconnaissance in 2005, Building 3674

appears to have been razed. Building 5032 appears to have been constructed over the site area. No samples were proposed to be obtained from this site (Scott AFB 2005f).

The sites listed as AOC 14 and 23 are the Cardinal Creek Village – South and North areas respectively. Each of these AOCs cover approximately 22 acres and both were former housing areas originally constructed in 1952 and completed around 1955. Renovations to these areas were conducted in the early 1980s. The south area consisted of approximately 60 buildings along with open spaces and playgrounds. The north area consisted of approximately 72 buildings also with common areas of open spaces and playgrounds. The units in both areas were vacated in 1999 and demolished shortly thereafter. From 1960 to the late 1980s, chlordane was used to treat the buildings for termite control. The treatment involved the injection of a solution beneath and around the foundation and slabs, thus contaminating the soil. In 1997, the Base conducted a PA/SI of both areas to determine the nature and extent of contamination. This study identified both chlordane and lead at concentrations in the soil that exceeded the Illinois EPA Tier 1 soil cleanup guidance (Scott AFB 1997a). As a result of this study, a soil management plan was prepared for the excavation and stockpiling of soils in the south area only in advance of relocating the 126 ARW to this area of Scott AFB (Scott AFB 1997b). In 1998, a Final Decision Document for the excavated soil from the south area was prepared (Scott AFB 1998). This document designated AOC 14 as a CERCLA Area of Contamination and described the selected remedy for the excavated soils. In cooperation with the Illinois EPA, approximately 5,000 cubic yards of soil were excavated from the former housing area and stockpiled into an engineered containment cell located east of Pryor Drive and north of Golf Course Road. Upon completion of the excavation, the soil was capped with an engineered barrier consisting of three feet of clean soil and graded to have the appearance of a landscape berm. This soil will remain at this location until a final action is implemented as part of the ERP (Scott AFB 1998). Regarding AOC 23, no investigation has occurred and soils remain in place. Although a future investigation is planned for this site, it is unknown when this will occur.

The site listed as AOC 18 refers to Basewide former coal storage piles. Coal was the primary source of fuel for the Base from the early 1930s to the mid 1960s. During that time, nearly every building was heated by steam or coal burning furnaces. The 2005 PA/SI recommended that one boring be installed at the location of the former coal storage area around existing Building 5022. In 2005, soil samples obtained from this boring indicated no evidence of coal or other contamination at this location (Scott AFB 2005g). The final report recommended no further action for this site.

AOC 22 consists of two former 25,000-gallon jet fuel ASTs used by the Army Reserve from the 1970s until 1994. These two ASTs were located near Building 3671, north-northeast of the jet engine test stand (former Building 3674), and along the south side of Taxiway G. Base maps indicate that aboveground piping for distributing fuel to fueling trucks was associated with these

ASTs (Scott AFB 2005f). During the recent reconnaissance in 2005, it was observed that Buildings 3671 and 3674 had been razed. (It appears that Building 5032 has been constructed over the Building 3674 area.) Also, a new set of ASTs and a truck fueling area were constructed south of AOC 22. Currently, the AOC 22 area is partially overlain by the taxiway leading to MidAmerica Airport. No remnants of the original ASTs or piping were observed during the 2005 survey. Five surface, ten subsurface and five groundwater samples were proposed to be obtained from this site (Scott AFB 2005f).

3.10.2.4 Solid Waste

The Scott AFB *Solid Waste Management Plan* (SWMP) provides guidance for personnel who work with solid wastes, and sets local management procedures for managing solid waste, preventing pollution, and establishing proper disposal and recycling options (Scott AFB 2001b). The plan incorporates current USEPA, state, and local requirements regarding the management of solid waste as they relate to environmental protection during operations conducted at this installation. Solid wastes, other than construction and demolition waste generated at Scott AFB, are disposed at an on-site recycling facility. Construction and demolition waste are transported to and disposed of at an off site landfill. Solid wastes at Scott AFB consist of regular waste from municipal, office, residential, and industrial sources; yard waste, including grass, brush, tree trimmings, and installation grounds and golf course maintenance; high value metal wastes such as brass casings; and roads and grounds maintenance (Scott AFB 2005h). The goals of Scott AFB for solid waste include minimizing waste generation by reusing and recycling materials whenever possible, and increasing use of materials that are reusable and recyclable. Descriptions of the recycled materials and their amounts are shown in Table 3.10-1. As of 2005, the installation recycled about 40 percent of its non-hazardous solid wastes.

3.11 INFRASTRUCTURE

3.11.1 DEFINITION OF THE RESOURCE

Infrastructure refers to the system of public works, such as transportation and utilities that provide the underlying framework for a community. Transportation and circulation refer to roadway and street systems, the movement of vehicles, pedestrian and bicycle traffic, and mass transit. Utilities include such amenities as water and power supply and waste management.

The infrastructure elements at the 126 ARW installation at Scott AFB include both transportation and utility systems. The ROI for this resource primarily consists of the installation.

Material	FY00	FY01	FY02	FY03	FY04
Scrap Metals	216	215	229	206	280
Aluminum Cans	NA	NA	NA	9	11
Auto, Batteries, Tires	NA	66	56	70	74
Fluorescent Bulbs	5	7	3	3	2
Toner Cartridges	1	3	9	1	6
Pallets	19	31	20	5	48
Glass	38	8	0	83	20
Cardboard/Boxboard	866	319	271	244	387
Office Paper and Books	623	470	454	324	440
White Ledger Paper	NA	NA	NA	99	130
Plastics	31	51	21	25	77
126 ARW	NA	NA	NA	16	0
ACR	NA	NA	NA	1	0
FEACR	NA	NA	NA	0	0
Yard Waste	758	1,091	1,447	498	1,573
Cardboard from the BX	NA	NA	NA	16	179
Cardboard from the Commissary	NA	NA	NA	152	621

 Table 3.10-1. Composition of Recycled Materials in Tons

Source: Scott AFB 2005f

3.11.2 EXISTING CONDITIONS

3.11.2.1 Transportation and Circulation

The 126 ARW installation at Scott AFB is located approximately 20 miles east of St. Louis, Missouri, and is near the cities of O'Fallon and Belleville, Illinois. I-64, which is located less than 1 mile north of Scott AFB, provides east-west regional access to the area. Illinois Route 158/Air Mobility Drive provides north-south access to the area. Several other highways are located in the area and provide access to Scott AFB and the 126 ARW.

Access to the 126 ARW installation is provided by the Shiloh or Belleville Gates. Additionally, during drill weekends, access to the 126 ARW installation is provided by Wherry Gate. East Drive, which provides access between Golf Course Road to the north and South Drive to the south, is the primary roadway providing access to the 126 ARW facilities (Scott AFB 2004a). Several other smaller roads originate from East Drive.

3.11.2.2 Utilities

The following discussion summarizes the major utility systems.

Water

The 126 ARW receives its potable water through the same system that serves Scott AFB. Potable water is supplied to Scott AFB from the Illinois American Water Company via two transmission mains. The distribution system includes about 65 miles of piping and serves approximately 15,000 personnel and over 2,000 facilities and housing units. Existing water storage capacity is considered adequate to meet required maximum daily demand. However, the potable water system has several deficiencies in that it is old (approximately 60 years old), deteriorated, and corroded. Consideration is being given to privatizing the system (Scott AFB 2004a). Once the decision has been made, the water system will be brought up to code within five years. The decision to privatize is projected to be made in January 2006 (personal communication, Smith 2005).

Wastewater Collection and Treatment

Wastewater generated on the 126 ARW installation is managed by the same wastewater utility system that serves Scott AFB. This system provides wastewater collection, onsite treatment, and disposal of treated wastewater and sludge from Scott AFB facilities and housing areas. The Scott wastewater treatment plant (WWTP) discharges treated wastewater via four permitted outfalls (001, 002, 003, and A01) under permit IL0026859 issued by the Illinois EPA. Similar to the potable water system, the wastewater system has several deficiencies including deterioration

and overloading during heavy rainfall. Consideration is currently being given to privatizing the system and relocating the WWTP (Scott AFB 2004a). Once the decision has been made, the system will be brought up to code within five years (personal communication, Smith 2005).

Storm Drainage

Scott AFB is divided into 12 defined drainage basins. Surface runoff generated within the 126 ARW installation is associated with portions of Basins N1 and T1. Basin N1 includes the aircraft ramp area of the 126 ARW, and Basin T1 includes the 126 ARW aircraft hangars, flightline, and bulk fuel storage. Surface water runoff from Basin N1 flows east to Silver Creek via the North Ditch. Surface water runoff from Basin T1 is conveyed via underground piping to an open channel. This surface water then drains into the wetland area located in the eastern portion of Scott AFB before reaching Silver Creek (Scott AFB 2004a).

Natural Gas

Natural gas is provided to the 126 ARW and Scott AFB by Illinois Power. No deficiencies with this system have been identified (Scott AFB 2004a).

Electricity

Electricity is provided to the 126 ARW installation by the Scott AFB electrical system, which comprises 34.5 kilovolts provided by Illinois Power. Primary distribution is largely provided by overhead wooden poles and wooden cross arms. Evaluation of the Scott AFB electrical system indicates that there are several deficiencies, and projects are planned to address these deficiencies (Scott AFB 2004a). Specifically, projects that are planned for the 126 ARW are providing two sources of backup power. These two projects are projected to be completed in 2006 and 2007 (personal communication, Smith 2005).

THIS PAGE INTENTIONALLY LEFT BLANK.

4.0 ENVIRONMENTAL CONSEQUENCES

This section of the EA assesses potential environmental consequences associated with the Proposed Action, the Alternative Action, and the No Action Alternative. Potential impacts are addressed in the context of the scope of the Proposed Action as described in Section 2.0 and in consideration of the potentially affected environment, as characterized in Section 3.0.

4.1 EARTH RESOURCES

4.1.1 METHODOLOGY

Protection of unique geologic features, minimization of soil erosion, and the siting of facilities in relation to potential geologic hazards and soil limitations are considered when evaluating impacts to earth resources. Generally, impacts can be avoided or minimized if proper construction techniques, erosion control measures, and structural engineering designs are incorporated into project development.

Analysis of potential impacts to geologic resources typically includes identification and description of resources that could potentially be affected, examination of the potential effects that an action may have on the resource, assessment of the significance of potential impacts, and provision of mitigation measures in the event that potentially significant impacts are identified. Analysis of impacts to soil resources resulting from proposed activities examines the suitability of locations for proposed operations and activities. Impacts to soil resources can result from earth disturbance that would expose soil to wind or water erosion.

4.1.2 IMPACTS

4.1.2.1 Proposed Action

Under the Proposed Action, up to approximately 1.80 acres of surface would be temporarily disturbed as a result of construction and demolition of the proposed facilities and parking areas. There would be approximately 1.75 acres of new impervious surface following completion of all of the proposed construction.

Most of the proposed construction and demolition under the Proposed Action would occur on the Bethalto Silt Loam mapping unit, and the remainder of the proposed activities would occur on the Caseyville Silt Loam and the Wakefield Silt Loam. Because of their slow permeability, drainage issues would have to be resolved prior to construction. Given that the vast majority of the construction proposed would occur on previously developed land, and that artificial drainage has been in place for previous uses, continued development of these parcels should not be

problematic. Construction techniques should include drainage away from facilities to minimize impacts to foundations.

The grading of existing soil and placement of structural fill for new facilities would not substantially alter existing soil conditions at the 126 ARW installation because much of this land has been previously disturbed. There are no special qualities associated with the soils or geologic resources at these sites. Implementation of construction BMPs would minimize impacts associated with erosion. These BMPs would include, but not be limited to installation of silt fencing and sediment traps, application of water sprays to keep soil from becoming airborne, and revegetation of disturbed areas as soon as possible, as appropriate. Therefore, potential impacts to earth resources as a result of the Proposed Action would be minimal.

4.1.2.2 Alternative Action

Under the Alternative Action, there would be three additional construction projects that would occur in addition to the eight demolition/construction projects as described under the Proposed Action. Under this alternative, up to 3.48 acres of surface would be temporarily disturbed as a result of construction and demolition of the proposed facilities and parking areas. There would be approximately 3.43 acres of new impervious surface following completion of all the proposed construction.

As with the Proposed Action, most of the proposed construction and demolition under this alternative would occur on the Bethalto Silt Loam mapping unit, and the remainder of the activities would occur on the Caseyville and the Wakefield Silt Loams. Because of their slow permeability, drainage issues would have to be resolved prior to construction. Given that the vast majority of the construction proposed would occur on previously developed land, and that artificial drainage has been in place for previous uses, continued development of these parcels should not be problematic.

Potential impacts to earth resources as a result of the Alternative Action would be minimal.

4.1.2.3 No Action Alternative

Under the No Action Alternative, none of the proposed construction or demolition activities or the change in aircraft model and associated increase in number of aircraft would occur and there would be no new impacts to earth resources. Conditions would remain as described in Section 3.1.2.

4.2 WATER RESOURCES

4.2.1 Methodology

Criteria for evaluating impacts related to water resources associated with the Proposed Action and its alternatives are water availability, water quality, and adherence to applicable regulations. Impacts are measured by the potential to reduce water availability to existing users; endanger public health or safety by creating or worsening health hazards or safety conditions; or violate laws or regulations adopted to protect or manage water resources. An impact to water resources would be significant if it would: 1) reduce water availability to or interfere with the supply of existing users; 2) create or contribute to overdraft of groundwater basins or exceed safe annual yield of water supply sources; 3) adversely affect water quality or endanger public health by creating or worsening adverse health hazard conditions; 4) threaten or damage unique hydrologic characteristics; or 5) violate established laws or regulations that have been adopted to protect or manage water resources of an area. Impacts of flood hazards related to proposed actions can be significant if such actions are proposed in areas with high probabilities of flooding or in some way alter flood stages. These impacts can often be mitigated through the use of specific design features to minimize the effects of flooding.

The Illinois EPA Bureau of Water and the USACE are the regulatory agencies that govern water resources in the state of Illinois. The CWA of 1977 regulates pollutant discharges and development activities that could affect aquatic life forms or human health and safety.

4.2.2 IMPACTS

4.2.2.1 Proposed Action

With regard to water resources, the primary concerns associated with the Proposed Action include effects on water quality during construction activities, and changes to surface water drainage and groundwater recharge.

In general, increases in impervious surfaces act to increase peak discharge volumes and speed delivery of water to nearby waterways, which ultimately increases the potential for flooding as well as the transport of pollutants to surface waters. In undeveloped land, rainfall is collected and stored in vegetation, in the soil column, or in topographic depressions. Water is then utilized by plants and respired, or it moves slowly into groundwater and/or eventually to water bodies where it slowly moves through the hydrologic cycle. Removal of vegetation and/or soil compaction decreases infiltration into the soil column and thereby increases the quantity and timing of surface runoff.

Replacement of vegetation with an impervious surface, such as asphalt or concrete, eliminates any potential for infiltration and also speeds up delivery of the water to nearby drainage channels. With less storage capacity in the soil column and vegetation, urban streams rise more quickly during storm events and have higher peak discharge rates, both of which increase the potential for flooding downstream and damage to public infrastructure and private property.

The Proposed Action would involve approximately 1.75 acres of new impervious surfaces for the building footprints and pavements of the proposed facilities. Under the conditions of the Scott AFB industrial storm water permit, the 126 ARW is required to file a Notice of Intent with the Illinois EPA for construction activities. Additionally, implementation of BMPs to minimize the potential for exposed soils or other contaminants from construction activities to reach nearby surface waters in accordance with the Scott AFB SWPPP is required. Such BMPs could include the use of silt fences, covering of soil stockpiles, use of secondary containment for the temporary storage of hazardous liquids, detention/retention ponds, and establishment of buffer areas, as appropriate.

The proposed facilities and pavements would increase the amount of impervious surfaces on the installation, resulting in an increase in the amount of surface runoff and a decrease in ground water recharge at the installation. The proposed construction activities would require modifications to the installation storm drainage system (e.g., drainage ditches and basins) and an update to the SWPPP in order to properly manage storm water. Site drainage would be addressed within the updated SWPPP such that there would be no deleterious impacts to receiving waters as a result of these projects.

As described in Section 3.2, portions of the 100-year and 500-year floodplains are located within the 126 ARW installation (Figure 4.2-1). Projects associated with the Proposed Action have been sited to avoid these floodplain areas, so no impacts to or from floodplains are anticipated under the Proposed Action. The increase of impervious surfaces on the installation is not expected to affect the 100-year or 500-year predicted flood elevations of Silver Creek or Cardinal Creek due to the implementation of BMPs in accordance with the SWPPP that would be required for the new construction.

The rate of groundwater recharge of the upper aquifer located directly beneath the installation may be minimally impacted due to the increase of approximately 1.75 acres of impervious surfaces. However, given the developed nature of the site and the high percentage of impervious surfaces already existing, the change in surface runoff and groundwater recharge is expected to be negligible.



Figure 4.2-1. Floodplains in Relation to Proposed Construction Activities at the 126 ARW Installation at Scott AFB, Illinois

4.2.2.2 Alternative Action

Impacts to water resources would be similar to those described above for the Proposed Action. Under the Alternative Action, however, the amount of new impervious surface would be 3.43 acres, or 1.68 acres more than under the Proposed Action. Implementation of appropriate BMPs during construction, as well as updates to the SWPPP would minimize impacts to water quality such that no deleterious impacts would be expected. Projects associated with the Alternative Action have been sited to avoid the 100-year and 500-year floodplains located within the installation. The increase of impervious surfaces on the installation is not expected to affect the 100-year or 500-year predicted flood elevations of Silver Creek or Cardinal Creek due to the implementation of BMPs in accordance with the SWPPP that would be required for the new construction.

4.2.2.3 No Action Alternative

Under the No Action Alternative, there would be no ground disturbing activities or increases in impervious surfaces and thus no impacts to water resources at the 126 ARW installation would occur.

4.3 BIOLOGICAL RESOURCES

4.3.1 METHODOLOGY

The evaluation of impacts to biological resources is based upon (1) the importance (legal, commercial, recreational, ecological, or scientific) of the resource, (2) the rarity of a species or habitat regionally, (3) the sensitivity of the resource to proposed activities, and (4) the duration and magnitude of ecological ramifications. Impacts to biological resources are considered to be greater if priority species or habitats are adversely affected over relatively large areas and/or disturbances cause reductions in population size or distribution of a priority species.

4.3.2 IMPACTS

4.3.2.1 Proposed Action

Vegetation

Approximately 1.75 acres of maintained turf grass and otherwise undeveloped land is expected to be directly impacted and permanently lost due to building construction, parking lots, and paving. While there are no structures currently at these project sites, the land has been previously disturbed. Although a few scattered maple and oak trees would be cleared for the

construction of the pavement and grounds facility, no forested areas would be impacted by implementation of the Proposed Action.

Wildlife

The permanent loss of approximately 1.75 acres of mowed grass area and otherwise undeveloped land would have minimal impact on resident wildlife, given that this area does not currently provide quality habitat and given the high level of human activity in the area of proposed activities. Temporary, indirect impacts to wildlife caused by increased noise and activity levels during construction are expected to be insignificant given that this is an USAF installation, where high noise levels are daily occurrences.

Rare, Threatened, and Endangered Species

The implementation of the Proposed Action would have no impact on state or federally listed species because these species are not known to occur within the project area. Additionally the Proposed Action is not expected to impact any potential habitat for state or federally listed species. The USFWS has indicated their concurrence with this assessment (Appendix A).

Wetlands and Other Aquatic Habitats

Based on a review of the recently completed wetland delineation report, no direct impacts to identified wetlands or other Waters of the U.S. are anticipated as a result of implementation of any of the projects included in the Proposed Action. Based on a site visit in July 2005, it is clear that none of these projects would directly impact wetlands or waters of the U.S.

Indirect impacts to wetlands and other aquatic habitats could result from implementation of the Proposed Action if proper controls and BMPs are not implemented. However, these types of impacts are not quantifiable and are typically avoided/minimized by implementation of storm water detention facilities. In addition, these impacts would be further mitigated by the maintenance of a 50-foot vegetative buffer strip between any wetlands or other waters of the U.S. and any construction areas associated with the Proposed Action. The SWPPP would be updated prior to implementation of the Proposed Action, to minimize the potential for runoff from construction sites to affect wetlands.

4.3.2.2 Alternative Action

Under the Alternative Action, the eight construction and demolition projects associated with the Proposed Action would occur along with additional construction activities that would impact approximately 1.7 additional acres. Implementation of the Alternative Action would also result in a net increase of four aircraft and 27 personnel. The addition of four aircraft and 27 assigned

personnel would result in a slight difference in aircraft operations, noise and emissions and human population on the installation. The slight increase of aircraft and personnel would not likely cause additional substantial impacts to wildlife and plant species as compared to the Proposed Action. Therefore, potential impacts to biological resources resulting from implementation of the Alternative Action would be minimal and nearly consistent with those resulting from implementation of the Proposed Action.

4.3.2.3 No Action Alternative

Under the No Action Alternative, none of the activities described under the Proposed Action or Alternative Action would occur, and no impacts other than those associated with the current dayto-day operation of the airfield, would occur to biological resources. The urban grass communities and sparse tree area would be unaffected and the current minimal wildlife use of the area would be expected to continue. This alternative would not result in impacts to biological resources over and above those that have occurred historically due to habitat fragmentation and the construction of buildings and parking lots.

4.4 AIR QUALITY

4.4.1 METHODOLOGY

Air emissions resulting from the Proposed Action were evaluated in accordance with federal, state, and local air pollution standards and regulations. Air quality impacts from a proposed activity or action would be significant if they:

- increase ambient air pollution concentrations above any NAAQS;
- contribute to an existing violation of any NAAQS;
- interfere with or delay timely attainment of NAAQS; or
- impair visibility within any federally mandated federal Class I area.

The approach to the air quality analysis was to estimate the increase in emission levels due to the action alternatives.

According to USEPA's General Conformity Rule in 40 CFR Part 51, Subpart W, any proposed federal action that has the potential to cause violations in a NAAQS nonattainment or maintenance area must undergo a conformity analysis. A conformity analysis is not required if the action occurs within an attainment area. Since St. Clair County is in nonattainment for the new PM_{2.5} and 8-hour O₃ standards, a conformity determination must be performed if project

emissions exceed the *de minimis* thresholds of 100 TPY for the ozone precursors NO_x and VOC. The *de minimis* threshold for $PM_{2.5}$ has not yet been designated by the USEPA.

As described in Section 3.4.1, Section 169A of the CAA established the PSD regulations to protect the air quality in regions that already meet the NAAQS. Certain national parks, monuments, and wilderness areas have been designated as PSD Class I areas, where appreciable deterioration in air quality is considered significant. The nearest PSD Class I area is more than 100 miles from the region potentially affected by the action alternatives. Therefore, these potential activities would be unlikely to have a significant impact on any PSD Class I areas.

4.4.2 IMPACTS

4.4.2.1 Proposed Action

The Proposed Action would involve construction, demolition, and paving activities, including construction of new structures, additions to or demolition of existing structures, and installation of new pavement.

Emissions during the construction period were quantified to **Construction Emissions.** determine the potential impacts on regional air quality. Calculations of VOC, NO_x, CO, and PM₁₀ emissions from construction, grading, and paving activities were performed using USEPA emission factors compiled in the California Environmental Quality Air Quality Handbook (South Coast Air Quality Management District 1993), Calculations Methods for Criteria Air Pollution Emission Inventories (Jagielski and O'Brien 1994), and Air Emissions Inventory Guidance Document for Mobile Sources at Air Force Installations (O'Brien and Wade 2002). The emission factors for building construction include contributions from engine exhaust emissions (i.e., construction equipment, material handling, and workers' travel) and fugitive dust emissions (e.g., from grading activities). Demolition emissions evaluated include fugitive dust and transport of demolition debris offsite. Site preparation, grading, and trenching emissions include fugitive dust from ground disturbance, plus combustive emissions from heavy equipment during the entire construction period. Paving emissions include combustive emissions from bulldozers, rollers, and paving equipment, plus emissions from a dump truck hauling pavement materials to the site. Estimated emissions that would occur from construction, demolition, grading, paving, and painting activities under the Proposed Action are presented in Table 4.4-1. The emissions shown would occur over the duration of the construction period.

Emissions generated by construction, demolition, and paving projects are temporary in nature and would end when construction activities were completed. The emissions from fugitive dust (PM_{10}) would likely be less than those presented in Table 4.4-1 due to the implementation of control measures in accordance with standard construction practices. For instance, frequent

spraying of water on exposed soil during construction, proper soil stockpiling methods, and prompt replacement of ground cover or pavement are standard landscaping procedures that could be used to minimize the amount of dust generated during construction. Using efficient practices and avoiding long periods where engines are running at idle may reduce combustion emissions from construction equipment also. Vehicular combustion emissions from construction worker commuting may be reduced by carpooling.

	EMISSIONS (IN TONS)						
Source	СО	$\begin{array}{ c c c c c c c c c c c c c c c c c c c$					
Construction	1.8	0.6	8.2	< 0.1	0.6		
Demolition	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1		
Grading/Trenching	2.6	0.5	4.2	0.4	0.3		
New Pavement	0.2	< 0.1	0.4	< 0.1	< 0.1		
TOTAL	4.6	1.1	12.9	0.4	0.9		

Table 4.4-1. Construction Emissions – Proposed Action

In general, combustive and fugitive dust emissions would produce localized, short-term elevated air pollutant concentrations, which would not result in any long-term impacts on the air quality in St. Clair County or AQCR 70. The temporary construction-related emissions of PM_{10} and SO_x would not be expected to adversely impact the air quality or visibility in any of the PSD Class I areas in the vicinity of the Base.

Operational Emissions. Air emissions after the Proposed Action is completed would be expected to be slightly more than current operations, due to utilities such as boilers, heaters, and emergency generators being included with the new facilities. However, new utility equipment would be more efficient and have lower air pollutant emissions than older boilers and heaters at the Base. Similarly, new fuel transfer and vehicle maintenance facilities would be constructed with modern equipment designed to minimize air emissions. Nevertheless, the installation or modification of any air emission sources, such as boiler and heaters, emergency generators, fuel storage, etc., may trigger permitting requirements with the Illinois EPA. It is expected that the new operational emissions would not result in any long-term impacts on the air quality in St. Clair County or AQCR 70.

These projected annual emissions are below the *de minimis* thresholds for conformity with Illinois SIP and less than 10 percent of the regional emissions shown in Table 3.4.3. A conformity determination, therefore, is not required for this action.
4.4.2.2 Alternative Action

The Alternative Action would include the same construction, demolition, and paving activities, including construction of new structures, additions to or demolition of existing structures, and installation of new pavement, as presented in the previous section for the Proposed Action. The Alternative Action would also include a conversion from 8 KC-135E to 12 KC-135R aircraft, construction of additional facilities and pavements, including an expansion of the aircraft parking apron, plus the addition of 27 personnel to the 126 ARW.

Construction Emissions. Emissions during the construction period under the Alternative Action were calculated using similar methodology as for the Proposed Action and are presented in Table 4.4-2.

	EMISSIONS (IN TONS)					
Source	СО	VOC	NO _x	SO_x	PM ₁₀	
Construction	1.9	0.6	8.5	< 0.1	0.6	
Demolition	< 0.1	< 0.1	< 0.1	< 0.1	< 0.1	
Grading/Trenching	11.1	1.9	14.3	1.4	1.0	
New Pavement	1.5	0.3	3.1	0.2	0.2	
TOTAL	14.6	2.8	26.0	1.6	1.8	

Table 4.4-2. Construction Emissions Under the Alternative Action

In general, combustive and fugitive dust emissions would produce localized, short-term elevated air pollutant concentrations, which would not result in any long-term impacts on the air quality in St. Clair County or AQCR 70. The temporary construction-related emissions of PM_{10} and SO_x would not be expected to adversely impact the air quality or visibility in any of the PSD Class I areas in the vicinity of Scott AFB.

Operational Emissions. In addition to the projected minor changes in utility emissions described for the Proposed Action, the emissions from aircraft operations at the Base, including landings and take-offs, touch-and-goes, and low approaches, would change due to the replacement of 8 KC-135E aircraft with 12 KC-135R aircraft. The effect of the increase in aircraft assigned to the 126 ARW under the Alternative Action is roughly offset by the differences in aircraft engine characteristics between the two models, with a net effect being decreases in emissions of CO, VOC, and PM_{10} , and an increase in emissions of NO_x and SO₂. No changes in emissions from ground-based mobile sources, such as on-road or off-road vehicles, are expected to occur as a result of the Alternative Action. Air emissions from stationary sources related to aircraft maintenance, including aerospace ground equipment, engine test cells, chemical usage, degreasing, and painting are expected to remain approximately the

same as described under the baseline emissions with the replacement of 8 KC-135E aircraft with 12 KC-135R aircraft. The KC-135R model is more efficient and would require less maintenance. Table 4.4-3 presents a summary of estimated air emissions from mobile sources under the Alternative Action and changes relative to baseline emissions.

	Additional Annual Emissions (TPY)					
	СО	VOC	NO _x	SO_2	PM ₁₀	
Aircraft Operations	-167	-233	+74	+5	-6	

 Table 4.4-3. Estimated Change in Aircraft Emissions at Scott AFB, as a Result of the Alternative Action

Note: A positive (+) change indicates an increase over baseline emissions; a negative (-) change indicates a decrease compared with baseline emissions.

It is expected that these changes in emissions due to the Alternative Action would not result in any long-term impacts on the air quality of St. Clair County or AQCR 70.

Indirect Emissions. Implementation of the Alternative Action would result in the addition of 27 new employees commuting to and from the installation. The resulting increase in commuting emissions were calculated using emission factors from *Air Emissions Inventory Guidance Document for Mobile Sources at Air Force Installations* (O'Brien and Wade 2002). Average vehicle occupancy was assumed to be 1.1 passengers per vehicle, with vehicles assumed to have an average model year of 1998. Annual criteria pollutant emissions from personally-owned vehicles commuting of 25 additional vehicles, assuming an average round-trip commuting distance of 20 miles, 5 days per week, 50 weeks per year, are shown in Table 4.4-4.

Table 4.4-4. Additional Indirect Emissions from Additional POV Commutingdue to the Alternative Action

	Additional Emissions (Tons / Year)				
Source	СО	VOC	NO_x	SO_x	PM ₁₀
Commuting	2.9	0.2	0.2	< 0.1	0.2

It is expected that these minor additional emissions due to personally-owned vehicle commuting would not result in any long-term impacts on the air quality of St. Clair County or AQCR 70.

4.4.2.3 No Action Alternative

Under the No Action Alternative, no construction emissions would occur and operational emissions would be identical to current baseline emissions that were presented in Section 3.4.2.

4.5 NOISE

4.5.1 METHODOLOGY

Noise associated with aircraft operations at Scott AFB and MidAmerica Airport, other transportation-related noise, and construction activities associated with the Action Alternatives will be considered and compared with current conditions to assess impacts. Data developed during this process will also support analyses in other resource areas.

Based on numerous sociological surveys and recommendations of federal interagency councils, the most common benchmark referred to is an L_{dn} of 65 dBA. This threshold is often used to determine residential land use compatibility around airports, highways, or other transportation corridors. Two other average noise levels are also useful:

- An L_{dn} of 55 dBA was identified by the USEPA as a level "... requisite to protect the public health and welfare with an adequate margin of safety" (USEPA 1974). Noise may be heard, but there is no risk to public health or welfare.
- An L_{dn} of 75 dBA is a threshold above which effects other than annoyance may occur. It is 10 to 15 dBA below levels at which hearing damage is a known risk (OSHA 1983). However, it is also a level above which some adverse health effects cannot be categorically discounted.

Public annoyance is the most common impact associated with exposure to elevated noise levels. When subjected to L_{dn} of 65 dBA, approximately 12 percent of persons so exposed will be "highly annoyed" by the noise. At levels below 55 dBA, the percentage of annoyance is correspondingly lower (less than 3 percent). The percentage of people annoyed by noise never drops to zero (some people are always annoyed), but at levels below 55 dBA it is reduced enough to be essentially negligible.

4.5.2 IMPACTS

4.5.2.1 Proposed Action

Under the Proposed Action, neither military nor civil aircraft operations at Scott AFB and MidAmerica Airport would change from current conditions. Therefore, aviation-related noise would remain as described in Section 3.5.2, and would continue to be the dominant noise source in the region's acoustic environment.

However, under this proposal, the 126 ARW would build some new facilities, demolish one older facility, and upgrade other aspects of the installation's supporting infrastructure through

additions and alterations. There are several aspects of this proposal that have the potential to create noise impacts in the ROI.

Construction would most likely occur over an extended time-frame, and at any one time, only a small number of projects would be expected to be ongoing simultaneously. Therefore, noise associated with active construction sites would be expected to be localized, intermittent, and of relatively limited duration. A hypothetical scenario was developed to assess potential noise associated with construction activities on a construction site. Primary noise sources during such activity would be expected to be heavy vehicles and earth moving equipment. Table 4.5-1 shows sound levels associated with typical heavy construction equipment under varying modes of operation.

		Sound Level (in dBA) Under Indicated Operational Mode ¹				
Equipment	Idle PowerMoving UIdle PowerFull PowerLoad					
Forklift	63	69	91			
Backhoe	62	71	77			
Dozer	63	74	81			
Front-End Loader	60	62	68			
Dump Truck	70	71	74			

 Table 4.5-1.
 Typical Equipment Sound Levels

Note: 1. Measured at 125 feet.

Source: USAF 1998.

For the assessment of construction noise, a hypothetical "construction area" was designated that approximated the estimated area that would be involved in supporting a major project under the proposal.

The first step in the analysis was to estimate equipment usage and calculate the total acoustic energy that would be expected to be generated on the site. These data also provided information on an individual piece of equipment's relative contribution to the total amount of acoustic energy generated on the site. Next, individual equipment was spatially distributed throughout the construction zone considering "most likely" areas of operation. This yielded an equipment-weighted contribution to total site acoustic energy at different points throughout the site. With this spatial distribution, it was then possible to calculate a mean and standard deviation for the distribution along an axis running through the site.

These data were then used to normally distribute the total acoustic energy throughout the site. Finally, the normally distributed energy from multiple source points throughout the site was aggregated at a range of points at varying distances from the site edge. This allowed a determination at those points of the total acoustic energy that had emanated off-site.

Calculations based on this conservative scenario indicate an equivalent noise level over an eighthour period ($L_{eq(8)}$) of 67 dBA at a distance of 500 feet from the edge of the site. This is then normalized to an equivalent noise level over a 24-hour period ($L_{eq(24)}$) of 62 dBA. Since no construction activity would be expected to occur at night, this would be equivalent to L_{dn} 62. At a distance of 1,000 feet from the site, noise levels are $L_{eq(8)}$ 62 dBA and $L_{eq(24)}$ 58 dBA. Due to the conservative nature of the scenario, and the fact that sound attenuation only due to spherical spreading was considered, actual levels emanating off-site would be expected to be lower.

It should be noted that the areas involving construction are situated within areas already exposed to elevated noise from airfield operations. All projects are located in, or immediately proximate to, air-side locations directly supporting aircraft operations (runways, taxi-ways, parking ramps, etc.). These areas are well within the L_{dn} 65 contour created by aircraft noise. Construction noise emanating off-site would probably be noticeable in the immediate site vicinity, but would not be expected to create adverse impacts, or alter noise contours associated with aircraft operations. Furthermore, construction-related noise is intermittent and transitory, ceasing at the completion of construction. The long-term acoustic environment on Scott AFB and the MidAmerica Airport would be expected to remain relatively unchanged from current conditions as a result of the Proposed Action.

4.5.2.2 Alternative Action

Under the Alternative Action, the 126 ARW would accomplish the proposed construction projects discussed in Section 2.2. Additionally, three other construction projects would be accomplished. There are no unique aspects of these added projects that would result in major, long-term changes to the acoustic environment. Noise associated with all of the construction associated with this alternative would be as described above in Section 4.5.2.1.

In addition to the construction activities described above, the 126 ARW's current KC-135E aircraft would be replaced with KC-135R aircraft. As part of this conversion, the unit's current PAA of 8 aircraft would be increased to 12 PAA aircraft. To assess noise impacts associated with these changes, an estimated 50 percent increase in based KC-135 aircraft operations is assumed. Table 4.5-2 reflects this change. Average daily operations at Scott AFB and MidAmerica Airport would increase from approximately 125 to 143, a nearly 14 percent increase.

	ARRIVALS		DEPAR	DEPARTURES		CLOSED PATTERNS ²	
Aircraft	Day	Night	Day	Night	Day	Night	
Based KC- 135R	6.094	0.678	6.763	0.012	18.274	2.030	
Other Based Military	7.785	0.526	7.638	0.077	9.914	0	
Transient Military	5.007	0.039	5.007	0.039	0	0	
Air Carrier / Air Taxi	2.443	0	2.443	0	0	0	
General Aviation	5.524	0	5.515	0	13.662	0	
Total	26.853	1.243	27.366	0.128	41.850	2.030	

 Table 4.5-2. Average Daily Operations at Scott and MidAmerica Airport with Aircraft

 Modifications¹

Notes 1. Daily operations are based on averages of annual operations; therefore, numbers do not round.

2. Since closed patterns consist of a landing and a takeoff (two aviation operations), the 43.880 closed patterns shown equate to 87.760 aviation operations.

Source: Personal communication, Wein 2005

Aircraft noise levels at Scott AFB and MidAmerica Airport resulting from this alternative are depicted in Figure 4.5-1, and the land areas encompassed by these levels are compared with current noise levels in Table 4.5-3.

Sound Level	ACRES OF LAND ¹			
(In L _{dn})	Current	Alternative	Net Change	Percent Change
65 - 70	1,736.1	1,424.7	-311.4	-17.9 %
70 – 75	850.09	587.07	-263.02	-30.9 %
75 - 80	410.52	319.64	-90.88	-22.1 %
80 - 85	193.731	140.538	-53.193	-27.5 %
> 85	81.259	71.352	-9.907	-12.2 %
Total > 65	3,271.7	2,543.3	-728.4	-22.3 %

Table 4.5-3. Land Area Exposed to Indicated Sound Levels Under Alternative Action

Note: 1. Land areas exposed to indicated sound levels. Total area exposed to L_{dn} 65 or greater is shown as Totals. Source: Wasmer and Maunsell 2002.

As shown, overall noise exposure around Scott AFB and MidAmerica Airport is reduced as a result of conversion from the KC-135E to the KC-135R model, despite the increase in number of aircraft and relative increase in number of daily operations. Although based-military operations increase, the reduced noise levels created by the engines on the KC-135R aircraft result in an overall decrease in land areas exposed to noise levels.



Figure 4.5-1. Scott AFB and MidAmerica Airport Noise Contours

Overall, the aircraft conversion would result in positive impacts (i.e., a general reduction of noise exposure) in the area around the airfields. Nevertheless, it should be noted that these reductions probably would not be noticeable away from the airfields.

4.5.2.3 No Action Alternative

Under this alternative, no proposed construction activities would occur. Since no construction would occur, the noise associated with such activities would not result. Since no changes to aircraft operations or other transportation activities would result from this alternative, noise levels at Scott AFB and MidAmerica Airport would remain as described in Section 3.5.2.

4.6 LAND USE AND VISUAL RESOURCES

4.6.1 METHODOLOGY

Land Use

Land use impacts can result if an action displaces an existing use or affects the suitability of an area for its current, designated, or formally planned use. Analysis considers whether the resulting changes improve public safety and well being, and whether they are compatible with surrounding uses and functions. A proposed activity may be incompatible with local plans and regulations that provide for orderly development to protect the general welfare of the public, or conflict with management objectives of a federal or state agency of an affected area. Compatible land use development would need to comply with federal and state environmental laws and regulations.

Criteria used to evaluate impacts on land use include:

- potential for displacement of an existing desired use;
- potential to disrupt an existing or planned future land use;
- potential to reduce the suitability of the surrounding land for its current or planned use;
- potential for inconsistency with the installation's plans, regulations, and guidelines (including the AICUZ program) that provide for appropriate development of the land; and
- potential for incompatibility of the action with plans and management objectives for adjacent areas under control of other entities (e.g., state, local, federal).

Projects are evaluated for their potential to affect existing and planned land uses either positively (a beneficial effect), or negatively (a detracting effect).

Visual Resources. Federal and state land custodians often adopt regulations and procedures to protect visual resources within their jurisdiction. In urban areas, local agencies may enforce standards to control the appearance of development. To assess impacts to visual resources, areas that have high visual value or low tolerance for visible modification or have prescribed guidelines are identified. The degree to which an action would modify the existing surroundings is used to assess the level of impact.

4.6.2 IMPACTS

4.6.2.1 Proposed Action

Land Use. Overall, the proposed projects all contribute some benefit to the overall functioning and organization of the 126 ARW installation. Some projects are sited on open space and would convert it to mission-supporting use. Others are located in areas that are already developed with similar or compatible adjacent uses. Demolishing the old carport would allow reuse of this site for the new TACTC. Each project has been sited appropriately, in consideration of existing environmental and operational constraints and opportunities. Individual proposed projects would result in an increase in safety and/or functionality for the 126 ARW and are consistent with the 126 ARW plans and regulations that guide them.

None of the projects are located in safety zones; they are each compatible with AICUZ guidelines and noise exposure level at specific sites; and they do not violate height criteria for safe airfield operations. Noise during construction may interfere with conversations in nearby facilities, but this would be temporary and have no long term impact on land use.

Proposed projects would not interfere with existing and planned future uses on Scott AFB. Particularly, the future land use plan and planned projects in the immediate vicinity of the 126 ARW would be compatible with these proposed projects. These include reconfiguring the golf course, augmenting outdoor recreation areas, and constructing a new Network Operations facility to the north and constructing a new Administrative Headquarters facility immediately to the east of the 126 ARW parcel. Also, none of the projects would generate safety constraints on other activities or limit future siting options outside the 126 ARW installation.

No impact to off-Base locations would result from the proposed projects since they are set back a substantial distance from the Base boundary and are buffered by intervening activities, vegetation, and terrain.

Visual Resources. None of the proposed projects would be visible from off Base. Therefore, there would be no impact on the visual character of the surrounding vicinity, or on the experience of persons approaching the Base or using local roads.

The construction projects would each create change in their immediate surroundings, but they would be typical of the existing context and would not detract from the visual character of the Base. Positive appearance of the built environment is valued for its effect on morale. The 126 ARW's existing facilities are new and provide a positive visual image. These additional structures would reinforce the high-caliber appearance of their enclave. The new brick façade on the Security Force's facility would improve its appearance and conformance with the architectural style of the Base.

4.6.2.2 Alternative Action

Land Use. Under the Alternative Action, the expanded ramp, new hydrant pit, and Mid-Field Fire Station projects are consistent with airfield and aviation uses. They would improve aviation and airfield capabilities and safety. They are compatible with existing and planned uses and development on Scott AFB. In effect, they would improve operating conditions and thus benefit the designated land use.

Similarly, noise conditions would improve under this alternative. Even though aircraft numbers and operations would increase, the quieter engines of the new models would result in substantially reduced noise levels at the airfield. As reported in Table 4.5-3, a 22-percent reduction (about 730 acres) in the area exposed to levels above L_{dn} 65 dB would result. This would benefit residential areas to the northwest and southeast of the MidAmerica airport runway. Figure 4.5-1 shows the resulting noise contours relative to surrounding communities and roadways. The land is predominantly agricultural in use, with some residences located along the major roads and highways. Future residential development would be controlled by St. Clair County's existing plans and zoning. Both the AEOZ and the Sub-Area plan for the Scott-Joint Use Area provide the framework for future land use decisions around the airport complex.

Visual Resources. As described under the Proposed Action, the projects would be noticeable, but consistent with surrounding airfield equipment, structures, and pavements. Therefore, visual resources would not be affected by the Alternative Action.

4.6.2.3 No Action Alternative

Land Use. Under the No Action Alternative, there would be no change in land use on the installation; therefore no impacts, either positive or negative would result.

Visual Resources. There would be no change in the visual environment from new construction under the No Action Alternative; therefore, no impacts would result. However, if Building 5542 is not removed, it will eventually deteriorate and become dilapidated. Also, the benefit of the new brick façade on the Security Forces building would not be realized. The resulting impact on visual resources from not undertaking these actions is negative but minor.

4.7 SOCIOECONOMICS AND ENVIRONMENTAL JUSTICE

4.7.1 METHODOLOGY

In order to assess the potential socioeconomic and environmental justice impacts of the Proposed Action, employment, race, ethnicity, poverty status and age characteristics of populations in the ROI were analyzed, as presented in Section 3.7. Potential socioeconomic impacts are assessed in terms of the direct effects of the proposal on the local economy and related effects on population and socioeconomic attributes. With regard to environmental justice issues, where impacts may result from implementing the proposal, the demographics and income levels of affected populations are examined to determine whether impacts are disproportionately borne by minorities, children, or low-income persons.

4.7.2 IMPACTS

4.7.2.1 Proposed Action

Under the Proposed Action, the 126 ARW would implement construction and demolition projects as described in Section 2.2. Implementing the proposed construction and demolition projects would result in expenditures of approximately \$14 million over the entire construction period. These expenditures would be spread out over the course of the approximately five-year time period. Although short-term, expenditures would benefit the local economy. Employment associated with construction activities would benefit the local workforce, but would tend to be temporary. Implementing these projects would help maintain an ongoing schedule of construction on Scott AFB that benefits the local communities.

The Proposed Action would not create adverse environmental or health impacts. Consequently, no impacts would be experienced by minority and/or low-income populations or children. None of the projects take place where children could be directly affected. Access to construction areas would be controlled, therefore limiting unauthorized access by any person, including children. Jobs generated by proposed construction projects would benefit the entire community.

4.7.2.2 Alternative Action

Under the Alternative Action, the additional construction projects would increase expenditures by about \$3 million over the \$14 million of the Proposed Action. Impacts to the local economy would be minor, but beneficial, as described under the Proposed Action.

This alternative would also provide increases in direct jobs for supporting four additional aircraft and for 27 firefighter positions. The total increase would be a small percentage of jobs at Scott AFB, but beneficial to the local community. Many of these positions would represent opportunities for part-time jobs and supplemental income for household in the local area.

The Alternative Action is not expected to create significantly adverse environmental or health impacts. Consequently, no impacts would be experienced by minority and/or low-income populations. In addition, there are no known environmental health or safety risks associated with the Alternative Action that may disproportionately affect children. Access to construction areas would be controlled, therefore limiting unauthorized access by any person, including children.

4.7.2.3 No Action Alternative

Under the No Action Alternative, the 126 ARW would maintain their existing facilities and would not construct the facilities proposed. Failure to implement the proposed improvements would not generate any of the beneficial construction-related employment or earnings impacts associated with the Proposed Action. Implementation of the No Action Alternative would not result in any significant adverse socioeconomic or environmental justice impacts.

4.8 CULTURAL RESOURCES

4.8.1 METHODOLOGY

A number of federal regulations and guidelines have been established for the management of cultural resources. Section 106 of the NHPA, as amended, requires federal agencies to take into account the effects of their undertakings on historic properties. Historic properties are cultural resources that are listed in, or eligible for listing in, the NRHP. Eligibility evaluation is the process by which resources are assessed relative to NRHP significance criteria for scientific or historic research, for the general public, and for traditional cultural groups. Under federal law, impacts to cultural resources may be considered adverse if the resources have been determined eligible for listing in the NRHP or have significance for Native American groups. In general, architectural and engineering resources need to be at least 50 years old to be considered for inclusion on the NRHP.

Analysis of potential impacts to cultural resources considers both direct and indirect impacts. Direct impacts may occur by physically altering, damaging, or destroying all or part of a resource; altering characteristics of the surrounding environment that contribute to the resource's significance; introducing visual or audible elements that are out of character with the property or alter its setting; or neglecting the resource to the extent that it deteriorates or is destroyed. Direct impacts are assessed by identifying the types and locations of proposed activities and determining the exact location of cultural resources that could be affected. Indirect impacts result primarily from the effects of project-induced population increases.

4.8.2 IMPACTS

4.8.2.1 Proposed Action

Under the Proposed Action, the 126 ARW would implement eight separate projects that include facility demolition, additions to existing facilities, and new construction. Although the projects of the Proposed Action involve some level of ground disturbance, all are situated in areas that are heavily disturbed or have been previously surveyed for cultural resources. No cultural resources have been located where the projects associated with the Proposed Action would occur; thus, none of the projects are expected to impact historic or Native American archaeological resources.

Impacts to architectural and engineering resources could occur under the Proposed Action (Table 4.8-1). Under the Proposed Action, five buildings would be altered or receive additions, and one would be demolished. Five of the structures were built between 1999 and 2001; in addition to being less than 50 years of age, none are associated with significant historic events that would merit NRHP eligibility. However, Building 5542, a Multi-Vehicle Carport, was constructed in 1951 and has not been evaluated with regard to its NRHP eligibility; not only is it more than 50 years old, but it could also be eligible under special considerations afforded to Cold War era architectural and engineering resources. The 126 ARW has coordinated with the Illinois SHPO (Appendix A), and the SHPO has indicated that none of the facilities proposed for alteration and/or demolition qualify as historic properties and therefore, the action would not impact historic resources. In the event of inadvertent discoveries of cultural resources during any project-related activities, all activities at that location would be halted until the find is evaluated by a qualified professional archaeologist, in compliance with federal laws and USAF regulations.

Impacts to traditional resources are not expected under the Proposed Action. The ILANG has initiated contact with the nearby tribes to identify any potential concerns they may have associated with the Proposed Action (Appendix A).

Building Number, Function	Build Date	NRHP Eligibility	Project/Action	Proposed Action	Alternative Action
5542 Multi-vehicle Carport	1951	Unevaluated	Facility Demolition	Х	Х
5010 Communication Facility	2000	Not eligible	Addition/Alteration to Communication Facility	X	Х
3901 Fire Station	2001	Not eligible	Addition/Alteration to Mid-Field Fire Station	X	Х
3901 Fire Station	2001	Not eligible	Further Addition to Mid- Field Fire Station		Х
5028 Deployment Processing Center	2001	Not eligible	Addition/Alteration to Deployment Processing Center	X	Х
5038 Pump House	2001	Not eligible	Construct Pump House Spill Containment	X	Х

 Table 4.8-1. Buildings Potentially Affected Under Action Alternatives

4.8.2.2 Alternative Action

Under the Alternative Action, the eight projects of the Proposed Action would be implemented along with three additional projects. Like the original eight projects, none of the three additional projects are near known historic or Native American archaeological resources. Additionally, all are situated on previously disturbed areas or areas previously surveyed for cultural resources. As a result, impacts to historic or Native American archaeological resources are not expected under this alternative.

While the three additional projects under this alternative do not involve the addition/alteration or demolition of any architectural and engineering resources, demolition of Building 5542 would also be accomplished under this alternative (refer to Table 4.8-1). As such, impacts to architectural and engineering resources under the Alternative Action are virtually the same as for the Proposed Action. In compliance with Section 106 of the NHPA, Building 5542 would need to be evaluated for NRHP eligibility prior to the implementation of the Alternative Action. In the event of inadvertent discoveries of cultural resources during any project-related activities, all activities at that location would be halted until the find is evaluated by a qualified professional archaeologist, in compliance with federal laws and USAF regulations.

Impacts to traditional resources are not expected under the Alternative Action. The ILANG has initiated contact with the nearby tribes to identify any potential concerns they may have associated with the Alternative Action. Tribal contact letters are included in Appendix A.

4.8.2.3 No Action Alternative

Under the No Action Alternative, the 126 ARW would not carry out the proposed construction and demolition projects associated with the action alternatives. No impacts to archaeological, architectural/engineering, or traditional resources would be expected. The facilities would continue to be managed in compliance with federal laws and USAF regulations.

4.9 SAFETY

4.9.1 METHODOLOGY

Impacts related to safety are assessed according to the potential to increase or decrease risks to personnel, the public, and property. Proposal-related activities are considered to determine if additional or unique safety risks are associated with their undertaking. If any proposal-related activity indicated a major variance from existing conditions, it would be considered a safety impact.

4.9.2 IMPACTS

4.9.2.1 Proposed Action

Under the Proposed Action, there would be no change to flying operations. Therefore, flight safety risks and BASH risks would remain as described in Section 3.9.2.

Providing new facilities for the 126 ARW that support operational requirements, and are properly sited with adequate space and a modernized supporting infrastructure would generally enhance ground, explosive, and flight safety during required operations, training, maintenance and support procedures, security functions, and other activities conducted by the unit. Fire and crash response capability would be improved by the proposed addition to the Mid-Field Fire Station.

Construction of the pumphouse spill containment complex would enhance ground safety, and minimize risks associated with fire and environmental contamination. Construction of a new Pavements and Grounds facility would also correct many ground safety concerns by eliminating existing health, safety, and fire code deficiencies. Proposed modifications to the Communication Facility would also enhance ground safety and equipment security by providing adequate environmental controls ensuring the safe and efficient operation of the communication function.

Overall unit safety and security would be improved by the proposed modifications to the Security Forces Facility.

Demolition of Building 5542 would also provide positive safety impacts inasmuch as it results in removing a facility that is dilapidated and functionally inadequate.

Activities involved in the proposed facility construction, modification, and demolition are not unique. Standard building and construction procedures and BMPs would be followed by the construction contractor(s). Implementation of the Proposed Action would involve ground activities that may expose workers performing the required site preparation, grading, and building construction to some risk. The U.S. Department of Labor, Bureau of Labor Statistics maintains data analyzing fatal and non-fatal occupational injuries based on occupation. Due to the varying range of events classified as non-fatal injuries, the considerations described below focus on fatal injuries since they are the most catastrophic. Data are categorized as incidence rates per 100,000 workers employed (on an annual average) in a specific occupation.

To assess relative risk associated with this proposal, it was assumed that the industrial classifications of workers involved are the Construction Trades. Based on U.S. Department of Labor data and considerations of worker exposure, the probability of a fatal injury would be statistically predicted to be 1.17 out of 10,000 (U.S. Department of Labor 2004). Although DoD guidelines for assessing risk hazards would categorize the hazard category as "catastrophic" (because a fatality would be involved), the expected frequency of the occurrence would be considered "remote" (MIL-STD-882 1993). While the potential result must be considered undesirable, risk is low. Strict adherence to all applicable occupational safety requirements would further minimize the relatively low risk associated with these construction activities.

In general, implementation of the Proposed Action would result in positive impacts to safety.

4.9.2.2 Alternative Action

Under the Alternative Action, the 126 ARW would accomplish the proposed construction projects discussed in Section 2.2. Under this alternative, three additional construction projects would be accomplished. Expanding the aircraft apron would enhance ground safety by providing adequate maneuvering space for the increased level of assigned aircraft. The further additions to the Mid-Field Fire Station, providing infrastructure support for 27 additional fire fighters, enhances ground safety, and fire and crash response capability. Finally, the construction of a new fuel hydrant pit and blast deflector on the aircraft parking apron would support more efficient fueling on the apron, and comply with all military Type III hydrant fueling system requirements.

Safety issues associated with the construction phase of this alternative would be as described under the Proposed Action.

Also under the Alternative Action, the unit's current 8 KC-135E aircraft would be replaced with 12 KC-135R aircraft. To assess safety impacts associated with these changes, an estimated 50 percent increase in based-aircraft operations was assumed. Under current conditions, with the unit flying approximately 3,900 hours annually, calculations statistically indicate that the unit could be expected to experience a Class A mishap once every 40 years. With an increase to approximately 5,850 annual hours, this projection is reduced to once every 27 years. Nevertheless, the probability of an aircraft assigned to the 126 ARW experiencing a Class A mishap remains relatively unchanged from the conditions described in Section 3.9.2 (mishap probability is 0.000006).

Increased flying hours result in increased exposure to bird-aircraft strikes. Based on historic data, the 126 ARW has experienced less than eight bird-strikes per year. Although risk exposure would increase with an increase in flying hours, it would not be expected to result in a serious flight risk issue.

In general, implementation of the Alternative Action would result in positive impacts to ground safety, while the potential for flying mishaps would increase.

4.9.2.3 No Action Alternative

Under the No Action Alternative, no facility improvements would be accomplished. The 126 ARW would continue operations and maintenance using existing facilities. The size, configuration, and infrastructure associated with these existing facilities would continue to be inadequate in some cases. The existing adverse ground, explosive, and safety issues would continue, and could ultimately impact on the unit's ability to support its required mission.

4.10 SOLID AND HAZARDOUS MATERIALS AND WASTE

This section addresses the potential impacts caused by solid waste, and hazardous materials and waste management practices that would result from implementation of the alternatives. Hazardous materials and petroleum products, hazardous and petroleum wastes, ERP sites, and solid wastes are discussed below.

4.10.1 Methodology

The qualitative and quantitative assessment of impacts focuses on how and to what degree the alternatives would affect hazardous materials usage and management, hazardous waste generation and management, and waste disposal. The assessment considers potential for increase

in the quantity or toxicity of hazardous substances used or generated. Significant impacts could result if a substantial increase in human health risk or environmental exposure was generated at a level that cannot be mitigated to acceptable standards.

Regulatory standards and guidelines have been applied in evaluating the potential impacts that may be caused by hazardous materials and wastes. The following criteria are used to identify potential impacts:

- Generation of 100 kilograms (or more) of hazardous waste or 1 kilogram (or more) of an acutely hazardous waste in a calendar month, resulting in increased regulatory requirements.
- A spill or release of a reportable quantity of a hazardous substance as defined by the USEPA in 40 CFR Part 302.
- Manufacturing, use, or storage of a compound that requires notifying the pertinent regulatory agency according to EPCRA.
- Exposure of the environment or public to any hazardous material and/or waste through release or disposal practices.

4.10.2 IMPACTS

4.10.2.1 Proposed Action

Hazardous Materials and Petroleum Products

Hazardous materials and petroleum products associated with 126 ARW operations would continue to be managed in accordance with all federal, state, and local regulations, as well as existing 126 ARW procedures. Construction activities associated with the Proposed Action would require the temporary use of certain hazardous materials such as sealants, primers, paints, solvents and preservatives. The construction equipment proposed for this project would utilize various fuels, coolants, lubricating oils and hydraulic fluids. If spilled or leaked onto the construction site, these could be regulated as hazardous substances. During construction, contractors would be required to conduct daily equipment inspections to minimize the potential for a release of hazardous substances. In addition, contractors would be required to store all fuels and other materials in appropriate containers in designated locations. Furthermore, the maintenance or repair of construction equipment would not be conducted on Scott AFB.

Hazardous and Petroleum Wastes

Under the Proposed Action, the 126 ARW would continue to perform the same functions as it is currently assigned. Because aircraft maintenance, vehicle maintenance, and facility maintenance would remain the same as under current operations, the amount of hazardous and petroleum wastes generated would remain the same over the long term. Although some additional hazardous and petroleum wastes would be generated by construction activities, generation of these wastes would occur only for the duration of the construction activities and would be managed in compliance with all applicable regulations.

The only demolition under the Proposed Action is Building 5542, a multi-vehicle carport that was constructed in 1951. Although it is not likely that ACM and LBP would be present due to the past use of Building 5542, based on the age of this building (i.e., constructed in 1951), they could be present. If asbestos is present, a licensed asbestos abatement contractor would remove all friable asbestos materials from the buildings prior to demolition. The 126 ARW would ensure compliance with all regulations pertaining to ACM and LBP removal.

Under the Proposed Action, construction and future operation of the proposed facilities would not have a substantial impact on the use, storage, or generation of hazardous wastes at the installation. If a contractor cannot avoid the generation of hazardous waste, the contractor would be responsible for the final disposition of those materials per contract specifications and environmental laws. The improper usage or disposal of hazardous materials or wastes during construction activities could result in Notices of Violation from the Illinois EPA. The construction of the spill containment facility around the pumphouse has the potential to reduce or eliminate the future release of hazardous substances from this building.

Under the Proposed Action, training requirements and aircraft sortie levels would remain the same as current operations and the amount of hazardous and petroleum wastes generated would remain the same over the long term. The new and remodeled facilities would be constructed with berms and drains leading to OWS, if required, to contain releases of petroleum products. Hazardous materials and waste management plans would be updated, as necessary, as successive construction projects are completed.

Environmental Restoration Program

With regard to the ERP sites, the Proposed Action includes the construction of facilities within or near four AOCs (Figure 4.10-1). The construction of the Pavements and Grounds Facility, the TACTC and the demolition of Building 5542 would occur within the area designated as AOC 23. The construction of the Security Forces and Communications Facility additions would occur within the area designated as AOC 14 and the addition to Building 5028 is proposed to occur near one of the former oil/water separators (OWS 15) identified as part of AOC 6. Although the construction associated with implementation of the Proposed Action would involve the disturbance of surface soils, all of the projects would be constructed on concrete slabs and on grade with no basements or subsurface structures other than the possible extension of buried utility lines. Preliminary investigations are currently ongoing at the various AOC sites and no formal risk assessments have been conducted at this time. Once preliminary investigation of these AOCs is complete, the 126 ARW would evaluate what steps need to be taken in order to properly address potential contamination associated with these sites, including remediation or other measures to minimize risks associated with the potential to encounter hazardous wastes. However, due to the limited amount of soil disturbance associated with the Proposed Action, implementation of the Proposed Action would not be anticipated to cause significant impacts to human health and the environment.

Solid Wastes

It is anticipated that the implementation of the Proposed Action at Scott AFB would have a minimal impact to the solid waste management program at Scott AFB or to the capacity of the area landfill. Solid waste generated from implementation of the Proposed Action would consist of a minimal amount of building materials such as drywall, solid pieces of concrete, metals (conduit, piping, and wiring), and lumber.

The demolition of Building 5542 would generate approximately 500 cubic yards of construction debris. Arrangements would be made for the storage or disposal of construction and demolition debris at a licensed disposal facility. All solid waste would be disposed of in accordance with applicable federal, state, local, and USAF regulations. With adequate available landfill capacity in the surrounding area, these quantities would not cause adverse impacts to the capacity of the area landfill.



Figure 4.10-1. ERP Sites and AOCs in Relation to Proposed Construction Activities on the 126 ARW Installation at Scott AFB, Illinois

4.10.2.2 Alternative Action

Under the Alternative Action, the eight construction and demolition projects associated with the Proposed Action would occur along with additional construction activities and a net increase in aircraft and personnel. The addition of four aircraft and 27 assigned personnel would result in an increase in aircraft operations; however, an increase in the subsequent use and generation of hazardous materials, substances and waste is not expected due to the maintenance schedule of the newer airframe. Under the Alternative Action, aircraft sortie levels would increase approximately 50 percent; however, the amount of hazardous and petroleum wastes generated would likely remain as described under the Proposed Action over the long-term, as the new model requires less maintenance.

The potential impacts to human health and the environment resulting from construction activities in and around ERP or AOC sites would not be substantially different than those associated with the Proposed Action (Figure 4.10-1). Solid waste associated impacts resulting from implementation of the Alternative Action would increase slightly due to the additional construction but generally remain as described under the Proposed Action. Therefore, potential impacts from solid and hazardous materials and wastes as a result of the Alternative Action would be minimal.

4.10.2.3 No Action Alternative

Under this alternative, there would be no change to the current operations of the installation. Therefore, conditions within the ROI would continue as described in Section 3.10.2 for solid and hazardous materials and waste management.

4.11 INFRASTRUCTURE

4.11.1 Methodology

Impacts to transportation and utilities are assessed with respect to the potential for disruption to or improvement of current circulation patterns and utility systems, deterioration or improvement of existing levels of service on roadways, changes in existing levels of transportation, changes in demand or utility systems, and utility safety. Impacts may arise from physical changes to circulation or utility corridors, construction activity and introduction of construction-related traffic and utility use. Adverse impacts on roadway capacities would be significant if roads with no history of capacity exceedance were forced to operate at or above their full design capacity.

4.11.2 IMPACTS

4.11.2.1 Proposed Action

Under the Proposed Action, the 126 ARW would implement construction and demolition projects described in detail in Section 2.2. Implementation of the Proposed Action would gradually remove and replace aging facilities, improve some aspects of utilities systems, and improve some aspects of the installation circulation system.

The Proposed Action would result in an increase in vehicle trips to and from the 126 ARW installation for construction workers and the delivery of materials and equipment, as well as removal of demolition debris. Increases in traffic volumes on roads surrounding the installation would be temporary, and because the elements of the Proposed Action would be phased over several years, the impacts to the local transportation system would be minor. Because the Proposed Action does not include any increases in personnel or operations of the 126 ARW, no permanent increases in vehicle trips would occur.

Currently, fuel accidentally released from the pumphouse would eventually flow into the storm drainage system, presenting substantial environmental concerns. Under the Proposed Action, the 126 ARW would construct pumphouse spill containment by installing new piping and drainage inlets, installing above ground fuel piping, encompassing all potential exists, and raising manholes within the existing containment area. As a result, the Proposed Action would reduce the potential for fuel to be released into the storm drainage system and would decrease potential health, safety, and environmental risks.

Construction activities associated with the Proposed Action could result in some temporary interruption of utility services and minor hindrance of transportation and circulation. These impacts would be temporary, occurring only briefly during the construction period, and would occur in phases.

4.11.2.2 Alternative Action

Under the Alternative Action, the 126 ARW would implement construction and demolition projects described in detail in Section 2.3. Implementation of the Alternative Action would gradually remove and replace aging facilities, improve utilities systems, and improve some aspects of the installation circulation system.

The Alternative Action would result in roughly the same minor impacts as the Proposed Action; however, due to a minor increase in the number of construction projects, the increase in vehicle trips to and from the 126 ARW installation for construction workers and the delivery of materials and equipment, as well as removal of demolition debris, would be somewhat greater. The

Alternative Action would also result in an increase of personnel and operations, therefore a permanent increase in vehicle trips would occur. However, it is not anticipated that the permanent increase in vehicle trips generated by 27 additional personnel would have a substantial effect on local traffic.

Construction activities associated with the Alternative Action could result in a somewhat larger number of temporary interruptions of utility services and minor hindrances of transportation and circulation due to the three additional construction projects. These impacts would be temporary, occurring only briefly during the construction period, and would occur in phases.

4.11.2.3 No Action Alternative

Under the No Action Alternative, the 126 ARW would maintain their existing facilities and would not undertake the construction and demolition projects described under the Proposed Action. Continued use and maintenance of the existing degraded and inefficient facilities and infrastructure would require that the 126 ARW continue to operate under unnecessarily inefficient conditions.

5.0 CUMULATIVE IMPACTS AND IRREVERSIBLE AND IRRETRIEVABLE COMMITMENT OF RESOURCES

5.1 CUMULATIVE IMPACTS

Cumulative impacts to environmental resources result from incremental effects of proposed actions when combined with other past, present, and reasonably foreseeable future projects in the ROI. Cumulative impacts can result from individually minor, but collectively substantial, actions undertaken over a period of time by various agencies (federal, state, and local) or individuals. In accordance with NEPA, a discussion of cumulative impacts resulting from projects that are proposed (or anticipated over the foreseeable future) is required.

The 126 ARW and the Scott AFB host, the 375 AW, both update facilities on a continual basis, as necessary. While it is not practical to catalog all minor projects that could occur over the short-term, the major projects in the ROI have been analyzed for the potential to create cumulative environmental impacts. Planning efforts in the ROI include the actions described within this EA, as well as those other projects that are ongoing, or planned over the short-term. Additional projects within the ROI are discussed below.

5.1.1 CURRENT AND REASONABLY FORESEEABLE ACTIONS IN THE REGION

Currently ongoing and other actions proposed over the next five years (in addition to those that are a component of this EA) at and surrounding Scott AFB are shown in Table 5.1-1.

As an active military installation, Scott AFB and its tenants undergo changes in mission and training requirements in response to defense policies, current threats, and tactical and technological advances, and as such, require new construction, facility improvements, infrastructure upgrades, and ongoing maintenance and repairs on a continual basis. Although such known construction and upgrades are a part of the analysis contained in this section, some future requirements cannot be predicted. As those requirements surface, future NEPA analysis will be conducted, as necessary.

Project Name/Description	Approximate Square Footage	Anticipated Fiscal Year for Implementation
On-going Projects		
Construct Explosive Ordnance Disposal Facility	6,405	On-going
Demolition of Various Buildings (800, 853, 854, 855, 878, 3164, 4141, 4157)	58,978	On-going
Proposed Projects over the Next Five Years		
Construct Security Forces Warehouse	6,000	2006-07
Construct C-40 Squadron Operations Facility	21,000	2006-07
Construct a Security Forces Complex	33,906	2007
Construct an Administrative Facility for Headquarters AMC and Headquarters USTC, Phase I and II	210,000	2007-08
Construct a Dormitory for 144 Enlisted Personnel	51,150	2007-08
Construct a Child Development Center	22,604	2008-09
Construct Fitness Center	39,600	2008-09
Construct Aeromedical Evacuation Squadron Warehouse	7,500	2009-10
Construct Defense Information Systems Administration Facility	100,000	2010-11
Construct Distribution and Processing Center	31,000	2010-11
Construct Base Exchange Parking Lot	87,120	2006-07
Construct Civil Engineering Complex	84,365	2010
Construct Golf Course Club House; Realign Course	20,000	2010
Addition/Alteration to C-40 Squadron Operations Facility	16,086	2011
Construct 375 th Communication Squadron Document Staging Warehouse	7,500	2007-08
Addition/Alteration to Vehicle Maintenance Facility	4,000	2008-09
Total Square Footage	807,214 SF (18.5 acres)	

Table 5.1-1. Ongoing and Proposed Projects in the ROI

The goal of this EA is to document the known projects required at the 126 ARW installation at Scott AFB over the next five years in support of their mission; provide an environmental analysis of these projects; and prepare to implement the appropriate facility improvements as funds become available. It is quite likely that during the course of the next five years, additional projects not included in this analysis may be required. The nature of the military today is that missions are dynamic and planners at the installation level must be proactive in addressing potential impacts associated with these changes. One of the primary purposes of preparing this EA is to streamline the NEPA process, where appropriate, by preparing a comprehensive document (herein) that will support future tiering of environmental analyses and application of categorical exclusions, as appropriate. Should additional projects be required, a checklist has been provided that should facilitate tiering and/or application of categorical exclusions. If the Environmental Manager can ensure that the following conditions are met, then it would likely be appropriate to use the existing EA for tiering or application of a categorical exclusion, depending on the particular situation:

- Wetlands and/or waters of the U.S. would not be impacted.
- Federally and/or state listed species of concern, and/or migratory birds would not be impacted.
- ERP sites would not be impacted.
- Historic properties, sites, American Indian traditional resources would not be impacted.
- No unapproved facilities would be located within quantity-distance arcs.
- NPDES permit would be updated, as necessary.
- Federal and/or state AAQS would not be exceeded.
- There would be no adverse impacts to disadvantaged and/or youth populations.

5.1.2 ANALYSIS OF CUMULATIVE IMPACTS

Earth Resources. In addition to the 1.80 acres of surface disturbance over the course of the construction program associated with the Proposed Action (and 3.43 acres under the Alternative Action) up to an additional 18.5 acres of surface disturbance could result over the next five years from other future construction associated with the Scott AFB projects described in Table 5.1-1. The grading of existing soil and placement of structural fill for new facilities would not substantially alter existing soil conditions at the installation because, to a large extent, the

construction described above is planned for areas where surface disturbance has previously occurred. BMPs would be used to limit soil movement, stabilize runoff, and control sedimentation. Cumulative impacts to earth resources are expected to be minimal.

Water Resources. In addition to a net increase of approximately 1.75 acres of impervious surface that would result under the Proposed Action (and 3.43 acres under the Alternative Action), up to an additional 18.5 acres of impervious surface would be added as a result of the projects described in Section 5.1.1. To a large extent, the construction described above is planned for areas that already contain a large amount of impervious surface, and therefore much of the proposed construction would occur on already impervious surfaces. The 126 ARW and Scott AFB would update their SWPPP to include these projects and would obtain, as appropriate, coverage under an NPDES Construction Storm Water permit from the Illinois EPA. Adherence to the requirements of the permit would include implementation of BMPs to minimize the potential for exposed soils or other contaminants from construction activities to reach nearby surface waters. It is expected that cumulative impacts to water resources would be minimal.

Biological Resources. In general, the Proposed Action, the Alternative Action, and the projects listed in Section 5.1.1 are at sites that are highly altered by man. The potential for any federally listed species to occur on the Base is low, with the exception of the Indiana Bat. Recent correspondence with the IDNR revealed that no records exist of any state-listed threatened or endangered species on or adjacent to the Base. Project proponents would coordinate, as necessary, with the USFWS prior to implementation of construction activities to ensure that impacts to sensitive species do not occur. Cumulative impacts to biological resources are expected to be minimal.

Air Quality. In general, combustive and fugitive dust emissions from proposed construction activities under either the Proposed Action or the Alternative Action, as well as those activities described in Section 5.1.1, would produce localized, elevated air pollutant concentrations that would occur for a short duration and would not result in any long-term impacts on the air quality of St. Clair County or AQCR 70. Cumulative impacts to air quality in the County and the AQCR are expected to be minimal.

Noise. Construction noise emanating off-site as a result of the Proposed Action and the activities described in Section 5.1.1 would probably be noticeable in the immediate construction site vicinity, but would not be expected to create adverse impacts. In addition, noise associated with the aircraft conversion and the associated construction activities described under the Alternative Action in concert with those actions described in Section 5.1.1 would similarly not be expected to create adverse impacts because the noise contours actually shrink under this alternative. The acoustic environment on and near the airfield property is expected to remain relatively

unchanged from existing conditions under all described activities. Cumulative impacts from noise would be expected to be minimal.

Land Use/Visual Resources. The proposed construction and demolition projects described under both the Proposed Action and the Alternative Action, as well as those described in Section 5.1.1, are expected to enhance overall installation planning and compatibility of functions on the 126 ARW installation and at Scott AFB in general. Some existing incompatibilities would be corrected. Visual resources are generally not expected to be impacted. Cumulative impacts to land use and visual resources are expected to be minimal.

Socioeconomics/Environmental Justice. There are no substantial long-term changes in population and/or employment as a result of implementation of the Proposed Action or the projects described in Section 5.1.1. Although there is an anticipated increase of 27 personnel under the Alternative Action, this is a minor increase and should not result in any appreciable impact. Additionally, the projects described under both action alternatives are not expected to create adverse environmental or health effects and therefore no disproportionately high or adverse impacts to minority, low-income, or youth populations are expected. Cumulative impacts to socioeconomics and environmental justice are expected to be minimal.

Cultural Resources. Activities associated with either the Proposed Action, or the Alternative Action, and the projects described in Section 5.1.1 are not expected to impact archaeological or traditional resources. Demolition of all proposed facilities, including Building 5542, would be coordinated with the Illinois SHPO to determine its eligibility to the NRHP prior to demolition. Cumulative impacts to cultural resources are expected to be minimal.

Safety. Implementation of the Proposed Action, or the Alternative Action, and the activities described in Section 5.1.1 do involve ground activities that may expose workers performing the required site preparation, grading, and building construction to some risk. Strict adherence to all applicable occupational safety requirements would minimize the relatively low risk associated with these construction activities. All projects have been sited outside any quantity-distance arcs, as appropriate. Additionally, the proposed projects would include measures to enhance and correct AT/FP shortfalls as part of the facility designs. Cumulative impacts to safety are expected to be minimal.

Hazardous Materials and Waste Management. The proposed construction and demolition projects associated with the Proposed Action or the Alternative Action, as well as those described in Section 5.1.1, would generate construction and demolition waste that would be recycled and/or taken to a local demolition landfill, as appropriate. There are no capacity issues with the existing landfills. Hazardous materials and wastes would be handled, stored, and disposed of in accordance with applicable regulations. Any ACM, LBP, or contaminated soils

associated with ERP sites would be removed and disposed of per applicable regulations. Cumulative impacts as a result of hazardous materials and waste management are expected to be minimal.

Infrastructure. The proposed construction and demolition projects associated with the Proposed Action or the Alternative Action, as well as those described in Section 5.1.1, would result in some temporary interruption of utility services and minor hindrance of transportation and circulation during construction activities. These impacts would be temporary, occurring only for the duration of the construction period. In general, infrastructure at the 126 ARW installation and Scott AFB would improve under these actions. Cumulative impacts to infrastructure are expected to be minimal.

5.2 IRREVERSIBLE AND IRRETRIEVABLE COMMITMENT OF RESOURCES

NEPA CEQ regulations require environmental analyses to identify "...any irreversible and irretrievable commitments of resources that would be involved in the Proposed Action should it be implemented" (40 CFR Section 1502.16). Irreversible and irretrievable resource commitments are related to the use of nonrenewable resources and the effects the uses of these resources have on future generations. Irreversible effects primarily result from the use or destruction of a specific resource (e.g., energy and minerals) that cannot be replaced within a reasonable time frame. Building construction material such as gravel and gasoline usage for construction equipment would constitute the consumption of non-renewable resources.

Neither the Proposed Action nor the Alternative Action would have irreversible impacts because future options for using these project locations would remain possible. The sites could be used for alternative uses in the future, ranging from natural open space to urban development. No loss of future options would occur as a result of either of these alternatives.

The primary irretrievable impacts of the Proposed Action or the Alternative Action would involve the use of energy, labor, materials and funds, and the conversion of some lands from an undeveloped condition through the construction of buildings and facilities. Irretrievable impacts would occur as a result of construction, facility operation, and maintenance activities. Direct losses of biological productivity and the use of natural resources from these impacts would be inconsequential.

6.0 SPECIAL OPERATING PROCEDURES AND MITIGATIONS

6.1 SPECIAL OPERATING PROCEDURES

Impact evaluations contained in this EA have determined that no significant environmental impacts would result from implementation of the Proposed Action or Alternative Action. However, this determination is based on the following procedures being completed by knowledgeable, responsible personnel from the 126 ARW, working through the appropriate federal, state, and local agencies.

- Construction BMPs would be employed during construction activities to minimize soil movement, stabilize runoff, and generally control sedimentation. These BMPs would include, but not be limited to: the development of a project specific SWPPP; regular and documented site inspections; the use of well maintained silt fences; minimizing surficial area disturbed at any given moment; stabilization of cut/fill slopes; minimization of earth-moving activities during wet weather; use of temporary detention ponds; and rapid revegetation.
- The 126 ARW would obtain any required permits, approvals, or certifications prior to implementing construction or demolition activities.
- The 126 ARW would obtain an NPDES Construction Storm Water Permit from the Illinois EPA Bureau of Water. The Scott AFB SWPPP would also be updated at this time.
- Personnel conducting construction and/or demolition activities would strictly adhere to all applicable occupational safety requirements during construction activities.
- Coordination with SHPO and interested tribes would occur prior to construction and/or demolition activities, as appropriate. Specifically, the 126 ARW would coordinate with Illinois SHPO regarding the demolition of Building 5542.
- If necessary, sampling for asbestos and LBP would occur prior to demolition of Building 5542, and materials would be handled in accordance with USAF policy.
- Contaminated soils associated with ERP sites would be removed and disposed of per applicable regulations, as determined necessary.

6.2 MITIGATION

Special operating procedures differ from mitigation in that the former are designed to prevent negative impacts during the implementation of an action while the latter remediate impacts that occur as a result of the action. Since there are no impacts anticipated as a result of implementation of the Proposed Action, no mitigation measures are required.

7.0 PERSONS AND AGENCIES CONTACTED

Berridge, SMSgt Ed. 126 ARW, Safety NCOIC, ILANG. 2005.

Cantwell, Tim. MidAmerica Airport Director. 2005.

Cheeks, Major Barbara. 126 ARW/MSF, ILANG. 2005.

- East-West Gateway Council of Governments. 2005.
- Federal Emergency Management Agency. 2005.
- Fogarty, Major John. 126 ARW Comptroller, ILANG. 2005.
- Hanks, Matt. 375 WG Geobase. 2005.
- Illinois Department of Natural Resources. 2005.
- Illinois State Geological Survey. 2005.
- Jacobs, Major Kevin. 126 ARW Environmental Manager, ILANG. 2005.
- Jacobson, LTC Jeff. 126 ARW, ILANG. 2005.
- Kaleel, Rob. Illinois Environmental Protection Agency, Air Quality Planning Division. 2005.
- Maceda, Capt. Jennifer. 375 AW/JA. 2005.
- McCoy, Mark. 375 CES/CEVR. 2005.
- Mitchell, SMSgt Jeff. 126 ARW/MXS, ILANG. 2005.
- National Oceanic and Atmospheric Administration. 2005.
- National Park Service. 2005.
- National Weather Service. 2005.
- Natural Resources Conservation Service. 2005.
- Newman, Kofi. 375 CES/CECP, Scott AFB, Illinois. 2005.

Nolan, Cindy. 375 CES/CEVR. 2005.

Pulse, Lt Col Paul. 126 ARW Chief of Safety. 2005.

Rodriguez, Andy. 375 CES/CEVR, Branch Chief. 2005.

Smith, William B. 375 CES/CEOI, Chief of Infrastructure. 2005.

Steed, LTC John. 126 ARW Base Civil Engineer, ILANG. 2005.

Tutterow, Brian. 375 CES/CEV. 2005.

United States Bureau of Economic Analysis. 2005.

United States Census Bureau. 2005.

United States Department of Labor. 2005.

United States Environmental Protection Agency. 2005.

United States Fish and Wildlife Service. 2005.

United States Geological Survey. 2005.

Wein, Lt John. 126 ARW. 2005.

8.0 LIST OF PREPARERS

Kate Bartz, Project Manager, SAIC
 M.S., Landscape Architecture & Environmental Planning, 1994
 B.S., Environmental Studies, 1987
 Years of Experience: 19

Tyrone Corn, Staff Archaeologist, SAIC B.S., Anthropology, 1997 Years of Experience: 9

- Tom Daues, Senior Biologist, SAIC M.S., Natural Resources, 1991 B.S., Biology, 1986 Years of Experience: 14
- Kimberly Wilson, Document Production Manager, SAIC Years of Experience: 18
- Susan Goodan, Senior Environmental Planner, SAIC M. Architecture, 1988 B.A., Ethics/Archaeology, 1975 Years of Experience: 17
- Heather Gordon, Environmental Analyst (GIS), SAIC B.A., Environmental Studies and Planning, 1996 Years of Experience: 7
- Lorraine Gross, RPA, Senior Archaeologist, SAIC M.S., Anthropology, 1986 B.A., Anthropology, 1975 Years of Experience: 25
- Carlos Jallo, Environmental Planner, SAIC B.A., Environment, Economics, Politics, 1994 Years of Experience: 10
- Jenny Lange, Environmental Analyst, SAIC B.S., Technical and Scientific Communications, 2002 Years of Experience: 3
- David Lingner, Senior Scientist, SAIC Ph.D., Chemistry, 1985 B.S., Chemistry and Mathematics, 1978 Years of Experience: 23

William Wuest, Senior Environmental Scientist, SAIC M.P.A., Public Administration, 1974
B.S., Political Science, 1963
Years of Experience: 39
9.0 **REFERENCES**

- 375th Airlift Wing (375 AW). 2004a. Final Environmental Assessment for the Construction of a New Security Forces Operations Facility, St. Clair County, Scott Air Force Base, Illinois. July 14th.
 - . 2004b. Scott AFB/MidAmerica Airport Joint Use Plan 91-202, Bird Aircraft Strike Hazard (BASH) Plan. 375 AW/SE, Department of the Air Force, Headquarters 375th Airlift Wing, Scott Air Force Base, Illinois. 1 March 2004.

_____. 2005. Scott AFB Final Floodplain Survey. Prepared by CH2MHill. Philadelphia, Pennsylvania. May.

- 375th Civil Engineering Squadron, Environmental Flight (375 CES/CEV). 2003. Integrated Natural Resources Management Plan for Scott Air Force Base. Prepared by 375th CES/CEV, Scott AFB, Illinois. June.
- Air Force Institute for Environment, Safety, and Occupational Health Risk Analysis (AFIERA). 2002. Hazardous Waste Management Plan, Scott Air Force Base, Illinois.
- American National Standards Institute. 1980. Sound Level Descriptors for Determination of Compatible Land Use. ANSI S3.23-1980.
- _____. 1988. Quantities and Procedures for Description and Measurement of Environmental Sound, Part 1. ANSI S12.9-1988.
- Bureau of Economic Analysis. 2003. Bearfacts 1993-2003. Illinois 2003 data. St. Clair, Illinois 2003 data. Downloaded on 10/11/2005. http://www.bea.gov/bea/regional/bearfacts.
- East-West Gateway Council of Governments. 2005. Air Quality History. Website last updated on 15 April 2005, http://www.ewgateway.org/environment/aq/AQHistory/aqhistory.htm.
- Fagan, B.M. 1991. *Ancient North America, the Archaeology of a Continent.* Thames and Hudson, London.
- Federal Emergency Management Agency (FEMA). 2003. National Flood Insurance Rate Map, St. Clair County, Illinois and Incorporated Areas (Panel 240 of 555, Map Number 17163C0240D). November.

Federal Interagency Committee on Noise (FICON). 1992. Federal Agency Review of Selected Airport Noise Analysis Issues.

_____. 2004. FEMA Regulatory Floodway Standards (Draft). May.

- Federal Interagency Committee on Urban Noise (FICUN). 1980. Guidelines for Considering Noise in Land Use Planning and Control. Washington, D.C. NIIS PB83-184838.
- Finegold, L.S., C.S. Harris, H.E. vonGierke. 1994. Community Annoyance and Sleep Disturbance: Updated Criteria For Assessing the Impacts of General Transportation Noise on People. Noise Control Engineering Journal, Jan-Feb.
- Global Security. 2005a. Scott AFB, Illinois. Accessed 9-29-2005. http://www.globalsecurity.org/military/facility/scott.htm
- . 2005b. 126th Air Refueling Wing (126th ARW). Accessed 9-29-05. http://www.globalsecurity.org/military/agency/usaf/126arw.htm
- Harrison, R.T. 1973. Forest Background Sound. Report to Record, ED&T 2428, USDA Forest Service, Technology and Development Center, San Dimas, California. In: Harrison, R.T., L.A. Hartmann, and W.J. Makel. 1990. Annoyance from Aircraft Overflights in Wilderness. NOISE-CON 90, University of Texas. Austin, Texas. October.
- Hoffmiester, D. 2002. Mammals of Illinois. University of Illinois, 334 pp.
- Illinois Administrative Code (IAC). 1992. Title 35: Environmental Protection; Subtitle B: Air Pollution; Chapter I: Pollution Control Board; Subchapter 1: Air Quality Standards and Episodes; Part 243: Air Quality Standards. Illinois Administrative Code, last amended on 15 May 1992.
- Illinois State Geological Survey. 2005. Pleistocene Glaciation in Illinois. Website accessed 9/29/05. http://www.isgs.uiuc.edu/tours/outliers/pleist-g.htm
- Illinois Department of Natural Resources (IDNR). 2004. Illinois Endangered Species Protection Board "2004 Endangered and Threatened Species List." http://dnr.state.il.us/espb/datelist.htm.
- Jagielski, K. and J. O'Brien. 1994. Calculations Methods for Criteria Air Pollution Emission Inventories. USAF, Armstrong Laboratory, AL/OE-TR-1994-0049. Brooks Air Force Base, Texas.
- Martin, C.O., R.A. Fischer, J.E. Hoffman, S.R. Hill. 2002. Environmental Assessment of Selected Fauna and Their Habitats. February.
- Military Standard System Safety Program Requirements (MIL-STD-882). 1993. Department of Defense, Washington, D.C. January 1993.

- Moulton, Carey L. 1990. Air Force Procedure for Predicting Aircraft Noise Around Airbases: Noise Exposure Model (NOISEMAP) User's Manual. Harry G. Armstrong Aerospace Medical Research Laboratory, Human Systems Division, Air Force Systems Command, Wright-Patterson AFB, Ohio. AAMRL-TR-90-011. February.
- National Oceanic and Atmospheric Administration. 1998. Climatic Wind Data for the United States. National Climatic Data Center, National Oceanic and Atmospheric Administration, Asheville, North Carolina, November 1998.
- National Park Service. 2005. Mingo National Wildlife Refuge Air Quality Information. Website last updated on 19 May 2005, http://www2.nature.nps.gov/air/Permits/ARIS/ming/
- National Weather Service. 2005. The Climatology of St. Louis and the Bi-State Area. National Weather Service, St. Charles, Missouri, website downloaded on 18 July 2005, http://www.crh.noaa.gov/lsx/climate/cli-sum1.htm.
- Natural Resources Conservation Service (NRCS). 2002. Soil Survey for St. Clair County Illinois.
 - _____. 2004. Lower Kaskaskia Watershed. Online information obtained at http://www.il.nrcs.usda.gov/programs/csp/csp_lwrkask.html on October 16, 2005. December.
- O'Brien, R.J. and M.D. Wade. 2002. Air Emissions Inventory Guidance Document for Mobile Sources at Air Force Installations. Air Force Institute for Environment, Safety, and Occupational Health Risk Analysis. IERA-RS-BR-SR-2001-0010. Brooks Air Force Base, Texas. January.
- Occupational Safety and Health Administration. 1983. Occupational Noise Exposure Standard. Code of Federal Regulations Title 29, Part 1910, Sec. 1910.95 (29 CFR 1910.95).
- OMEGA108. NOISEFILE Data Base. Harry G. Armstrong Aerospace Medical Research Laboratory (AAMRL), Wright Patterson Air Force Base, Ohio.
- Parsons. 1995. Final Multi-Site Preliminary Assessment/Site Investigation.
- Scott Air Force Base (AFB). n.d. History of Environmental Restoration Effects at the Installation. Internal Document.

_____. 1996. Lead Based Paint Management Plan.

_____. 1997a. Preliminary Assessment/Site Inspection Report Cardinal Creek Housing Area Scott AFB, Illinois. Prepared by Philip Environmental. July 10, 1997

- _____. 1997b. Soil Management Plan. Cardinal Creek Village Housing Area-South Scott AFB, Illinois. Prepared by CH2MHill. October 1997.
- _____. 1998. Final Decision Document. Interim Action for Cardinal Creek Housing Area-South. Scott AFB, Illinois. January 1998.
- _____. 2000a. Scott AFB Asbestos Management Plan.
- _____. 2000b. Asbestos Operations Plan.
- . 2001a. Spill Prevention, Control and Countermeasure Plan for Scott Air Force Base, Illinois. Prepared by Pacific Environmental Services. September 2001.
- _____. 2001b. Final Solid Waste Management Plan for Scott AFB.
- ______. 2003a. Integrated Cultural Resources Management Plan (ICRMP). Prepared for the Air Force Institute for Environment, Safety, and Occupational Health Risk Analysis, Brooks AFB, San Antonio, Texas, Prepared by Parsons.
- . 2003b. Draft Remedial Investigation Report for Site SS-12(Former Golf Course Maintenance/Entomology Shop). Prepared by MWH Americas, Inc. August 2003.
- _____. 2004a. Scott Air Force Base General Plan. Scott AFB, Illinois. October.
- _____. 2004b. Final Storm Water Pollution Prevention Plan. Scott Air Force Base, Illinois. June.
- _____. 2004c. Draft Environmental Assessment for the Joint Use Agreement, St. Clair County, Scott AFB, Illinois. December.
- _____. 2005a. Final Threatened and Endangered Species Habitat Survey. July.
- _____. 2005b. Final Wetland Delineation Report. June.
- _____. 2005c. 2004 Air Emissions Inventory. Prepared by CH2M Hill, St. Louis, Missouri. May.
 - . 2005d. Fact Sheet Scott Air Force Base Statistics. Downloaded from internet 07/12/2005. Updated on 02/08/2005. http://public.scott.amc.af.mil/375FactSheets.cfm
 - ____. 2005e. Fact Sheet, United States Air Force. Scott Air Force Base history. 375th Airlift Wing Public Affairs Office. Accessed 9-29-2005. http://public.scott.amc.af.mil/FactSheet/SAFB%20History.pdf
 - _____. 2005f. Final Work Plan Preliminary Assessment/Site Inspection At AOC 06, AOC 08, AOC 11, AOC 20, and AOC 22

- _____. 2005g. Final Preliminary Assessment/Site Inspection for AOC 18 (Coal Storage Piles Basewide) AOC 19 (Lead Sites Basewide). Prepared by SAIC. February 2005.
- _____. 2005h. Recycling Option Plan and Feasibility Study for Scott AFB. Prepared by CH2M Hill. May 2005.

South Coast Air Quality Management District. 1993. CEQA Air Quality Handbook.

Unified Facilities Criteria. 2003. UFC 4-010-01, DoD Minimum Antiterrorism Standards for Buildings. 8 October.

United States Air Force (USAF). n.d. Installation Force Protection Guide.

_____. 1991. Final EIS for Joint Military-Civilian Use of Scott AFB, IL Volume 1: Impacts Analysis. July.

_____. 1993. Air Force Policy and Guidance on Lead Based Paint in Facilities. June 1993.

- _____. 1998. 307 Red Horse EA Noise Survey. Kelly Air Force Base, Texas. 27 October 1998.
- _____. 2001. Air Installation Compatible Use Zone Study. Scott Air Force Base, Illinois. Feb.

. 2004. Air Force Instruction (AFI) 91-204. Safety. Safety Investigations and Reports. 12 April.

- United States Air Force (USAF) Bird-Aircraft Strike Hazard (BASH) Team. 2005. Selected Wildlife Strike Statistics. http://afsafety.af.mil/AFSC/Bash/stats
- United States Air Force Safety Center. 2005. C-135 Flight Mishap Statistics. Downloaded April 24, 2005. http://safety.kirtland.af.mil/AFSC/RDBMS/Flight/stats
- United States Army Corps of Engineers (USACE). 2002. U.S. Army Engineer Research and Development Center. Environmental Laboratory. Environmental Assessment of Selected Fauna and their Habitats at Scott AFB Illinois: Bat Surveys, Seasonal Avian Inventories, and Botanical Survey of Forested Areas. Vicksburg, Mississippi. February.
- United States Census Bureau (USCB). 2000a. Demographic Profile Highlights from Census 2000. United States, Illinois, St Clair County. American FactFinder. Population Finder. Downloaded on 10/11/2005. http://factfinder.census.gov/home/saff/main.html?_lang=en
 - _____. 2000b. Census 2000 Summary File 3 (SF 3) Sample Data. Geographic Areas: United States, Illinois, St. Clair County (Illinois). Downloaded from internet on 10/14/2005. http://factfinder.census.gov/

- . 2004. American Community Survey Data Profile Highlights-Factsheet. 2004. Population Estimate: United States, Illinois, St. Clair County, Belleville. American FactFinder. Downloaded on 10/11/2005. http://factfinder.census.gov/
- United States Department of Labor. 2004. News, United States Department of Labor. Bureau of Labor Statistics, National Census of Fatal Occupational Industries in 2003. September 22, 2004. http://stats.bis.gov/oshhome.htm
- United States Environmental Protection Agency (USEPA). 1974. Information on Levels of Environmental Noise Requisite to Protect the Public Health and Welfare with an Adequate Margin of Safety. EPA Report 550/9-74-004.
- . 2003. 1999 National Emission Inventory. Database downloaded from USEPA website on 11/25/2003, http://www.epa.gov/ttn/chief/net/1999inventory.html.
- . 2005. 8-Hr Ozone and Designated PM-2.5 Nonattainment Areas. United States Environmental Protection Agency, Green Book, http://www.epa.gov/oar/oaqps/greenbk/qncl3.html updated 10/14/2005.
- United States Fish and Wildlife Service (USFWS). 1995. Nongame Birds of Management Concern -- the 1995 List. U.S. Fish and Wildlife Service, Office of Migratory Bird Management, Washington, D.C.
- United States Geological Survey. 2005. National Earthquake Information Center, United States Geologic Survey. Website accessed October 20. http://neic.usgs.gov/neis/eqlists/large_usa.html
- Wasmer, F. and F. Maunsell. 2002. NMPlot Computer Program. Wasmer Consulting.

APPENDIX A INTERAGENCY AND INTERGOVERNMENTAL COORDINATION FOR ENVIRONMENTAL PLANNING (IICEP)

INTERAGENCY AND INTERGOVERNMENTAL COORDINATION FOR ENVIRONMENTAL PLANNING (IICEP) DISTRIBUTION LIST FOR THE 126 ARW ENVIRONMENTAL ASSESSMENT

US EPA Region 5 77 W. Jackson Blvd. Chicago, IL 60604

Illinois Department of Transportation 2300 S. Dirksen Pkwy. Springfield, IL 62764 Phone: 217-782-7820

Illinois Department of Natural Resources Region IV 4521 Alton Commerce Pkwy. Alton, IL 62002 Phone: 618-462-1181

Illinois Environmental Protection Agency 2009 Mall St. Collinsville, IL 62234 Phone: 618-346-5120

US Army Corps of Engineers St. Louis District 1222 Spruce St. St. Louis, MO 63103-2822 Phone: 314-331-8010

Mr. Bob Delaney St. Clair County Clerk's Office #10 Public Square, 2nd Floor Belleville, IL 62220 Phone: 618- 277-6600 Ext. 2380

US Fish and Wildlife Service Two Rivers National Wildlife Refuge HCR 82 Box 107 Brussels, IL 62013 Phone: 618-883-2524 Richard Nelson, Field Supervisor US Fish and Wildlife Service Rock Island Ecological Services Field Office 4469 48th Ave. Court Rock Island, IL 61201

Mr. William Wheeler, State Historic Preservation Officer Illinois Historic Preservation Agency 1 Old State Capitol Plaza Springfield, IL 62701-1512 Phone: 217-785-1153 Fax: 217-524-7525

Mr. Charles Hartke, Director Illinois Department of Agriculture P.O. Box 19281, State Fairgrounds Springfield, IL 62794-9281 Phone: 217-782-2172

Mr. John P. Froman, Chief Peoria Tribe of Indians of Oklahoma P.O. Box 1527 Miami, OK 74355-1527 Phone: 918-540-2535 Fax: 918-540-2538 jfroman@peoriatribe.com

ANG/CEVP 3500 Fetchet Avenue Andrews AFB MD 20762-5157

«Company» «Address1» «Address2» «Address3» «City», «State» «PostalCode»

Dear Sir/Madam

The Illinois Air National Guard at Scott Air Force Base, Illinois, has prepared a Draft Environmental Assessment (EA) for a proposal to implement construction, demolition, and other projects for improvements at their installation, as well as other actions (Attachment 1).

The environmental analysis for the Proposed Action is being conducted by the Air National Guard in accordance with the Council on Environmental Quality guidelines pursuant to the National Environmental Policy Act of 1969. In accordance with Executive Order 12372, Intergovernmental Review of Federal Programs, we request your participation by reviewing this Draft EA, and solicit your comments concerning the proposal and any potential environmental consequences of the action. Section 5.0 of this EA analyzes the cumulative impacts of this and other actions in the region of influence (ROI). If there are other known actions in the ROI that are not included in Section 5.0 please list those actions in your comments. A listing of federal and state agencies that have been contacted is attached (Attachment 2). If there are any additional agencies that you feel should review and comment on the Draft EA, please indicate these in your response to us.

Any questions concerning the proposal should be directed to our consultant, Science Applications International Corporation (SAIC). The point of contact is Ms. Kate L. Bartz. She can be reached at (520) 326-0951 if you have any questions or concerns. Please forward your written comments to Ms. Bartz, in care of SAIC, 2617 East 7th Street, Tucson, Arizona 85716, no later than March 10, 2006. Thank you for your assistance.

Sincerely

HARRY A. KNUDSEN, JR. Chief, Environmental Planning Branch

Attachments:

- 1. Draft Environmental Assessment for ILANG Construction Projects
- 2. Distribution list



www.illinois-history.gov PLEASE REFER TO: IHPA LOG #019020906 St. Clair County O'Fallon Security Forces Facility, Tanker Airlift Command Training Center; Addition/Facility 5038, Building 3901, Building 5028, Building 5010; Building 5542 Construction/Modification & Demolition/126th Air Refueling Wing/Scott Air Force Base

Voice (217) 782-4836

March 6, 2006

Ms. Kate L. Bartz SAIC 2617 East 7th Street Tucson, AZ 85716

Dear Madam:

We have reviewed the documentation submitted for the referenced project(s) in accordance with 36 CFR Part 800.4. Based upon the information provided, no historic properties are affected. We, therefore, have no objection to the undertaking proceeding as planned.

Please retain this letter in your files as evidence of compliance with section 106 of the National Historic Preservation Act of 1966, as amended. This clearance remains in effect for two (2) years from date of issuance. It does not pertain to any discovery during construction, nor is it a clearance for purposes of the Illinois Human Skeletal Remains Protection Act (20 ILCS 3440).

If you are an applicant, please submit a copy of this letter to the state or federal agency from which you obtain any permit, license, grant, or other assistance.

Sincerely, Haaker

Anne E. Haaker Deputy State Historic Preservation Officer

AEH

cc: Kevin Jacobs, Major, ILANG

Printed on Recycled Paper



United States Department of the Interior

FISH AND WILDLIFE SERVICE Marion Illinois Suboffice (ES) 8588 Route 148 Marion, IL 62959 (618) 997-3344

March 31, 2006

Ms. Kate L. Bartz Science Applications International 2617 East 7th Street Tucson, Arizona 85716

Dear Ms. Bartz:

We have reviewed the draft Environmental Assessment (EA), dated February 6, 2006, prepared for the Illinois Air National Guard to implement construction, demolition and other projects associated with improvements to their facilities located at Scott Air Force Base, St. Clair County, Illinois. The proposed action includes: 1) construct Pumphouse spill containment; 2) addition to Mid-Field Fire Station; 3) construct Pavements and Grounds Facility; 4) construct new Security Forces Facility; 5) addition/alteration to Deployment Processing Center; 6) construct Tanker Airlift Command Training Center; and, 7) demolish Building 5542. These comments are provided under the authority of and in accordance with the provisions of the Fish and Wildlife Coordination Act (48 Stat. 401, as amended; 16 U.S.C. 661 et seq.); and the Endangered Species Act of 1973, as amended.

According to the draft EA, approximately 1.75 acres of maintained turf grass and otherwise undeveloped land is expected to be directly impacted and permanently lost due to building construction, parking lots and paving. This includes a few scattered maple and oak trees. Soils and water will be managed in accordance with Best Management Practices and a Storm Water Pollution Prevention Plan. Hazardous materials and wastes will be handled and disposed in accordance with environmental laws. The construction of the spill containment facility around the pumphouse has the potential to reduce or eliminate the future release of hazardous substances from this building.

Air emissions as a result of construction are predicted to increase, but are not likely to result in any long-term impacts to air quality. Similarly, new operational emissions are not expected to result in any long-term impacts on air quality. Finally, over the long-term, noise is not expected to increase in the project vicinity.

Based upon the information provided in the draft EA and summarized above, the proposed action is not likely to result in any measurable direct or indirect effects to federally listed threatened or endangered species. Therefore, we concur that the proposed activity is not likely to adversely affect any known federally listed threatened and endangered species. This precludes the need for further action on this project as required under Section 7 of the Endangered Species Act, as amended. Should the project be modified, or new information indicate listed or proposed species may be affected, consultation or additional coordination with this office, as appropriate, should be initiated.

Given the highly developed nature of the project area, we also concur that the proposed activity will have minimal impacts on fish and wildlife resources in the project area, including migratory birds.

Thank you for the opportunity to review the draft EA and provide comments. If you have any questions, please contact me at 618/997-3344, ext. 340.

Sincerely, (dl Collins

Assistant Field Supervisor

cc: IDNR (Rettig, Kath, Kemper)

SIP	State Implementation Plan
SO_2	sulfur dioxide
SO_x	sulfur oxides
SPCC	Spill Prevention Control and Countermeasure
SWDA	Solid Waste Disposal Act
SWMP	Solid Waste Management Plan
SWPPP	Storm Water Pollution Prevention Plan
SY	square yard
TACC	Tanker Airlift Command Center
TACTC	Tanker Airlift Command Training Center
TPY	tons per year
U.S.	United States
USACE	United States Army Corps of Engineers
USAF	United States Air Force
USC	United States Code
USCB	United States Census Bureau
USEPA	United States Environmental Protection Agency
USFWS	United States Fish and Wildlife Service
UST	underground storage tank
UTA	unit training assembly
VOC	volatile organic compound
WWTP	Waste Water Treatment Plant